CALIX, INC (CALX)

10-Q

Quarterly report pursuant to sections 13 or 15(d) Filed on 11/02/2012 Filed Period 09/29/2012



UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, DC 20549

	FC	ORM 10-Q
(Mark O	ne)	
X	QUARTERLY REPORT PURSUANT TO SEC 1934	TION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF
	For the quarterly	y period ended September 29, 2012 OR
	TRANSITION REPORT PURSUANT TO SEC 1934	TION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF
	For the transiti	ion period from to
	Commissi	ion File Number: 001-34674
		Calix, Inc. egistrant as Specified in Its Charter)
	Delaware	68-0438710
	(State or Other Jurisdiction of Incorporation or Organization)	(I.R.S. Employer Identification No.)
	(Address of Pri	owell Blvd., Petaluma, CA 94954 incipal Executive Offices) (Zip Code) (707) 766-3000 ephone Number, Including Area Code)
during t		reports required to be filed by Section 13 or 15(d) the Securities Exchange Act of 1934 gistrant was required to file such reports), and (2) has been subject to such filing
required	·	ectronically and posted on its corporate Web site, if any, every Interactive Data File on S-T (§232.405 of this chapter) during the preceding 12 months (or for such shorter Yes: No:
	Indicate by check mark whether the registrant is a large acceler inition of "large accelerated filer," "accelerated filer" and "sma	rated filer, an accelerated filer, a non-accelerated filer or a smaller reporting company. aller reporting company" in Rule 12b-2 of the Exchange Act.
Large A	Accelerated Filer	Accelerated Filer
Non-ac	celerated filer	eporting Company) Smaller Reporting Company
	Indicate by check mark whether the registrant is a shell compar As of October 27, 2012, there were 48,418,075 share s of the F	ny (as defined in Rule 12b-2 of the Exchange Act). Yes: □ No: ⊠ Registrant's common stock, par value \$0.025 outstanding.

Calix, Inc.

Form 10-Q

TABLE OF CONTENTS

PART I. FINANCIAL INFORMATION (Unaudited)

	Item 1. Financial Statements	3
	Condensed Consolidated Balance Sheets	3
	Condensed Consolidated Statements of Comprehensive Loss	4
	Condensed Consolidated Statements of Cash Flows	5
	Notes to Condensed Consolidated Financial Statements	6
	Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations	<u>14</u>
	<u>Overview</u>	<u>14</u>
	Critical Accounting Policies and Estimates	<u>16</u>
	Results of Operations	<u>17</u>
	Liquidity and Capital Resources	<u>20</u>
	Item 3. Quantitative and Qualitative Disclosures About Market Risk	<u>21</u>
	Item 4. Controls and Procedures	<u>22</u>
PA]	RT II. OTHER INFORMATION	<u>23</u>
	Item 1. Legal Proceedings	<u>23</u>
	Item 1A. Risk Factors	<u>23</u>
	Item 2. Unregistered Sales of Equity Securities and Use of Proceeds	<u>38</u>
	Item 3. Defaults Upon Senior Securities	38
	Item 4. Mine Safety Disclosures	<u>38</u>
	Item 5. Other Information	<u>38</u>
	Item 6. Exhibits	<u>39</u>
SIG	<u>PNATURES</u>	<u>40</u>
	2	

PART I. FINANCIAL INFORMATION

ITEM 1. Financial Statements

CALIX, INC.

CONDENSED CONSOLIDATED BALANCE SHEETS (In thousands, except share and per share data)

	September 29, 2012	December 31, 2011
	(Unaudited)	
ASSETS		
Current assets:		
Cash and cash equivalents	\$ 57,422	\$ 38,938
Restricted cash	_	754
Accounts receivable, net	56,392	47,943
Inventory	30,175	44,604
Deferred cost of revenue	16,960	8,324
Prepaids and other current assets	4,383	4,429
Total current assets	165,332	144,992
Property and equipment, net	17,381	16,130
Goodwill	116,175	116,175
Intangible assets, net	66,941	80,048
Other assets	1,774	2,194
Total assets	\$ 367,603	\$ 359,539
LIABILITIES AND STOCKHOLDERS' EQUITY		
Current liabilities:		
Accounts payable	\$ 16,436	\$ 14,250
Accrued liabilities	34,137	36,214
Deferred revenue	30,674	16,783
Total current liabilities	81,247	67,247
Long-term portion of deferred revenue	15,299	13,347
Other long-term liabilities	896	1,528
Total liabilities	97,442	82,122
Commitments and contingencies (See Note 7)		
Stockholders' equity:		
Preferred stock, \$0.025 par value; 5,000,000 shares authorized; no shares issued and outstanding as of September 29, 2012 and December 31, 2011	_	_
Common stock, \$0.025 par value; 100,000,000 shares authorized; 48,417,915 shares and 47,825,200 shares issued and outstanding as of September 29, 2012 and December 31, 2011, respectively	1,210	1,195
Additional paid-in capital	754,755	740,309
Accumulated other comprehensive income	133	98
Accumulated deficit	(485,937)	(464,185)
Total stockholders' equity	270,161	277,417
Total liabilities and stockholders' equity	\$ 367,603	\$ 359,539

See accompanying notes to condensed consolidated financial statements.

CALIX, INC.

CONDENSED CONSOLIDATED STATEMENTS OF COMPREHENSIVE LOSS (In thousands, except per share data) (Unaudited)

	Three Months Ended		Nine Mor		nths Ended			
	Sej	otember 29, 2012	Sej	otember 24, 2011	Se	ptember 29, 2012	Se	ptember 24, 2011
Revenue	\$	81,301	\$	83,655	\$	238,794	\$	253,084
Cost of revenue:								
Products and services (1)		45,707		49,002		132,797		143,209
Merger-related expenses		_		_		_		19,966
Amortization of intangible assets		2,088		2,806		5,451		7,510
Total cost of revenue		47,795		51,808		138,248		170,685
Gross profit		33,506		31,847		100,546		82,399
Operating expenses:								
Research and development (1)		16,165		16,717		49,604		50,340
Sales and marketing (1)		15,093		12,593		44,880		38,831
General and administrative (1)		6,773		5,475		19,682		21,450
Merger-related and other expenses (1)		_		1,404		_		12,927
Amortization of intangible assets		2,552		2,552		7,656		6,016
Total operating expenses		40,583		38,741		121,822		129,564
Loss from operations		(7,077)		(6,894)		(21,276)		(47,165)
Interest and other income (expense), net:								
Interest income		3		11		14		80
Interest expense		(41)		(48)		(140)		(139)
Other income (expense), net		19		35		(70)		64
Loss before provision for income taxes		(7,096)		(6,896)		(21,472)		(47,160)
Provision for income taxes		44		38		280		176
Net loss	\$	(7,140)	\$	(6,934)	\$	(21,752)	\$	(47,336)
Net loss per common share:								
Basic and diluted	\$	(0.15)	\$	(0.15)	\$	(0.45)	\$	(1.06)
Weighted-average number of shares used to compute net loss per common share:								
Basic and diluted		48,353		47,128		48,059		44,866
Other comprehensive income, net of tax:								
Unrealized loss on investment, net	\$	_	\$	(2)	\$	_	\$	(22)
Foreign currency translation adjustment, net		24		42		35		76
Total other comprehensive income, net of tax		24		40		35		54
Comprehensive loss	\$	(7,116)	\$	(6,894)	\$	(21,717)	\$	(47,282)
(1) Includes stock-based compensation as follows:	_		_		_		_	
Cost of revenue	\$	345	\$	306	\$	1,089	\$	1,141
Research and development		1,017		886		3,129		3,761
Sales and marketing		1,263		1,127		3,840		3,256
General and administrative		1,682		1,407		4,903		7,845
Merger-related and other expenses	Φ	4 207	•	70 3 706	Ф.	12.061	<u>¢</u>	1,234
	\$	4,307	\$	3,796	\$	12,961	\$	17,237

See accompanying notes to condensed consolidated financial statements.

CALIX, INC. CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (In thousands) (Unaudited)

	Nine Mor	nths Ended
	September 29, 2012	September 24 2011
Operating activities:		
Net loss	\$ (21,752)	\$ (47,336
Adjustments to reconcile net loss to net cash provided by operating activities:		
Amortization of premiums relating to available-for-sale securities	_	229
Depreciation and amortization	6,351	5,949
Loss on retirement of property and equipment	262	2,27
Amortization of intangible assets	13,107	13,520
Stock-based compensation	12,961	17,23
Changes in operating assets and liabilities:		
Restricted cash	754	_
Accounts receivable, net	(8,449)	12,32
Inventory	14,429	9,63
Deferred cost of revenue	(8,636)	(1,16
Prepaids and other assets	491	(2,29
Accounts payable	2,186	(10,12
Accrued liabilities	(2,098)	2,85
Deferred revenue	15,844	5,80
Other long-term liabilities	(633)	(17
Net cash provided by operating activities	24,817	8,73
nvesting activities:		
Purchase of property and equipment	(7,879)	(6,27
Sales and maturities of marketable securities	_	29,75
Acquisition of Occam Networks, net of cash acquired	_	(60,80
Net cash used in investing activities	(7,879)	(37,32
inancing activities:		
Proceeds from exercise of stock options and other	158	76
Proceeds from employee stock purchase plan	2,222	2,06
Taxes withheld upon vesting of restricted stock units and restricted stock awards	(880)	(10,37
Net cash provided by (used in) financing activities	1,500	(7,54
ffect of exchange rate changes on cash and cash equivalents	46	7
let increase (decrease) in cash and cash equivalents	18,484	(36,05
ash and cash equivalents at beginning of period	38,938	66,30
ash and cash equivalents at end of period	\$ 57,422	\$ 30,24
on-cash investing activities:	7 2.,,22	
Value of common stock issued in acquisition		117,25
Fair value of equity awards assumed in connection with acquisition		1,37

CALIX, INC.

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

1. Company and Basis of Presentation

Company

Calix, Inc. (together with its subsidiaries, "Calix," the "Company," "our," "we," or "us") was incorporated in August 1999, and is a Delaware corporation. We are a leading provider in North America of broadband communications access systems and software for fiber- and copper-based network architectures that enable communications service providers ("CSPs") to transform their networks and connect to their residential and business subscribers. We enable CSPs to provide a wide range of revenue-generating services, from basic voice and data to advanced broadband services, over legacy and next-generation access networks. We focus solely on CSP access networks, the portion of the network that governs available bandwidth and determines the range and quality of services that can be offered to subscribers. We develop and sell carrier-class hardware and software products, which we refer to as the Unified Access portfolio that are designed to enhance and transform CSP access networks to meet the changing demands of subscribers rapidly and cost-effectively.

Basis of Presentation

The accompanying unaudited condensed consolidated financial statements, including the accounts of Calix, Inc. and its wholly owned subsidiaries, have been prepared in accordance with the requirements of the U.S. Securities and Exchange Commission ("SEC") for interim reporting. As permitted under those rules, certain footnotes or other financial information that are normally required by U.S. generally accepted accounting principles ("GAAP") can be condensed or omitted. In the opinion of management, the financial statements include all normal and recurring adjustments that are considered necessary for the fair presentation of the Company's financial position and operating results. All significant intercompany balances and transactions have been eliminated in consolidation. The condensed consolidated balance sheet at December 31, 2011 has been derived from the audited financial statements at that date.

The results of the Company's operations can vary during each quarter of the year. Therefore, the results and trends in these interim financial statements may not be the same as those for the full year or any future periods. The information included in this quarterly report on Form 10-Q should be read in conjunction with the audited financial statements for the year ended December 31, 2011, included in the Company's Form 10-K.

The Company operates on a 4-4-5 fiscal calendar which divides the year into four quarters, with each quarter grouped into two 4-week months and one 5-week month. The Company's fiscal year ends on December 31. The preparation of financial statements in conformity with GAAP for interim financial reporting requires management to make estimates and assumptions that affect the amounts reported in the condensed consolidated financial statements and accompanying notes. Actual results could differ from those estimates.

2. Significant Accounting Policies

The Company's significant accounting policies are disclosed in our Annual Report on Form 10-K for the year ended December 31, 2011. Our significant accounting policies did not materially change during the nine months ended September 29, 2012.

Recent Accounting Pronouncements

In the nine months ended September 29, 2012, there were no accounting standard updates that would materially impact the Company's financial statements.

3. Goodwill

Goodwill is not amortized but instead is subject to an annual impairment test or more frequently if events or changes in circumstances indicate that it may be impaired. We evaluate goodwill on an annual basis as of the end of the second quarter of each year. Management has determined that we operate as a single reporting unit and, therefore, evaluates goodwill impairment at the enterprise level.

To evaluate for impairment, the Company utilizes a two-step process. The first step requires the Company to compare its fair value to its carrying value including goodwill. The Company determines its fair value using both an income approach and a market approach. Under the income approach, the Company determines fair value based on estimated future cash flows, discounted by an estimated weighted-average cost of capital, which reflects the overall level of inherent risk of the Company and the rate of return an outside investor would expect to earn. Under the market-based approach, the Company utilizes

information regarding the Company as well as publicly available industry information to determine earnings multiples that are used to value the Company. If the carrying value of the Company exceeds its fair value, the Company performs the second step of the goodwill impairment test to determine the amount of impairment loss. The second step of the goodwill impairment test involves comparing the implied fair value of goodwill with the carrying value of goodwill. An impairment charge is recognized for the excess of the carrying value of goodwill over its implied fair value.

In accordance with our annual goodwill impairment test and in consideration of the recent significant decline in our stock price, we evaluated the potential impairment of goodwill as of June 30, 2012 in July 2012. The goodwill impairment testing process involved the use of significant assumptions, estimates and judgments, and is subject to inherent uncertainties and subjectivity in determination of the fair value of our sole reporting unit. In performing the first step of the goodwill impairment test, we determined the fair value of our reporting unit by combining two valuation methods, a discounted cash flow analysis (DCF) and market multiples of comparable publicly traded companies. Under the DCF method, we prepared annual projections of future cash flows over a period of five years (the "discrete projection period") and applied a terminal value assumption to the final year within the discrete projection period to estimate the total value of the cash flows beyond the final year. These projected cash flow estimates were then discounted using a discount rate that reflected market-based estimates based on comparable publicly traded companies, and included an additional risk premium specific to Calix. Under the market multiples method, fair value was determined by applying multiples of Revenue and EBITDA (earnings before interest, taxes, depreciation and amortization). In addition, we analyzed the fair value of our reporting unit and our total market capitalization for reasonableness, taking into account certain factors, including control premium, which were based on values observed in market transactions. Based on our analyses, we determined that the fair value of our reporting unit exceeded the carrying value by approximately 15%, and therefore the second step of the goodwill test did not need to be performed. However, significant changes in these estimates and assumptions could create future impairment losses to goodwill.

At the end of the third quarter of 2012, the Company reviewed events and changes to its business subsequent to the impairment test performed in July 2012 and concluded that there were no indicators of impairment to the carrying value of its goodwill during the three months ended September 29, 2012. As of September 29, 2012, there was no impairment to the carrying value of the Company's goodwill.

4. Intangible Assets

Intangible assets are carried at cost, less accumulated amortization. The details of intangible assets as of September 29, 2012 and December 31, 2011 are disclosed in the following table (in thousands):

	September 29, 2012					December 31, 2011						
		Gross Carrying Amount		Accumulated Amortization		Net		Gross Carrying Amount		.ccumulated .mortization		Net
Core developed technology	\$	68,964	\$	(36,897)	\$	32,067	\$	52,694	\$	(31,447)	\$	21,247
Customer relationships		54,740		(19,866)		34,874		54,740		(12,209)		42,531
Purchase order backlog		_		_		_		4,260		(4,260)		_
Trade name		_		_		_		2,290		(2,290)		_
Total amortizable intangible assets		123,704		(56,763)	Ī	66,941		113,984		(50,206)		63,778
In-process technology		_		_		_		16,270		_		16,270
Total intangible assets, excluding goodwill	\$	123,704	\$	(56,763)	\$	66,941	\$	130,254	\$	(50,206)	\$	80,048

At the end of the first quarter of 2012, upon the completion of the research and development efforts associated with our \$16.3 million in-process technology that was acquired from Occam, we determined that this technology had a useful life of 5 years and therefore reclassified it as core developed technology. We began amortizing this intangible asset to cost of revenue during the second quarter of 2012.

Amortization expense was \$4.6 million and \$13.1 million for the three and nine months ended September 29, 2012, respectively, and \$5.4 million and \$13.5 million for the three and nine months ended September 24, 2011, respectively.

Expected future amortization expense for the fiscal years indicated is as follows (in thousands):

<u>Period</u>	Expected Amortization Expense
Remainder of 2012	\$ 4,640
2013	18,561
2014	18,561
2015	18,561
2016	5,805
2017	813
Total	\$ 66,941

5. Cash and Cash Equivalents

Cash and cash equivalents consisted of the following (in thousands):

	Se	ptember 29, 2012	Ι	December 31, 2011
Cash	\$	37,586	\$	19,109
Money market funds		19,836		19,829
Total cash and cash equivalents	\$	57,422	\$	38,938

As of September 29, 2012, the Company did not hold any marketable securities and therefore there were no unrealized gains or losses. Realized gains and losses on the investments are determined on the specific identification method and are reflected in results of operations. The Company did not experience any significant realized gains or losses on its investments through September 29, 2012.

6. Balance Sheet Details

Inventory consisted of the following (in thousands):

	\$ September 29, 2012	December 31, 2011
Raw materials	\$ 2,351	\$ 3,077
Finished goods	27,824	41,527
Total inventory	\$ 30,175	\$ 44,604

Accounts receivable, net consisted of the following (in thousands):

	S	September 29, 2012	December 31, 2011
Accounts receivable	\$	57,608	\$ 49,180
Allowance for doubtful accounts		(439)	(402)
Product return reserve		(777)	 (835)
Accounts receivable, net	\$	56,392	\$ 47,943

Property and equipment, net consisted of the following (in thousands):

	 September 29, 2012	Ι	December 31, 2011
Computer equipment and purchased software	\$ 30,210	\$	28,477
Test equipment	33,562		29,849
Furniture and fixtures	1,469		1,480
Leasehold improvements	6,622		6,342
Total	71,863		66,148
Accumulated depreciation and amortization	(54,482)		(50,018)
Property and equipment, net	\$ 17,381	\$	16,130

Accrued liabilities consisted of the following (in thousands):

	Se	September 29, 2012		cember 31, 2011
Accrued warranty	\$	11,697	\$	12,104
Accrued compensation and related benefits		11,168		12,406
Accrued professional and consulting fees		2,128		1,741
Accrued excess and obsolete inventory at contract manufacturers		1,745		3,784
Accrued customer rebates		1,506		1,549
Accrued business travel expenses		1,137		383
Sales and use tax payable		1,065		861
Accrued other		3,691		3,386
Total accrued liabilities	\$	34,137	\$	36,214

7. Commitments and Contingencies

Commitments

The Company's principal commitments consist of obligations under operating leases for office space and non-cancelable outstanding purchase obligations. These commitments as of December 31, 2011 are disclosed in our Annual Report on Form 10-K, and have not changed materially during the nine months ended September 29, 2012.

In addition, on August 20, 2012, Calix and Ericsson Inc. ("Ericsson") entered into an Asset Purchase Agreement under which Calix will make a one-time cash payment to acquire Ericsson's fiber access assets ("EFAA"), including the Ericsson EDA 1500 GPON solution and its complementary ONT portfolio. The transaction is expected to close in the fourth quarter of 2012, and will be accounted for using the acquisition method of accounting in accordance with the accounting standard for business combinations.

Accrued Warranty

The Company provides a warranty for its hardware products. Hardware generally has a one to five-year warranty from the date of shipment. The Company accrues for potential warranty claims based on the Company's historical claims experience. The adequacy of the accrual is reviewed on a periodic basis and adjusted, if necessary, based on additional information as it becomes available.

Changes in the Company's warranty reserve were as follows (in thousands):

		Three Mo	onths E	nded		Nine Mo	nths Ended			
	Se	ptember 29, 2012	Se	ptember 24, 2011	Se	ptember 29, 2012	Se	ptember 24, 2011		
Balance at beginning of period	\$	11,277	\$	13,133	\$	12,104	\$	3,789		
Accrued warranty from the Occam acquisition		_		_		_		8,500		
Warranty charged to cost of revenue		1,416		1,341		3,436		4,560		
Utilization of warranty		(996)		(2,218)		(3,843)		(4,593)		
Balance at end of period	\$	11,697	\$	12,256	\$	11,697	\$	12,256		

Litigation

From time to time, the Company is involved in various legal proceedings arising from the normal course of business activities.

On September 16, 2010, the Company, two direct, wholly owned subsidiaries of the Company, and Occam entered into an Agreement and Plan of Merger and Reorganization (the "Merger Agreement"). In response to the announcement of the Merger Agreement, on September 17, 2010, September 20, 2010 and September 21, 2010, three purported class action complaints were filed in the California Superior Court for Santa Barbara County: Kardosh v. Occam Networks, Inc., et al. (Case No. 1371748), or the Kardosh complaint; Kennedy v. Occam Networks, Inc., et al. (Case No. 1371762); and Moghaddam v. Occam Networks, Inc., et al. (Case No. 1371802). The three complaints, which are referred to collectively as the California class action complaints, were substantially similar. Each of the California class action complaints named Occam, the pre-acquisition members of the Occam board of directors and us as defendants.

On November 2, 2010, the three California class action complaints were consolidated into a single action, with the plaintiffs in the Kardosh complaint appointed as the lead plaintiffs, and on November 19, 2010, the California Superior Court issued an order staying the California class action complaints in favor of a substantively identical stockholder class action pending in the Delaware Court of Chancery (see below). On July 31, 2012, the consolidated California class action complaints were dismissed without prejudice as to all defendants.

On October 6, 2010, a purported class action complaint was filed by stockholders of Occam in the Delaware Court of Chancery: Steinhardt v. Howard-Anderson, et al. (Case No. 5878-VCL). On November 24, 2010, these stockholders filed an amended complaint, or the amended Steinhardt complaint. The amended Steinhardt complaint names Occam and the members of the Occam board of directors as defendants. The amended Steinhardt complaint does not name Calix as a defendant.

The amended Steinhardt complaint generally alleges that the members of the Occam board breached their fiduciary duties in connection with the acquisition of Occam by Calix, by, among other things, engaging in an allegedly unfair process and agreeing to an allegedly unfair price for the merger transaction. The amended Steinhardt complaint also alleges that Occam and the former members of the Occam board breached their fiduciary duties by failing to disclose certain allegedly material facts about the merger transaction in the preliminary proxy statement and prospectus included in the Registration Statement on Form S-4 that Calix filed with the SEC on November 2, 2010. The amended Steinhardt complaint sought injunctive relief rescinding the merger transaction and award of damages in an unspecified amount, as well as plaintiffs' costs, attorney's fees, and other relief.

The merger transaction was completed on February 22, 2011. On January 6, 2012, the Delaware court ruled on a motion for sanctions brought by the defendants in the Delaware case against certain of the lead plaintiffs. The Delaware court found that lead plaintiffs Michael Steinhardt, Steinhardt Overseas Management, L.P., and Ilex Partners, L.L.C., collectively the "Steinhardt Plaintiffs," had engaged in improper trading of Calix shares, and dismissed the Steinhardt Plaintiffs from the case with prejudice. The court further held that the Steinhardt Plaintiffs are: (i) barred from receiving any recovery from the litigation, (ii) required to self-report to the SEC, (iii) directed to disclose their improper trading in any future application to serve as lead plaintiff, and (iv) ordered to disgorge trading profits of \$0.5 million, to be distributed to the remaining members of the class of former Occam stockholders. The Delaware court also granted the motion of the remaining lead plaintiffs, Herbert Chen and Derek Sheeler, for class certification, and certified Messrs. Chen and Sheeler as class representatives. Chen and Sheeler, on behalf of the class of similarly situated former Occam stockholders, continue to seek an award of damages in an unspecified amount.

The Company believes that the allegations in the Delaware action are without merit and intends to continue to vigorously contest the action. However, there can be no assurance that the Company will be successful in defending this ongoing action. In addition, the Company has obligations, under certain circumstances, to hold harmless and indemnify each of the former Occam directors against judgments, fines, settlements and expenses related to claims against such directors and otherwise to the fullest extent permitted under Delaware law and Occam's bylaws and certificate of incorporation. Such obligations may apply to this lawsuit.

The Company is not currently a party to any other legal proceedings that, if determined adversely to the Company, would individually or in the aggregate have a material adverse effect on the Company's business, operating results or financial condition.

8. Fair Value Measurements

In accordance with Accounting Standard Codification Topic 820, Fair Value Measurements and Disclosures, ("ASC Topic 820"), the Company measures its cash equivalents and marketable securities at fair value on a recurring basis. ASC Topic 820 clarifies that fair value is an exit price, representing the amount that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants. As such, fair value is a market-based measurement that should be determined based on assumptions that market participants would use in pricing an asset or liability. As a basis for considering such assumptions, ASC Topic 820 establishes a three-tier value hierarchy, which prioritizes the inputs used in measuring fair value as follows:

- Level 1 Observable inputs that reflect quoted prices (unadjusted) for identical assets or liabilities in active markets.
- Level 2 Observable inputs other than quoted prices included in Level 1 for similar instruments in active markets, quoted prices for identical or similar instruments in markets that are not active, and model-driven valuations in which all significant inputs and significant value drivers are observable in active markets.
- Level 3 Unobservable inputs to the valuation derived from fair valuation techniques in which one or more significant inputs or significant value drivers are unobservable. The fair value hierarchy also requires the Company to maximize the use of observable inputs, when available, and to minimize the use of unobservable inputs when determining inputs and determining

fair value.

As of September 29, 2012 and December 31, 2011, the fair values of certain of the Company's financial assets were determined using the following inputs (in thousands):

As of September 29, 2012	1	Level 1	Total
Money market funds	\$	19,836	\$ 19,836
Total	\$	19,836	\$ 19,836
As of December 31, 2011		evel 1	Total
Money market funds	\$	19,829	\$ 19,829
Total	\$	19,829	\$ 19,829

The Company's valuation techniques used to measure the fair values of money market funds were derived from quoted market prices as active markets for these instruments exist. The Company has no level 2 or level 3 financial assets.

9. Net Loss per Share

The following table sets forth the computation of basic and diluted net loss per share for the periods indicated (in thousands, except per share data):

		Three Mo	nths I	Ended		Nine Moi	nths E	hs Ended	
	Se	ptember 29, 2012	Se	ptember 24, 2011	Se	eptember 29, 2012	Se	ptember 24, 2011	
Numerator:									
Net loss	\$	(7,140)	\$	(6,934)	\$	(21,752)	\$	(47,336)	
Denominator:									
Weighted-average common shares outstanding		48,353		47,128		48,059		44,866	
Basic and diluted net loss per common share	\$	(0.15)	\$	(0.15)	\$	(0.45)	\$	(1.06)	

As the Company incurred net losses in the periods presented, the following table displays the Company's other outstanding common stock equivalents that were excluded from the computation of diluted net loss per share, as the effect of including them would have been anti-dilutive (in thousands):

	September 29, 2012	S	September 24, 2011
Restricted stock units	2,016	\$	2,074
Stock options	2,204		1,647
Employee stock purchase plan	349		119
Warrants	23		24
Total common stock equivalents	4,592		3,864

10. Stockholders' Equity

Capital Structure

The Company maintains three equity incentive plans, the 2000 Stock Plan, the 2002 Stock Plan and the 2010 Equity Incentive Award Plan (together, the "Plans"). These plans were approved by the stockholders and are described in the Company's Form 10-K filed with the SEC on February 24, 2012. In January 2012, the board, upon recommendation by our compensation committee, approved the Long Term Incentive Program, under the 2010 Equity Incentive Award Plan. Under the Long Term Incentive Program certain key employees of the Company are eligible for equity awards based on the Company's stock price performance.

Preferred Stock

The board of directors has the authority, without action by stockholders with the exception of stockholders who hold board positions, to designate and issue up to 5.0 million shares of preferred stock in one or more series and to fix the rights, preferences, privileges and restrictions thereof. These rights, preferences and privileges could include dividend rights,

conversion rights, voting rights, terms of redemption, liquidation preferences, sinking fund terms and the number of shares constituting any series or the designation of such series, any or all of which may be greater than the rights of common stock. The issuance of the Company's preferred stock could adversely affect the voting power of holders of common stock and the likelihood that such holders will receive dividend payments and payments upon liquidation. In addition, the issuance of preferred stock could have the effect of delaying, deferring or preventing a change in control of the Company or other corporate action. Subsequent to the Company's initial public offering and the conversion of all preferred stock outstanding at that date, the board of directors has not designated any rights, preference or powers of any preferred stock and no shares of preferred stock have been issued.

Stock Based Compensation

Stock-based compensation expense associated with stock options, restricted stock units ("RSUs"), performance restricted stock units, restricted stock awards ("RSAs") and purchase rights under our Employee Stock Purchase Plan ("ESPP") is measured at the grant date, based on the fair value of the award, and is recognized as expense over the remaining requisite service period. During the three and nine months ended September 29, 2012, the Company recorded stock-based compensation expense of \$4.3 million and \$13.0 million, respectively. During the three and nine months ended September 24, 2011, the Company recorded stock-based compensation expense of \$3.8 million and \$17.2 million, respectively.

Stock Options

The Company estimates the fair value of stock options in accordance with ASC Topic 718, Compensation – Stock Compensation, ("ASC Topic 718"). The fair value of each option grant is estimated at the grant date using the Black-Scholes option-pricing model with the following weighted-average assumptions:

Stock Options	Three Mo	nths Ended	Nine Months Ended					
	September 29, 2012	September 24, 2011	September 29, 2012	September 24, 2011				
Expected volatility	55%	52%	56%	52%				
Expected life (years)	6.25	6.25	6.25	6.25				
Expected dividend yield	_	_	_	_				
Risk-free interest rate	0.96%	1.41%	1.07%	2.20%				

The Company's computation of expected volatility is based on the Company's peer group of similar companies. The Company's computation of expected term utilizes the simplified method in accordance with Staff Accounting Bulletin No. 110, (SAB 110). The risk-free interest rate is based on the U.S. Treasury yield curve in effect at the time of grant with maturities approximating the grant's expected life. In addition, ASC Topic 718 requires the Company to estimate the number of options that are expected to vest. Thus, the Company applies an estimated forfeiture rate based on actual forfeiture experience. The Company recognizes stock-based compensation expense for the fair values of these awards on a straight-line basis over the requisite service period of each of these awards.

During the three months ended September 29, 2012, the Company granted 235,000 stock options at a weighted-average grant date fair value of \$3.47 per share. During the nine months ended September 29, 2012, the Company granted 732,420 stock options at a weighted-average grant date fair value of \$4.75 per share. During the three months ended September 29, 2012, 33,123 stock options were exercised at a weighted-average exercise price of \$0.57 per share. During the nine months ended September 29, 2012, 91,362 stock options were exercised at a weighted-average exercise price of \$1.73 per share. As of September 29, 2012, unrecognized stock-based compensation expense related to stock options of \$8.4 million, net of estimated forfeitures, was expected to be recognized over a weighted-average period of 2.7 years.

Restricted Stock Units

In September 2009, the Company began to grant RSUs to eligible employees, executives and outside directors. Each RSU represents a right to receive one share of the Company's common stock (subject to adjustment for certain specified changes in the capital structure of the Company) upon the completion of a specific period of continued service.

The Company values the RSUs at fair value, or the market price of the Company's common stock on the date of grant. The Company recognizes stock-based compensation expense for the fair values of these RSUs on a straight-line basis over the requisite service period of these awards.

During the three months ended September 29, 2012, the Company granted 116,851 RSUs with a weighted-average grant date fair value of \$5.15 per share. During the nine months ended September 29, 2012, the Company granted 514,001 RSUs

with a weighted-average grant date fair value of \$6.56 per sh are. During the three and nine months ended September 29, 2012, 219,952 and 235,651 RSUs were released, respectively. As of September 29, 2012, unrecognized stock-based compensation expense related to RSUs of \$16.3 million, net of estimated forfeitures, was expected to be recognized over a weighted-average period of 2.3 years.

Performance Restricted Stock Units

In the first quarter of 2012, the Company began to grant performance RSUs to its executives. The performance criterion is based on the relative total shareholder return ("TSR") of Calix common stock as compared to the TSR of the Company's peer group. The Company established two-year and three-year performance periods that are from January 1, 2012 to December 31, 2013 and 2014, respectively. The TSR is calculated by dividing (a) the average closing trading price for the 90-day period ending on the last day of the applicable performance period by (b) the average closing trading price for the 90-day period immediately preceding January 1, 2012. This TSR is then used to derive the achievement ratio which is then multiplied by the number of units in the grant to derive the common stock to be issued for each performance period.

These performance RSUs are valued in accordance with the guidance of Topic 718, using the Monte Carlo Simulation Technique, which simulates a range of possible future stock prices for Calix and its peer group to determine the fair value for each performance period.

In the first quarter of 2012, the Company granted 170,000 performance RSUs with a weighted-average grant date fair value per RSU of \$15.61. In the third quarter of 2012, the Company granted 20,000 performance RSUs with a weighted-average grant date fair value per RSU of \$8.04. As of September 29, 2012, unrecognized stock-based compensation expense related to performance RSUs of \$1.8 million, net of estimated forfeitures, was expected to be recognized over a weighted-average period of 1.6 years.

Restricted Stock Awards

The Company values RSAs at fair value or the market price of the Company's common stock on the date of grant. The Company recognizes stock-based compensation expense for the fair values of these RSAs on a straight-line basis over the requisite service period of these awards. As of September 29, 2012, unrecognized stock-based compensation expense related to RSAs of \$5.8 million, net of estimated forfeitures, was expected to be recognized over a weighted-average period of 2.8 years.

Employee Stock Purchase Plan

The Company's 2010 Employee Stock Purchase Plan, as amended ("2010 ESPP") allows employees to purchase shares of the Company's common stock through payroll deductions of up to 15 percent of their annual compensation subject to certain Internal Revenue Code limitations. The price of common stock purchased under the plan is 85 percent of the lower of the fair market value of the common stock on the commencement date and exercise date of each six-month offering period.

At the 2012 Annual Meeting of Stockholders, stockholders approved an amendment to our 2010 ESPP to increase the number of shares of common stock reserved for issuance from 1,000,000 shares to 4,300,000 shares. During the nine months ended September 29, 2012, 325,617 shares were purchased under the 2010 ESPP. As of September 29, 2012, there were 3,552,979 shares available for issuance.

As of September 29, 2012, unrecognized stock-based compensation expense related to the ESPP of \$0.2 million was expected to be recognized over a remaining service period of 2 months.

11. Credit Facility

The Company has a revolving credit facility of \$30.0 million based upon a percentage of eligible accounts receivable. Included in the revolving line are amounts available under letters of credit and cash management services. The Company had outstanding letters of credit totaling \$3.6 million and \$2.8 million as of September 29, 2012 and December 31, 2011, respectively. There were no outstanding borrowings under the revolving credit facility as of September 29, 2012 and December 31, 2011. As of September 29, 2012, there was approximately \$26.4 million available for borrowing under this revolving credit facility. The Company is also required to pay commitment fees of 0.25% per year on any unused portions of the facility. As of September 29, 2012 and December 31, 2011, the Company was in compliance with its financial covenants under the credit facility. The revolving credit facility matures on June 30, 2013.

12. Income Taxes

The Company has incurred operating losses since inception and, as such, has not received a tax benefit for these losses. The provision relates to state taxes not based on income, alternative minimum tax, and foreign income tax. Significant components affecting the tax rate include alternative minimum taxes, state tax, foreign tax and the utilization of losses carried forward.

ASC Topic 740, *Accounting for Income Taxes*, provides for the recognition of deferred tax assets if realization of such assets is more likely than not. The Company has established and continues to maintain a full valuation allowance against the Company's net deferred tax assets as the Company does not believe that realization of those assets is more likely than not.

ITEM 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

This report includes "forward-looking statements" within the meaning of Section 27A of the Securities Act and Section 21E of the Securities and Exchange Act of 1934, as amended. All statements other than statements of historical facts are "forward-looking statements" for purposes of these provisions, including any projections of earnings, revenues or other financial items, any statement of the plans and objectives of management for future operations, any statements concerning proposed new products or licensing, any statements regarding product development, any statements regarding future economic conditions or performance, and any statement of assumptions underlying any of the foregoing. In some cases, forward-looking statements can be identified by the use of terminology such as "may," "will," "expects," "plans," "anticipates," "estimates," "potential," or "continue" or the negative thereof or other comparable terminology. Although we believe that the expectations reflected in the forward-looking statements contained herein are reasonable, there can be no assurance that such expectations or any of the forward-looking statements will prove to be correct, and actual results could differ materially from those projected or assumed in the forward-looking statements. Our future financial condition and results of operations, as well as any forward-looking statements, are subject to inherent risks and uncertainties, including but not limited to the Risk Factors set forth under Part II, Item 1A below, and for the reasons described elsewhere in this report. All forward-looking statements or reasons why results may differ included in this report are made as of the date hereof, and we assume no obligation to update these forward-looking statements or reasons why actual results might differ.

Overview

We are a leading provider in North America of broadband communications access systems and software for fiber- and copper-based network architectures that enable communications service providers ("CSPs") to connect to their residential and business subscribers. We enable CSPs to provide a wide range of revenue-generating services, from basic voice and data to advanced broadband services, over legacy and next-generation access networks. We focus solely on CSP access networks, the portion of the network that governs available bandwidth and determines the range and quality of services that can be offered to subscribers. We develop and sell carrier-class hardware and software products, which is referred to as the Unified Access portfolio that are designed to enhance and transform CSP access networks to meet the changing demands of subscribers rapidly and cost-effectively.

We market our access systems and software to CSPs globally through our direct sales force as well as a limited number of resellers. As of September 29, 2012, we have shipped over fifteen million ports of our Unified Access portfolio to more than 1,000 customers worldwide, whose networks serve over 50 million subscriber lines in total. Our customers include 18 of the 20 largest U.S. Incumbent Local Exchange Carriers, or ILECs. In addition, we have over 400 commercial video customers and have enabled over 750 customers to deploy gigabit passive optical network, or GPON, Active Ethernet and point-to-point Ethernet fiber access networks.

Our revenue decreased to \$81.3 million and \$238.8 million for the three and nine months ended September 29, 2012, respectively, from \$83.7 million and \$253.1 million for the three and nine months ended September 24, 2011, respectively. Revenue growth will depend on our ability to continue to sell our access systems and software to existing customers and to attract new customers, including in particular, those customers in the large CSP and international markets. During the second and the third quarters of fiscal 2012, we experienced softness in our business due to lower demand across multiple customer markets. We believe this was due to a slowdown in capital expenditures by service providers increasingly concerned about macro-economic conditions and uncertainties associated with the implementation of regulatory reforms. We expect these issues to continue and these issues may negatively impact our results for the remainder of 2012. Additionally, we expect that our planned acquisition of Ericsson's fiber access assets will have a positive impact to revenue beyond 2012. Since our inception we have incurred significant losses, and as of September 29, 2012, we had an accumulated deficit of \$485.9 million. Our net loss was \$7.1 million and \$21.8 million for the three and nine months ended September 29, 2012, respectively. Our net loss was \$6.9 million and \$47.3 million for the three and nine months ended September 24, 2011, respectively.

Revenue fluctuations result from many factors, including but not limited to: increases or decreases in customer orders for

our products and services, large customer purchase agreements with special revenue considerations, varying budget cycles for our customers and seasonal buying patterns of our customers. More specifically, our customers tend to spend less in the first fiscal quarter as they are finalizing their annual budgets. Customers then typically decide to purchase our products during our second fiscal quarter. In our third fiscal quarter, customers are in the process of deploying such products and as a result there is typically less spending. In addition, difficulties related to deploying products during the winter also tend to limit spending in the third quarter. Finally, in our fourth fiscal quarter, customer purchases typically increase as customers are attempting to spend the rest of their budget for the year. As of September 29, 2012, our deferred revenue of \$46.0 million primarily included certain contracts with customers who receive government supported loans and grants from the U.S. Department of Agriculture's Rural Utility Service ("RUS") that include installation services, special customer arrangements and ratably recognized services. The timing of deferred recognition may cause significant fluctuations in our revenue and operating results from period to period.

Cost of revenue is strongly correlated to revenue and will tend to fluctuate from all of the aforementioned factors that could impact revenue. Other factors that impact cost of revenue include changes in the mix of products delivered to our customers and changes in the cost of our inventory. Cost of revenue includes fixed expenses related to our internal operations which could impact our cost of revenue as a percentage of revenue, if there are large sequential fluctuations to revenue.

Our gross profit and gross margin have been, and will likely be, impacted by several factors, including new product introduction or upgrades to existing products, changes in customer mix, changes in the mix of products demanded and sold, shipment volumes, changes in our product costs, changes in pricing and the extent of customer rebates and incentive programs. We believe our gross margin could increase due to favorable changes in these factors, for example, increases in sales of our advanced E-Series Ethernet service access platforms, upgrades to our C7 platform, new introductions of our P-Series optical network terminals and reductions in the impact of rebate or similar programs. We believe our gross margin could decrease due to unfavorable changes in factors such as increased product costs, pricing decreases due to competitive pressure and an unfavorable customer or product mix. Changes in these factors could have a material impact on our future average selling prices and unit costs. Also, the timing of deferred revenue recognition and related deferred costs can have a material impact on our gross profit and gross margin results. The timing of recognition and the relative size of these arrangements could cause large fluctuations in our gross profit from period to period.

Our operating expenses have fluctuated based on the following factors: timing of variable compensation expenses due to fluctuations in order volumes, timing of salary increases which have historically occurred in the second quarter, timing of bonus accrual due to changes in the Company's performance, timing of research and development expenses including prototype builds and intermittent outsourced development projects and increases in stock-based compensation expenses resulting from modifications to outstanding stock options. Our operating expenses for fiscal 2011 include merger-related expenses and amortization of intangible assets from our acquisition of Occam as discussed in more detail below. As a result of the acquisition we have also incurred increased compensation costs across all operating expense categories due to additional headcount and increased facility related costs. We anticipate that our operating expenses will increase as a result of our planned acquisition of Ericsson's fiber access assets as discussed in more detail below.

As a result of the fluctuations described above and a number of other factors, many of which are outside our control, our quarterly operating results fluctuate from period to period. Comparing our operating results on a period-to-period basis may not be meaningful, and you should not rely on our past results as an indication of our future performance.

Planned Acquisition of Ericsson's Fiber Access Assets

On August 20, 2012, Calix and Ericsson Inc. ("Ericsson") entered into an Asset Purchase Agreement under which Calix will make a one-time cash payment to acquire Ericsson's fiber access assets ("EFAA"), including the Ericsson EDA 1500 GPON solution and its complementary ONT portfolio. In connection with this planned acquisition, Calix will offer employment to up to 61 U.S.-based employees of Ericsson, and will transition ongoing support of the acquired products from Ericsson to Calix.

On August 22, 2012, Calix and Ericsson also signed a global reseller agreement, under which Calix will become Ericsson's preferred global partner for broadband access applications. We expect this agreement to provide Calix with an extensive new global reseller channel, while our acquisition of Ericsson's fiber access portfolio delivers powerful new complements to our industry-leading Unified Access portfolio. This agreement will also provide Ericsson's existing fiber access customers with world-class support and maintenance, and an expanded portfolio of access systems and software from a leading company totally focused on access.

The transaction is expected to close in the fourth quarter of 2012 and will be accounted for using the acquisition method of accounting in accordance with the accounting standard for business combinations. We will consolidate EFAA's financial results in the condensed consolidated financial statement from the date of acquisition. We expect the acquisition to have a

positive impact on our international business over time.

Acquisition of Occam Networks

On February 22, 2011, we completed our acquisition of Occam Networks, Inc. ("Occam"), a provider of innovative broadband access products designed to enable telecommunications service providers to offer bundled voice, video and high speed internet, or Triple Play, services over both fiber optic and copper networks in a stock and cash transaction valued at approximately \$213.1 million which consisted of \$94.5 million of cash consideration and a value of \$118.6 million of common stock issued and equity awards assumed.

As a result of this acquisition, we recorded \$50.6 million in goodwill and \$97.7 million in other intangible assets. We are amortizing the definite-lived intangible assets over their useful lives. Under the acquisition method of accounting rules, we revalued the Occam assets and liabilities acquired at the time of the acquisition, based on their fair value.

Critical Accounting Policies and Estimates

Our financial statements are prepared in accordance with U.S. generally accepted accounting principles, or GAAP. These accounting principles require us to make certain estimates and judgments that can affect the reported amounts of assets and liabilities as of the date of the financial statements, as well as the reported amounts of revenue and expenses during the periods presented. Management bases its estimates, assumptions and judgments on historical experience and on various other factors that are believed to be reasonable under the circumstances. To the extent there are material differences between these estimates and actual results, our financial statements will be affected. Our management evaluates its estimates, assumptions and judgments on an ongoing basis.

Our critical accounting policies and estimates are described under "Critical Accounting Policies and Estimates" in "Management's Discussion and Analysis of Financial Condition and Results of Operations" included in our Annual Report on Form 10-K for the year ended December 31, 2011. During the nine months ended September 29, 2012, there have been no significant changes in our critical accounting policies and estimates.

Impairment of Goodwill and Intangible Assets

Goodwill is not amortized but instead is subject to an annual impairment test or more frequently if events or changes in circumstances indicate that it may be impaired. We evaluate goodwill on an annual basis as of the end of the second quarter of each year. Management has determined that we operate as a single reporting unit and, therefore, evaluates goodwill impairment at the enterprise level. Intangible assets are reviewed for impairment whenever events or changes in circumstances indicate an asset's carrying value may not be recoverable.

To evaluate for impairment, the Company utilizes a two-step process. The first step requires the Company to compare its fair value to its carrying value including goodwill. The Company determines its fair value using both an income approach and a market approach. Under the income approach, the Company determines fair value based on estimated future cash flows, discounted by an estimated weighted-average cost of capital, which reflects the overall level of inherent risk of the Company and the rate of return an outside investor would expect to earn. Under the market-based approach, the Company utilizes information regarding the Company as well as publicly available industry information to determine earnings multiples that are used to value the Company. If the carrying value of the Company exceeds its fair value, the Company performs the second step of the goodwill impairment test to determine the amount of impairment loss. The second step of the goodwill impairment test involves comparing the implied fair value of goodwill with the carrying value of goodwill. An impairment charge is recognized for the excess of the carrying value of goodwill over its implied fair value.

In accordance with our annual goodwill impairment test and in consideration of the recent significant decline in our stock price, we evaluated the potential impairment of goodwill as of June 30, 2012 in July 2012. The goodwill impairment testing process involved the use of significant assumptions, estimates and judgments, and is subject to inherent uncertainties and subjectivity in determination of the fair value of our sole reporting unit. In performing the first step of the goodwill impairment test, we determined the fair value of our reporting unit by combining two valuation methods, a discounted cash flow analysis (DCF) and market multiples of comparable publicly traded companies. Under the DCF method, we prepared annual projections of future cash flows over a period of five years (the "discrete projection period") and applied a terminal value assumption to the final year within the discrete projection period to estimate the total value of the cash flows beyond the final year. These projected cash flow estimates were then discounted using a discount rate that reflected market-based estimates based on comparable publicly traded companies, and included an additional risk premium specific to Calix. Under the market multiples method, fair value was determined by applying multiples of Revenue and EBITDA (earnings before interest, taxes, depreciation and amortization). In addition, we analyzed the fair value of our reporting unit and our total market capitalization for reasonableness, taking into account certain factors, including control premium, which were based on values observed in market transactions. Based on our analyses, we determined that the fair value of our reporting unit exceeded the carrying value

by approximately 15%, and therefore the second step of the goodwill test did not need to be performed. However, significant changes in these estimates and assumptions could create future impairment losses to goodwill.

At the end of the third quarter of 2012, we reviewed events and changes to our business subsequent to the impairment test performed in July 2012 and concluded that there were no indicators of impairment to the carrying value of goodwill during the three months ended September 29, 2012. As of September 29, 2012, there was no impairment to the carrying value of goodwill.

Recent Accounting Pronouncements

In the nine months ended September 29, 2012, there were no new accounting standard updates that would materially impact the Company's financial statements.

Results of Operations

Comparison of the Three and Nine Months Ended September 29, 2012 and September 24, 2011

Revenue

The following table sets forth our revenue:

				Three Months E	nded			Nine Months Ended									
	Se	ptember 29, 2012	Se	ptember 24, 2011		Variance in Dollars	Variance in Percent	s	September 29, 2012	s	eptember 24, 2011		Variance in Dollars	Variance in Percent			
							(in thousands, e	except	percentages)								
Revenue	\$	81,301	\$	83,655	\$	(2,354)	(3)%	\$	238,794	\$	253,084	\$	(14,290)	(6)%			

Our revenue is principally derived in the United States. During the three and nine months ended September 29, 2012 and September 24, 2011, revenue generated in the United States represented approximately 94% and 93%, respectively. Revenue decreased during the three and nine months ended September 29, 2012 compared with the corresponding periods of fiscal 2011, primarily due to a decrease in shipment volume resulting from the softness in demand across multiple customer markets which the company believes is due to a slowdown in capital expenditures by service providers increasingly concerned about macro-economic conditions and uncertainties associated with the implementation of regulatory reforms. We expect these issues to continue and these issues may negatively impact our results for the remainder of 2012.

Cost of Revenue and Gross Profit

The following table sets forth our costs of revenue:

			7	Three Months	Ended		Nine Months Ended						
	Se	eptember 29, 2012	S	eptember 24, 2011	Variance in Dollars	Variance in Percent	s	eptember 29, 2012	s	eptember 24, 2011	Variance in Dollars	Variance in Percent	
					(in	thousands, e	xcep	t percentages)					
Cost of revenue:													
Products and services	\$	45,707	\$	49,002	\$ (3,295)	(7)%	\$	132,797	\$	143,209	\$ (10,412)	(7)%	
Merger-related expenses		_		_	_	— %		_		19,966	(19,966)	(100)%	
Amortization of intangible assets		2,088		2,806	(718)	(26)%		5,451		7,510	(2,059)	(27)%	
Total cost of revenue	\$	47,795	\$	51,808	\$ (4,013)	(8)%	\$	138,248	\$	170,685	\$ (32,437)	(19)%	
Gross profit	\$	33,506	\$	31,847	\$ 1,659	5 %	\$	100,546	\$	82,399	\$ 18,147	22 %	
Gross margin		41%		38%				42%		33%			

Cost of revenues decreased during the three and nine months ended September 29, 2012 compared with the corresponding periods of fiscal 2011, primarily due to the fact that we did not incur any further merger-related expenses subsequent to 2011, a decrease in cost of products and service revenues resulting from decreased revenues and decreased write-down charges for excess and obsolete inventory as a result of improved inventory management, and a decrease in amortization of intangible assets.

The decreases in amortization of intangible assets during the three and nine months ended September 29, 2012 are due to the completion of the amortization of certain intangibles that we acquired from Occam, offset in part by an increase in the

amortization of core developed technologies that began amortizing during the second quarter of 2012 as discussed below. At the end of the first quarter of 2012, upon the completion of the research and development efforts associated with the in-process technology, we determined that this technology had a useful life of 5 years and therefore reclassified it as core developed technology. We began amortizing this intangible asset to cost of revenue during the second quarter of 2012.

Gross margin increased during the three and nine months ended September 29, 2012 compared with the corresponding periods of fiscal 2011, primarily due to the absence of merger-related expenses, lower excess and obsolete inventory write-down charges, and lower intangible assets amortization expenses. Excluding merger-related expenses, gross margin increased slightly from 38% and 40% for the three and nine months ended September 29, 2011, respectively, to 41% and 42% for the three and nine months ended September 29, 2012, respectively, primarily due to a combination of product mix and cost reductions.

Operating Expenses

Research and Development Expenses

The following table sets forth our research and development expenses:

		Three Months Ended							Nine Months Ended							
	Se	eptember 29, 2012	S-	September 24, 2011		variance in Dollars	Variance in Percent	s	eptember 29, 2012	s	eptember 24, 2011		ariance in Dollars	Variance in Percent		
						(ir	thousands, ex	cept	percentages)							
Research and development	\$	16,165	\$	16,717	\$	(552)	(3)%	\$	49,604	\$	50,340	\$	(736)	(1)%		
Percent of total revenue		20%		20%					21%		20%					

Research and development expenses decreased during the three months ended September 29, 2012 compared with the corresponding period in fiscal 2011, primarily due to a decrease in prototype and consulting expenses resulting from the timing of new product development activities, offset in part by an increase in compensation expenses due to increased headcount.

Research and development expenses decreased during the nine months ended September 29, 2012 compared with the corresponding period in fiscal 2011, primarily due to a decrease in prototype expenses resulting from the timing of new product development activities, and a decrease in stock-based compensation expense. These decreases were offset in part by an increase in compensation expenses due to increased headcount resulting from the acquisition of Occam on February 22, 2011 and the expansion of our China development center, and an increase in consulting expenses in connection with our pursuit of OSMINE certification.

Research and development expenses as a percentage of revenue for the three and nine months ended September 29, 2012 compared to the corresponding periods in fiscal 2011 remained relatively flat.

We are continuing our strategic investments in our Unified Access portfolio. We intend to continue to dedicate significant resources to research and development and to develop new product capabilities to support the performance, scalability and management of our Unified Access portfolio.

Sales and Marketing Expenses

The following table sets forth our sales and marketing expenses:

		Three Months Ended							Nine Months Ended							
	Se	eptember 29, 2012	S	September 24, 2011		ariance in Dollars	Variance in Percent	S	eptember 29, 2012	S	eptember 24, 2011		ariance in Dollars	Variance in Percent		
						(in	thousands, ex	cept	percentages)							
Sales and marketing	\$	15,093	\$	12,593	\$	2,500	20%	\$	44,880	\$	38,831	\$	6,049	16%		
Percent of total revenue		19%		15%					19%		15%					

Sales and marketing expenses increased during the three months ended September 29, 2012 compared with the corresponding period in fiscal 2011, primarily due to an increase in compensation and related costs driven by an increase in headcount resulting from the hiring of additional employees to pursue our international expansion.

Sales and marketing expenses increased during the nine months ended September 29, 2012 compared with the corresponding period in fiscal 2011, primarily due to an increase in compensation and related costs, an increase in travel-related

costs, which were driven by an increase in headcount resulting from our acquisition of Occam on February 22, 2011 and to pursue our international expansion.

The increases in sales and marketing expenses as a percentage of revenue for the three and nine months ended September 29, 2012 compared to the corresponding periods in fiscal 2011 were primarily due to the increase in expenses in the respective periods.

We will continue our investments in sales and marketing in order to extend our market reach and grow our business in support of our key strategic initiatives.

General and Administrative Expenses

The following table sets forth our general and administrative expenses:

		Three Months Ended						Nine Months Ended							
	Se	ptember 29, 2012	Se	September 24, 2011		ariance in Dollars	Variance in Percent	_	eptember 29, 2012	S	eptember 24, 2011	Variance in Dollars	Variance in Percent		
						(II	ı thousands, ex	cept	percentages)						
General and administrative	\$	6,773	\$	5,475	\$	1,298	24%	\$	19,682	\$	21,450	\$ (1,768)	(8)%		
Percent of total revenue		8%		7%					8%		8%				

General and administrative expenses increased during the three months ended September 29, 2012 compared with the corresponding period in fiscal 2011, primarily due to an increase in compensation and related costs resulting from increased headcount, as well as increases in consulting and professional service expenses.

General and administrative expenses decreased during the nine months ended September 29, 2012 compared with the corresponding period in fiscal 2011, primarily due to a decrease in stock-based compensation expense, offset in part by increases in professional and consulting service expenses as well as an increase in compensation expenses due to increased headcount.

General and administrative expenses as a percentage of revenue for the three and nine months ended September 29, 2012 compared to the corresponding periods in fiscal 2011 remained relatively flat.

Merger-related and Other Expenses

The following table sets forth our merger-related and other expenses included in operating expenses:

		Three Months	Ended		Nine Months Ended								
	September 29, 2012	September 24, 2011	Variance in Dollars	Variance in Percent	September 29, 2012	September 24, 2011	Variance in Dollars	Variance in Percent					
			(in	thousands, e	xcept percentages)								
Merger-related and other expenses	\$ —	\$ 1,404	\$ (1,404)	(100)%	\$ —	\$ 12,927	\$ (12,927)	(100)%					
Percent of total revenue	%	2%			%	5%							

In connection with our acquisition of Occam in the first quarter of 2011, we incurred operating merger-related and other expenses, which primarily consist of legal and professional expenses, severance for terminated Occam employees, and salaries paid to transitional Occam employees. In addition, we incurred expenses associated with consolidating facilities and stock-based compensation expense primarily related to accelerated vesting for certain Occam executives that terminated subsequent to the acquisition date. In connection with the planned acquisition of Ericsson fiber access assets, we expect to incur merger-related expenses in the fourth quarter of 2012.

Amortization of Intangible Assets

The following table sets forth our amortization of intangible asset expenses included in operating expenses:

		Three Months Ended						Nine Months Ended							
	Sej	otember 29, 2012	Se	ptember 24, 2011		in ollars	Variance in Percent	Se	eptember 29, 2012	Se	eptember 24, 2011		ariance in Dollars	Variance in Percent	
						(in	thousands, ex	xcept]	percentages)						
Amortization of intangible assets	\$	2,552	\$	2,552	\$	_	%	\$	7,656	\$	6,016	\$	1,640	27%	
Percent of total revenue		3%		3%					3%		2%				

In connection with the acquisition of Occam on February 22, 2011, \$51.0 million of the total purchase price was allocated to the amortizable intangible asset, customer relationships, which are being amortized to operating expenses over their estimated useful lives.

Liquidity and Capital Resources

We have funded our operations primarily through cash generated from operations and the 2010 initial public offering of our common stock. At September 29, 2012, we had cash and cash equivalents of \$57.4 million, which consisted of deposits held at banks and money market mutual funds held at major financial institutions. We also have a revolving credit facility of \$30.0 million based upon a percentage of eligible accounts receivable. Included in the revolving line are amounts available under letters of credit and cash management services.

Operating Activities

Our operating activities provided cash of \$24.8 million and \$8.7 million in the nine months ended September 29, 2012 and September 24, 2011, respectively. The increase in cash provided by operating activities was due primarily to a favorable change of \$19.0 million in our operating results after adjustment of non-cash charges, offset partially by a \$3.0 million decrease in net cash inflow resulting from changes in operating assets and liabilities.

In the nine months ended September 29, 2012, non-cash charges were \$32.7 million (the majority of which consist of depreciation and amortization expense and stock-based compensation expense). Cash inflows from changes in operating assets and liabilities primarily resulted from a \$14.4 million decrease in inventory due to improved inventory management, a \$15.8 million increase in deferred revenue as a result of increased shipments relating to certain RUS-funded contracts, and a \$2.2 million increase in accounts payable due to the timing of inventory receipts and payments. Cash outflows from changes in operating assets and liabilities included primarily an \$8.4 million increase in net accounts receivable due to the timing of sale and billing activities, an \$8.6 million increase in deferred cost of revenue primarily related to the deferral of certain RUS-funded contracts, and a \$2.1 million decrease in accrued liabilities.

Our operating activities provided cash of \$8.7 million in the nine months ended September 24, 2011. This resulted primarily from non-cash charges of \$39.2 million (the majority of which consist of stock-based compensation expense and depreciation and amortization expense) and positive net changes in operating assets and liabilities, largely offset by our net loss of \$47.3 million. Cash inflows from changes in operating assets and liabilities included a net decrease of \$12.3 million in accounts receivable due to strong cash collections, \$9.6 million related to the sell through of inventory, an increase in deferred revenue of \$5.8 million of certain RUS-funded contracts and an increase in accrued liabilities of \$2.9 million. These inflows were partially offset by cash outflows from accounts payable of \$10.1 million resulting primarily from payments of accounts payable assumed from Occam, an increase of \$2.3 million in prepaid and other current assets and an increase in deferred cost of revenue of \$1.2 million, primarily related to the deferral of revenue of certain RUS-funded contracts.

Investing Activities

Our investing activities used cash of \$7.9 million and \$37.3 million in the nine months ended September 29, 2012 and September 24, 2011, respectively.

Our cash used in investing activities in the nine months ended September 29, 2012 consisted of capital expenditures primarily as a result of purchases of computer equipment and software.

Our cash used in investing activities in the nine months ended September 24, 2011 primarily consisted of our acquisition of Occam for \$60.8 million, net of \$33.6 million of Occam cash assumed in the transaction, and capital expenditures of \$6.3 million, partially offset by maturities of marketable securities of \$29.8 million.

Financing Activities

Our financing activities provided cash of \$1.5 million in the nine months ended September 29, 2012, which consisted of proceeds of \$2.2 million from the issuance of common stock under the employee stock purchase plan ("ESPP") and proceeds of \$0.2 million from the exercises of stock options, offset by \$0.9 million payment of payroll taxes for the vesting of restricted stock units and restricted stock awards.

Our cash used in financing activities of \$7.5 million in the nine months ended September 24, 2011, primarily consisted of payment of payroll taxes for the vesting of restricted stock units of \$10.4 million, offset by proceeds of \$2.1 million from the issuance of common stock under the ESPP and proceeds of \$0.8 million from the exercise of stock options.

Working Capital and Capital Expenditure Needs

Other than our cash commitment in connection with our planned acquisition of Ericsson's fiber access assets as discussed in the section below, we currently have no material cash commitments, except for normal recurring trade payables, expense accruals, operating leases and firm purchase commitments. In addition, we believe that our outsourced approach to manufacturing provides us significant flexibility in both managing inventory levels and financing our inventory. We may be required to issue performance bonds to satisfy requirements under our RUS-funded contracts. We issue letters of credit under our existing credit facility to support these performance bonds. In the event we do not have sufficient capacity under our credit facility to support these bonds, we will have to purchase certificates of deposit, which could materially impact our working capital or limit our ability to satisfy such contract requirements. At December 31, 2011, we had cash of \$0.8 million restricted for the issuance of surety performance bonds we acquired through our acquisition of Occam. There were no restrictions on our cash at September 29, 2012. In the event that our revenue plan does not meet our expectations, we may eliminate or curtail expenditures to mitigate the impact on our working capital.

We believe based on our current operating plan, our existing cash, cash equivalents and existing amounts available under our revolving line of credit will be sufficient to meet our anticipated cash needs for at least the next twelve months. Our future capital requirements will depend on many factors including our rate of revenue growth, the timing and extent of spending to support development efforts, the expansion of sales and marketing activities, the timing of introductions of new products and enhancements to existing products, the acquisition of new capabilities or technologies and the continued market acceptance of our products. In the event that additional financing is required from outside sources, we may not be able to raise it on terms acceptable to us or at all. If we are unable to raise additional capital when desired, our business, operating results and financial condition would be harmed.

Contractual Obligations and Commitments

The Company's principal commitments consist of obligations under operating leases for office space and non-cancelable outstanding purchase obligations. These commitments as of December 31, 2011 are disclosed in our Annual Report on Form 10-K for the year ended December 31, 2011, and have not changed materially during the nine months ended September 29, 2012.

In addition, on August 20, 2012, Calix and Ericsson Inc. ("Ericsson") entered into an Asset Purchase Agreement under which Calix will make a one-time cash payment to acquire Ericsson's fiber access assets, including the Ericsson EDA 1500 GPON solution and its complementary ONT portfolio. The transaction is expected to close in the fourth quarter of 2012.

Off-Balance Sheet Arrangements

As of September 29, 2012 and December 31, 2011, we did not have any off-balance sheet arrangements.

ITEM 3. Quantitative and Qualitative Disclosures About Market Risk

Interest Rate Risk

The primary objectives of our investment activity are to preserve principal, provide liquidity and maximize income without significantly increasing risk. By policy, we do not enter into investments for trading or speculative purposes. At September 29, 2012, we had cash and cash equivalents of \$57.4 million, which was held primarily in cash or money market funds. Due to the nature of these money-market funds, we believe that we do not have any material exposure to changes in the fair value of our cash equivalents as a result of changes in interest rates.

Our exposure to interest rate risk also relates to the amount of interest we must pay on our outstanding debt instruments. Any outstanding borrowings under our revolving credit facility bear a variable rate of interest based upon the applicable LIBOR or prime rate and is adjusted monthly based upon changes in the Federal Reserve's prime rate. As of September 29,

2012, we had no outstanding borrowings under the revolving credit facility.

Foreign Currency Exchange Risk

Our sales contracts and vendor payables are primarily denominated in U.S. dollars and, therefore, the majority of our revenues and operating expenses are not subject to foreign currency exchange risk.

ITEM 4. Controls and Procedures

Evaluation of Disclosure Controls and Procedures

Based on their evaluation as of September 29, 2012, our Chief Executive Officer and Chief Financial Officer, with the participation of our management, have concluded that our disclosure controls and procedures (as defined in Rules 13a–15(e) and 15d–15(e) of the Securities Exchange Act of 1934) were effective at the reasonable assurance level.

Limitations on the Effectiveness of Controls

Our disclosure controls and procedures provide our Chief Executive Officer and Chief Financial Officer reasonable assurances that our disclosure controls and procedures will achieve their objectives. However, our management, including our Chief Executive Officer and Chief Financial Officer, does not expect that our disclosure controls and procedures or our internal control over financial reporting can or will prevent all human error. A control system, no matter how well designed and implemented, can provide only reasonable, not absolute, assurance that the objectives of the control system are met. Furthermore, the design of a control system must reflect the fact that there are internal resource constraints, and the benefit of controls must be weighed relative to their corresponding costs. Because of the limitations in all control systems, no evaluation of controls can provide complete assurance that all control issues and instances of error, if any, within our company are detected. These inherent limitations include the realities that judgments in decision-making can be faulty, and that breakdowns can occur due to human error or mistake. Additionally, controls, no matter how well designed, could be circumvented by the individual acts of specific persons within the organization. The design of any system of controls is also based in part upon certain assumptions about the likelihood of future events, and there can be no assurance that any design will succeed in achieving its stated objectives under all potential future conditions.

Changes in Internal Control over Financial Reporting

There was no change in our internal control over financial reporting identified in connection with the evaluation required by Rule 13a-15(d) and 15d-15(d) of the Exchange Act that occurred during the period covered by this report that has materially affected, or is reasonably likely to materially affect, our internal control over financial reporting.

PART II. OTHER INFORMATION

ITEM 1. Legal Proceedings

For a description of our material pending legal proceedings, please refer to Note 7 "Commitments and Contingencies – Litigation" of the Notes to Condensed Consolidated Financial Statements included in Part I, Item 1 of this Quarterly Report on Form 10-Q, which is incorporated by reference.

ITEM 1A. Risk Factors

We have identified the following additional risks and uncertainties that may affect our business, financial condition and/or results of operations. The risks described below include any material changes to and supersede the description of the risk factors disclosed in Part I, Item 1A of our Annual Report on Form 10-K for the year ended December 31, 2011, as filed with the Securities and Exchange Commission on February 24, 2012. Investors should carefully consider the risks described below, together with the other information set forth in this Quarterly Report on Form 10-Q, before making any investment decision. The risks described below are not the only ones we face. Additional risks not presently known to us or that we currently believe are immaterial may also significantly impair our business operations. Our business could be harmed by any of these risks. The trading price of our common stock could decline due to any of these risks, and investors may lose all or part of their investment.

Risks Related to Our Business and Industry

Our markets are rapidly changing and we have a limited operating history, which make it difficult to predict our future revenue and plan our expenses appropriately.

We were incorporated in August 1999 and shipped our first product in December 2001. We have a limited operating history and compete in markets characterized by rapid technological change, changing needs of communications service providers, or CSPs, evolving industry standards and frequent introductions of new products and services. We have limited historical data and have had a relatively limited time period in which to implement and evaluate our business strategies as compared to companies with longer operating histories. In addition, we likely will be required to reposition our product and service offerings and introduce new products and services as we encounter rapidly changing CSP requirements and increasing competitive pressures. We may not be successful in doing so in a timely and responsive manner, or at all. Also, softness in demand across any of our customer markets, including due to macroeconomic conditions beyond our control or uncertainties associated with the implementation of regulatory reforms, could lead to unexpected slowdown in capital expenditures by service providers, such as what occurred in the second quarter of 2012. As a result, it is difficult to forecast our future revenues and plan our operating expenses appropriately, which also makes it difficult to predict our future operating results.

We have a history of losses, and we may not be able to generate positive operating income and cash flows in the future.

We have experienced net losses in each year of our existence. For the year ended December 31, 2011, December 31, 2010, and December 31, 2009, we incurred net losses of \$52.6 million, \$18.6 million, and \$22.4 million, respectively. For the nine months ended September 29, 2012, we incurred a net loss of \$21.8 million. As of September 29, 2012, we had an accumulated deficit of \$485.9 million.

We expect to continue to incur significant expenses for research and development, sales and marketing, customer support and general and administrative functions as we expand our operations. Given our rapid growth rate and the intense competitive pressures we face, we may be unable to control our operating costs.

We cannot guarantee that we will achieve profitability in the future. We will have to generate and sustain significant and consistent increased revenue, while continuing to control our expenses, in order to achieve and then maintain profitability. We may also incur significant losses in the future for a number of reasons, including the risks discussed in this "Risk Factors" section and factors that we cannot anticipate. If we are unable to generate positive operating income and cash flow from operations, our liquidity, results of operations and financial condition will be adversely affected.

Fluctuations in our quarterly and annual operating results may make it difficult to predict our future performance, which could cause our operating results to fall below investor or analyst expectations, which could adversely affect the trading price of our stock.

A number of factors, many of which are outside of our control, may cause or contribute to significant fluctuations in our quarterly and annual operating results. These fluctuations may make financial planning and forecasting difficult. Comparing our operating results on a period-to-period basis may not be meaningful, and you should not rely on our past results as an indication of our future performance. If our revenue or operating results fall below the expectations of investors or securities analysts, or below any guidance we may provide to the market, the price of our common stock would likely decline. Moreover.

we may experience delays in recognizing revenue under applicable revenue recognition rules, particularly from government-funded contracts, such as those funded by RUS. The extent of these delays and their impact on our revenues can fluctuate over a given time period depending on the number and size of purchase orders under these contracts during such time period. In addition, unanticipated decreases in our available liquidity due to fluctuating operating results could limit our growth and delay implementation of our expansion plans.

In addition to the other risk factors listed in this "Risk Factors" section, factors that may contribute to the variability of our operating results include:

- our ability to predict our revenue and plan our expenses appropriately;
- the capital spending patterns of CSPs and any decrease or delay in capital spending by CSPs due to macro-economic conditions, regulatory
 implementation or uncertainties, or other reasons;
- · the impact of government-sponsored programs on our customers;
- intense competition;
- our ability to develop new products or enhancements that support technological advances and meet changing CSP requirements;
- our ability to achieve market acceptance of our products and CSPs' willingness to deploy our new products;
- the concentration of our customer base;
- · the length and unpredictability of our sales cycles;
- our focus on CSPs with limited revenue potential;
- our lack of long-term, committed-volume purchase contracts with our customers;
- our ability to increase our sales to larger North American as well as international CSPs;
- our exposure to the credit risks of our customers;
- fluctuations in our gross margin;
- the interoperability of our products with CSP networks;
- our dependence on sole and limited source suppliers;
- our ability to manage our relationships with our contract manufacturers;
- our ability to forecast our manufacturing requirements and manage our inventory;
- our products' compliance with industry standards;
- our ability to expand our international operations;
- our ability to protect our intellectual property and the cost of doing so;
- the quality of our products, including any undetected hardware errors or bugs in our software;
- our ability to estimate future warranty obligations due to product failure rates;
- our ability to obtain necessary third-party technology licenses;
- any obligation to issue performance bonds to satisfy requirements under RUS contracts;
- the attraction and retention of qualified employees and key personnel; and
- our ability to maintain proper and effective internal controls.

Our business is dependent on the capital spending patterns of CSPs, and any decrease or delay in capital spending by CSPs, in response to economic conditions, uncertainties associated with the implementation of regulatory reforms, or otherwise, would reduce our revenues and harm our business.

Demand for our products depends on the magnitude and timing of capital spending by CSPs as they construct, expand and upgrade their access networks. For the year ended December 31, 2011, CenturyLink, Inc., or CenturyLink, purchased a significant amount of our access systems and software. However, we cannot anticipate the level of CenturyLink's purchases in the future. On April 1, 2011, CenturyLink completed their merger with Qwest Communications. This merger could create uncertainty for us as to whether we will continue to be chosen as a preferred network equipment vendor for the combined organization. In addition, the recent economic downturn has contributed to a slowdown in telecommunications industry spending, including in the specific geographies and markets in which we operate. In response to reduced consumer spending, challenging capital markets or declining liquidity trends, capital spending for network infrastructure projects of CSPs could be delayed or cancelled. In addition, capital spending is cyclical in our industry and sporadic among individual CSPs, and can change on short notice. As a result, we may not have visibility into changes in spending behavior until nearly the end of a given quarter. CSP spending on network construction, maintenance, expansion and upgrades is also affected by seasonality in their purchasing cycles, reductions in their budgets and delays in their purchasing cycles. In addition, we believe the capital

expenditures amongst some CSPs have also been adversely affected due to recent fiber shortages in certain portions of the market. Many factors affecting our results of operations are beyond our control, particularly in the case of large CSP orders and network infrastructure deployments involving multiple vendors and technologies where the achievement of certain thresholds for acceptance is subject to the readiness and performance of the customer or other providers, and changes in customer requirements or installation plans. Further, CSPs may not pursue infrastructure upgrades that require our access systems and software. Infrastructure improvements may be delayed or prevented by a variety of factors including cost, regulatory obstacles (including uncertainties associated with the implementation of regulatory reforms), mergers, lack of consumer demand for advanced communications services and alternative approaches to service delivery. Reductions in capital expenditures by CSPs may slow our rate of revenue growth. As a consequence, our results for a particular quarter may be difficult to predict, and our prior results are not necessarily indicative of results likely in future periods.

Government-sponsored programs could impact the timing and buying patterns of CSPs, which may cause fluctuations in our operating results.

Many of our customers are Independent Operating Companies, or IOCs, which have revenues that are particularly dependent upon interstate and intrastate access charges, and federal and state subsidies. The Federal Communications Commission, or FCC, and some states are considering changes to such payments and subsidies, and these changes could reduce IOC revenues. Furthermore, many IOCs use or expect to use, government-supported loan programs or grants, such as RUS loans and grants and the Broadband Stimulus programs under the American Recovery and Reinvestment Act of 2009, or ARRA, to finance capital spending. Changes to these programs could reduce the ability of IOCs to access capital and reduce our revenue opportunities.

To the extent that our customers do receive grants or loans under these stimulus programs, our customers may be encouraged to accelerate their network development plans and purchase substantial quantities of products, from us or other suppliers, while the programs and funding are in place. Customers may thereafter substantially curtail future purchases of products as ARRA funding winds down or because all purchases have been completed. Award grants under the Broadband Stimulus programs have been issued between December 2009 and September 2010. The timetable for completion of funded projects varies between the two agencies administering the awards. Projects funded under the Broadband Technology Opportunities Program (BTOP), which is administered by the National Telecommunications and Information Administration (NTIA), must be completed by September 30, 2013. Projects funded under the Broadband Initiatives Program (BIP), which is administered by the Rural Utilities Service, must be completed by June 30, 2015.

We have experienced continued delays in purchasing commitments from our customers who have been awarded Broadband Stimulus funds, which have negatively impacted our operating results and additional delays could continue to adversely impact our operating results. In addition, the revenue recognition guidelines related to the sales of our access systems to CSPs who have received Broadband Stimulus funds may create uncertainties around the timing of our revenue, which could harm our financial results. In addition, any changes in government regulations and subsidies could cause our customers to change their purchasing decisions which could have an adverse effect on our operating results and financial condition.

We face intense competition that could reduce our revenue and adversely affect our financial results.

The market for our products is highly competitive, and we expect competition from both established and new companies to increase. Our competitors include companies such as ADTRAN, Inc., Alcatel- Lucent S.A., Ciena Corporation, Huawei Technologies Co., Ltd., Tellabs, Inc., ZTE Corporation.

Our ability to compete successfully depends on a number of factors, including:

- the successful development of new products;
- our ability to anticipate CSP and market requirements and changes in technology and industry standards;
- our ability to differentiate our products from our competitors' offerings based on performance, cost-effectiveness or other factors;
- · our ability to gain customer acceptance of our products; and
- our ability to market and sell our products.

The market for broadband access equipment is dominated primarily by large, established vendors. In addition, some of our competitors have merged, made acquisitions or entered into partnerships or other strategic relationships with one another to offer more comprehensive solutions than they individually had offered. Examples include Ciena Corporation's acquisitions of World Wide Packets, Inc. in 2008 and Nortel's Metro Ethernet Networks business in March 2010, Enablence Technologies, Inc.'s acquisition of Teledata Networks, Ltd. in June 2010, our acquisitions of Occam in February 2011 and of Ericsson's fiber access business expected to close in the fourth quarter of 2012, and Adtran's acquisition of Nokia Siemens' broadband access line business in May 2012. We expect this trend to continue as companies attempt to strengthen or maintain their market

positions in an evolving industry. Many of our current or potential competitors have longer operating histories, greater name recognition, larger customer bases and significantly greater financial, technical, sales, marketing and other resources than we do and are better positioned to acquire and offer complementary products and services technologies. Many of our competitors have broader product lines and can offer bundled solutions, which may appeal to certain customers. Our competitors may invest additional resources in developing more compelling product offerings. Potential customers may also prefer to purchase from their existing suppliers rather than a new supplier, regardless of product performance or features, because the products that we and our competitors offer require a substantial investment of time and funds to install. In addition, as a result of these transition costs, competition to secure contracts with potential customers is particularly intense. Some of our competitors may offer substantial discounts or rebates to win new customers. If we are forced to reduce prices in order to secure customers, we may be unable to sustain gross margins at desired levels or achieve profitability. Competitive pressures could result in increased pricing pressure, reduced profit margins, increased sales and marketing expenses and failure to increase, or the loss of, market share, any of which could reduce our revenue and adversely affect our financial results.

Product development is costly and if we fail to develop new products or enhancements that meet changing CSP requirements, we could experience lower sales.

Our market is characterized by rapid technological advances, frequent new product introductions, evolving industry standards and unanticipated changes in subscriber requirements. Our future success will depend significantly on our ability to anticipate and adapt to such changes, and to offer, on a timely and cost-effective basis, products and features that meet changing CSP demands and industry standards.

We intend to continue making significant investments in developing new products and enhancing the functionality of our existing products. Developing our products is expensive, complex and involves uncertainties. We may not have sufficient resources to successfully manage lengthy product development cycles. For the years ended December 31, 2011, 2010 and 2009, our research and development expenses were \$67.7 million, or 20% of our revenue, \$55.4 million, or 19% of our revenue, and \$46.1 million or 20% of our revenue, respectively. For the nine months ended September 29, 2012, our research and development expenses were \$49.6 million, or 21% of our revenue. We believe that we must continue to dedicate a significant amount of resources to our research and development efforts to maintain our competitive position. These investments may take several years to generate positive returns, if ever. In addition, we may experience design, manufacturing, marketing and other difficulties that could delay or prevent the development, introduction or marketing of new products and enhancements. If we fail to meet our development targets, demand for our products will decline.

In addition, the introduction of new or enhanced products also requires that we manage the transition from older products to these new or enhanced products in order to minimize disruption in customer ordering patterns, fulfill ongoing customer commitments and ensure that adequate supplies of new products are available for delivery to meet anticipated customer demand. If we fail to maintain compatibility with other software or equipment found in our customers' existing and planned networks, we may face substantially reduced demand for our products, which would reduce our revenue opportunities and market share. Moreover, as customers complete infrastructure deployments, they may require greater levels of service and support than we have provided in the past. We may not be able to provide products, services and support to compete effectively for these market opportunities. If we are unable to anticipate and develop new products or enhancements to our existing products on a timely and cost-effective basis, we could experience lower sales which would harm our business.

Our new products are early in their life cycles and are subject to uncertain market demand. If our customers are unwilling to install our products or deploy new services or we are unable to achieve market acceptance of our new products, our business and financial results will be harmed.

Our new products are early in their life cycles and are subject to uncertain market demand. They also may face obstacles in manufacturing, deployment and competitive response. Potential customers may choose not to invest the additional capital required for initial system deployment. In addition, demand for our products is dependent on the success of our customers in deploying and selling services to their subscribers. Our products support a variety of advanced broadband services, such as high-speed Internet, Internet protocol television, mobile broadband, high-definition video and online gaming, and basic voice and data services. If subscriber demand for such services does not grow as expected or declines, or if our customers are unable or unwilling to deploy and market these services, demand for our products may decrease or fail to grow at rates we anticipate.

Our customer base is concentrated, and there are a limited number of potential customers for our products. The loss of any of our key customers, a decrease in purchases by our key customers or our inability to grow our customer base would adversely impact our revenues.

Historically, a large portion of our sales has been to a limited number of customers. For example, for the years ended December 31, 2011, 2010 and 2009, CenturyLink accounted for 20%, 29% and 38%, respectively, of our revenue.

We anticipate that a large portion of our revenues will continue to depend on sales to a limited number of customers. In addition, some larger customers may demand discounts and rebates or desire to purchase their access systems and software from multiple providers. As a result of these factors, our future revenue opportunities may be limited and our margins could be reduced, and our profitability may be adversely impacted. The loss of, or reduction in, orders from any key customer would significantly reduce our revenues and harm our business.

Furthermore, in recent years, the CSP market has undergone substantial consolidation. Industry consolidation generally has negative implications for equipment suppliers, including a reduction in the number of potential customers, a decrease in aggregate capital spending, and greater pricing leverage on the part of CSPs over equipment suppliers. Continued consolidation of the CSP industry, including the merger between CenturyLink and Qwest Communications, which closed in April 2011, and among the Incumbent Local Exchange Carrier, or ILEC, and IOC customers, who represent a large part of our business, could make it more difficult for us to grow our customer base, increase sales of our products and maintain adequate gross margins.

Our sales cycles can be long and unpredictable, and our sales efforts require considerable time and expense. As a result, our sales are difficult to predict and may vary substantially from quarter to quarter, which may cause our operating results to fluctuate significantly.

The timing of our revenues is difficult to predict. Our sales efforts often involve educating CSPs about the use and benefits of our products. CSPs typically undertake a significant evaluation process, which frequently involves not only our products but also those of our competitors and results in a lengthy sales cycle. We spend substantial time, effort and money in our sales efforts without any assurance that our efforts will produce any sales. In addition, product purchases are frequently subject to budget constraints, multiple approvals and unplanned administrative, processing and other delays. If sales expected from a specific customer for a particular quarter are not realized in that quarter or at all, we may not achieve our revenue forecasts and our financial results would be adversely affected.

Our focus on CSPs with relatively small networks limits our revenues from sales to any one customer and makes our future operating results difficult to predict.

We currently focus a large portion of our sales efforts on IOCs, cable multiple system operators and selected international CSPs. In general, our current and potential customers generally operate small networks with limited capital expenditure budgets. Accordingly, we believe the potential revenues from the sale of our products to any one of these customers is limited. As a result, we must identify and sell products to new customers each quarter to continue to increase our sales. In addition, the spending patterns of many of our customers are characterized by small and sporadic purchases. As a consequence, we have limited backlog and will likely continue to have limited visibility into future operating results.

We do not have long-term, committed-volume purchase contracts with our customers, and therefore have no guarantee of future revenues from any customer.

Our sales are made predominantly pursuant to purchase orders, and typically we have not entered into long-term, committed-volume purchase contracts with our customers, including our key customers which account for a material portion of our revenues. As a result, any of our customers may cease to purchase our products at any time. In addition, our customers may attempt to renegotiate the terms of our agreements, including price and quantity. If any of our key customers stop purchasing our access systems and software for any reason, our business and results of operations would be harmed.

Our efforts to increase our sales to larger North American as well as international CSPs, including MSOs, may be unsuccessful.

Our sales and marketing efforts have been focused on Communication Service Providers (CSPs), including Multi System Operators ("MSOs"), in North America. A part of our long-term strategy is to increase sales to larger North American as well as international CSPs, including MSOs. We will be required to devote substantial technical, marketing and sales resources to the pursuit of these CSPs, who have lengthy equipment qualification and sales cycles, without any assurance of generating sales. In particular, sales to these CSPs may require us to upgrade our products to meet more stringent performance criteria, develop new customer-specific features or adapt our product to meet international standards. If we are unable to successfully increase our sales to larger CSPs, our operating results and long-term growth may be negatively impacted.

We are exposed to the credit risks of our customers, and if we have inadequately assessed their credit we may have more exposure to accounts receivable risk than we anticipate, and failure to collect our accounts receivable in amounts that we anticipate could adversely affect our operating results and financial condition.

In the course of our sales to customers, we may encounter difficulty collecting accounts receivable and could be exposed to risks associated with uncollectible accounts receivable. We maintain an allowance for doubtful accounts for estimated losses

resulting from the inability or unwillingness of our customers to make required payments. However, these allowances are based on our judgment and a variety of factors about which our judgment may be wrong or that may change. For example, we perform credit evaluations of our customers' financial condition. Our evaluation of the creditworthiness of customers may not be accurate if they do not provide us with accurate financial information, or if their situation changes since the time we last evaluated their credit. While we attempt to monitor these situations carefully and attempt to adjust our allowances for doubtful accounts as appropriate, and take appropriate measures to collect accounts receivable balances, we have written down accounts receivable and written off doubtful accounts in prior periods and may be unable to avoid additional write-downs or write-offs of doubtful accounts in the future. Such write-downs or write-offs could negatively affect our operating results for the period in which they occur, and could harm our operating results.

Our gross margin may fluctuate over time and our current level of product gross margins may not be sustainable.

Our current level of product gross margins may not be sustainable and may be adversely affected by numerous factors, including:

- changes in customer, geographic or product mix, including the mix of configurations within each product group;
- increased price competition, including the impact of customer discounts and rebates;
- our ability to reduce and control product costs;
- loss of cost savings due to changes in component pricing or charges incurred due to inventory holding periods if parts ordering does not
 correctly anticipate product demand;
- · introduction of new products;
- · changes in shipment volume;
- · changes in distribution channels;
- increased warranty costs;
- excess and obsolete inventory and inventory holding charges;
- · expediting costs incurred to meet customer delivery requirements; and
- liquidated damages relating to customer contractual terms.

Our products must interoperate with many software applications and hardware products found in our customers' networks. If we are unable to ensure that our products interoperate properly, our business would be harmed.

Our products must interoperate with our customers' existing and planned networks, which often have varied and complex specifications, utilize multiple protocol standards, software applications and products from multiple vendors and contain multiple generations of products that have been added over time. As a result, we must continually ensure that our products interoperate properly with these existing and planned networks. To meet these requirements, we must undertake development efforts that require substantial capital investment and employee resources. We may not accomplish these development efforts quickly or cost-effectively, if at all. If we fail to maintain compatibility with other software or equipment found in our customers' existing and planned networks, we may face substantially reduced demand for our products, which would reduce our revenue opportunities and market share.

We have entered into interoperability arrangements with a number of equipment and software vendors for the use or integration of their technology with our products. These arrangements give us access to, and enable interoperability with, various products that we do not otherwise offer. If these relationships fail, we may have to devote substantially more resources to the development of alternative products and processes, and our efforts may not be as effective as the combined solutions under our current arrangements. In some cases, these other vendors are either companies that we compete with directly, or companies that have extensive relationships with our existing and potential customers and may have influence over the purchasing decisions of those customers. Some of our competitors have stronger relationships with some of our existing and potential other vendors and, as a result, our ability to have successful interoperability arrangements with these companies may be harmed. Our failure to establish or maintain key relationships with third-party equipment and software vendors may harm our ability to successfully sell and market our products.

As we do not have manufacturing capabilities, we depend upon a small number of outside contract manufacturers and we do not have supply contracts with these manufacturers. Our operations could be disrupted if we encounter problems with these contract manufacturers.

We do not have internal manufacturing capabilities, and rely upon a small number of contract manufacturers to build our products. In particular, we rely on Flextronics International Ltd., or Flextronics, for the manufacture of most of our products. Our reliance on a small number of contract manufacturers makes us vulnerable to possible capacity constraints and reduced control over component availability, delivery schedules, manufacturing yields and costs. We do not have supply contracts with

Flextronics or our other manufacturers. Consequently, these manufacturers are not obligated to supply products to us for any specific period, in any specific quantity or at any certain price. In addition, we have limited control over our contract manufacturers' quality systems and controls, and therefore may not be able to ensure levels of quality manufacture suitable for our customers.

The revenues that Flextronics generates from our orders represent a relatively small percentage of Flextronics' overall revenues. As a result, fulfilling our orders may not be considered a priority in the event Flextronics is constrained in its ability to fulfill all of its customer obligations in a timely manner. In addition, a substantial part of our manufacturing is done in Flextronics facilities that are located outside of the United States. We believe that the location of these facilities outside of the United States increases supply risk, including the risk of supply interruptions or reductions in manufacturing quality or controls. If Flextronics or any of our other contract manufacturers were unable or unwilling to continue manufacturing our products in required volumes and at high quality levels, we would have to identify, qualify and select acceptable alternative contract manufacturers. An alternative contract manufacturer may not be available to us when needed or may not be in a position to satisfy our production requirements at commercially reasonable prices and quality. Any significant interruption in manufacturing would require us to reduce our supply of products to our customers, which in turn would reduce our revenues and harm our relationships with our customers.

We depend on sole source and limited source suppliers for key components and products. If we are unable to source these components on a timely basis, we will not be able to deliver our products to our customers.

We depend on sole source and limited source suppliers for key components of our products. For example, certain of our application-specific integrated circuits processors and resistor networks are purchased from sole source suppliers. We may from time to time enter into original equipment manufacturer, or OEM, or original design manufacturer, or ODM, agreements to manufacture and/or design certain products in order to enable us to offer products into key markets on an accelerated basis. For example, a third party assisted in the design and manufacture of our E5-100 platform family. Any of the sole source and limited source suppliers, OEMs and ODMs upon whom we rely could stop producing our components or products, cease operations or be acquired by, or enter into exclusive arrangements with, our competitors. We generally purchase our products through purchase orders and our purchase volumes are currently too low for us to be considered a priority customer by most of our suppliers. As a result, most of these suppliers could stop selling to us at commercially reasonable prices, or at all. Any such interruption or delay may force us to seek similar components or products from alternative sources, which may not be available. Switching suppliers, OEMs or ODMs may require that we redesign our products to accommodate new components, and may potentially require us to re-qualify our products with our customers, which would be costly and time-consuming. Any interruption in the supply of sole source or limited source components for our products would adversely affect our ability to meet scheduled product deliveries to our customers, could result in lost revenue or higher expenses and would harm our business.

Although we do not have manufacturing facilities in Japan, a small number of Japanese factories produce some of the components we use in our products and other components that our customers use that are required to be secured as a precursor to buying our products. A few of these manufacturers have reported disruptions in to their production because of damage to their facilities as a result of the natural disasters in March 2011 and their aftermath in Japan. Although most of these factories are back on line, such interruptions or delays may force us or our customers to seek similar components from alternative sources, which may not be available at favorable prices, or at all. Interruptions in supply resulting from such unforeseen natural disasters could result in lost revenue or higher expenses and would harm our business.

If we fail to forecast our manufacturing requirements accurately and manage our inventory with our contract manufacturers, we could incur additional costs, experience manufacturing delays and lose revenue.

We bear inventory risk under our contract manufacturing arrangements. Lead times for the materials and components that we order through our contract manufacturers vary significantly and depend on numerous factors, including the specific supplier, contract terms and market demand for a component at a given time. Lead times for certain key materials and components incorporated into our products are currently lengthy, requiring us or our contract manufacturers to order materials and components several months in advance of manufacture. If we overestimate our production requirements, we or our contract manufacturers may purchase excess components and build excess inventory. If our contract manufacturers, at our request, purchase excess components that are unique to our products or build excess products, we could be required to pay for these excess parts or products and recognize related inventory write-down costs. Historically, we have reimbursed our primary contract manufacturer for a portion of inventory purchases when our inventory has been rendered obsolete, for example due to manufacturing and engineering change orders resulting from design changes manufacturing discontinuation of parts by our suppliers, or in cases where inventory levels greatly exceed projected demand. If we experience inventory write-downs associated with excess or obsolete inventory, this would have an adverse effect on our gross margins, financial condition and results of operations. We have experienced unanticipated increases in demand from customers, which resulted in delayed

shipments and variable shipping patterns. If we underestimate our product requirements, our contract manufacturers may have inadequate component inventory, which could interrupt manufacturing of our products and result in delays or cancellation of sales.

If we fail to comply with evolving industry standards, sales of our existing and future products would be adversely affected.

The markets for our products are characterized by a significant number of standards, both domestic and international, which are evolving as new technologies are deployed. Our products must comply with these standards in order to be widely marketable. In some cases, we are compelled to obtain certifications or authorizations before our products can be introduced, marketed or sold in new markets or to customers that we have not historically served. For example, our ability to obtain OSMINE certification for our products will affect our ongoing ability to sell our products to CenturyLink and other Tier 1 CSPs. In addition, our ability to expand our international operations and create international market demand for our products may be limited by regulations or standards adopted by other countries that may require us to redesign our existing products or develop new products suitable for sale in those countries. Although we believe our products are currently in compliance with domestic and international standards and regulations in countries in which we currently sell, we may not be able to design our products to comply with evolving standards and regulations in the future. Accordingly, this ongoing evolution of standards may directly affect our ability to market or sell our products. Further, the cost of complying with the evolving standards and regulations, or the failure to obtain timely domestic or foreign regulatory approvals or certification such that we may not be able to sell our products where these standards or regulations apply, would result in lower revenues and lost market share.

We may be unable to successfully expand our international operations. In addition, we may be subject to a variety of risks that could harm our business.

We currently generate most of our sales from customers in North America and have limited experience marketing, selling and supporting our products and services outside North America or managing the administrative aspects of a worldwide operation. While we are in the process of expanding our international operations, we may not be able to create or maintain international market demand for our products. In addition, as we expand our operations internationally, our support organization will face additional challenges including those associated with delivering support, training and documentation in languages other than English. If we invest substantial time and resources to expand our international operations and are unable to do so successfully and in a timely manner, our business, financial condition and results of operations will suffer.

In the course of expanding our international operations and operating overseas, we will be subject to a variety of risks, including:

- differing regulatory requirements, including tax laws, trade laws, labor regulations, tariffs, export quotas, custom duties or other trade restrictions;
- greater difficulty supporting and localizing our products;
- · different or unique competitive pressures as a result of, among other things, the presence of local equipment suppliers;
- challenges inherent in efficiently managing an increased number of employees over large geographic distances, including the need to implement appropriate systems, policies, compensation and benefits and compliance programs;
- limited or unfavorable intellectual property protection;
- · risk of change in international political or economic conditions; and
- restrictions on the repatriation of earnings.

We may have difficulty managing our growth, which could limit our ability to increase sales.

We have experienced significant growth in sales and operations in recent years. We expect to continue to expand our research and development, sales, marketing and support activities. Our historical growth has placed, and planned future growth is expected to continue to place, significant demands on our management, as well as our financial and operational resources, to:

- manage a larger organization;
- · expand our manufacturing and distribution capacity;
- · increase our sales and marketing efforts;
- broaden our customer support capabilities;
- · implement appropriate operational and financial systems; and
- maintain effective financial disclosure controls and procedures.

If we cannot grow, or fail to manage our growth effectively, we may not be able to execute our business strategies and our business, financial condition and results of operations would be adversely affected.

We may not be able to protect our intellectual property, which could impair our ability to compete effectively.

We depend on certain proprietary technology for our success and ability to compete. As of September 29, 2012, we held 67 U.S. patents and had 39 pending U.S. patent applications. Two of the U.S. patents are also covered by granted international patents, one in five countries and the other in three countries. We currently have no pending international patent applications. We rely on intellectual property laws, as well as nondisclosure agreements, licensing arrangements and confidentiality provisions, to establish and protect our proprietary rights. U.S. patent, copyright and trade secret laws afford us only limited protection, and the laws of some foreign countries do not protect proprietary rights to the same extent. Our pending patent applications may not result in issued patents, and our issued patents may not be enforceable. Any infringement of our proprietary rights could result in significant litigation costs. Further, any failure by us to adequately protect our proprietary rights could result in our competitors offering similar products, resulting in the loss of our competitive advantage and decreased sales.

Despite our efforts to protect our proprietary rights, attempts may be made to copy or reverse engineer aspects of our products or to obtain and use information that we regard as proprietary. Accordingly, we may be unable to protect our proprietary rights against unauthorized third-party copying or use. Furthermore, policing the unauthorized use of our intellectual property would be difficult for us. Litigation may be necessary in the future to enforce our intellectual property rights, to protect our trade secrets or to determine the validity and scope of the proprietary rights of others. Litigation could result in substantial costs and diversion of resources and could harm our business.

We could become subject to litigation regarding intellectual property rights that could harm our business.

We may be subject to intellectual property infringement claims that are costly to defend and could limit our ability to use some technologies in the future. Third parties may assert patent, copyright, trademark or other intellectual property rights to technologies or rights that are important to our business. Such claims may involve patent holding companies or other adverse patent owners who have no relevant product revenue, and therefore our own issued and pending patents may provide little or no deterrence. We have received in the past and expect that in the future we may receive, particularly as a public company, communications from competitors and other companies alleging that we may be infringing their patents, trade secrets or other intellectual property rights and/or offering licenses to such intellectual property or threatening litigation. In addition, we have agreed, and may in the future agree, to indemnify our customers for any expenses or liabilities resulting from claimed infringements of patents, trademarks or copyrights of third parties. Any claims asserting that our products infringe, or may infringe on, the proprietary rights of third parties, with or without merit, could be time-consuming, resulting in costly litigation and diverting the efforts of our engineering teams and management. These claims could also result in product shipment delays or require us to modify our products or enter into royalty or licensing agreements. Such royalty or licensing agreements, if required, may not be available to us on acceptable terms, if at all.

The quality of our support and services offerings is important to our customers, and if we fail to continue to offer high quality support and services, we could lose customers, which would harm our business.

Once our products are deployed within our customers' networks, they depend on our support organization to resolve any issues relating to those products. A high level of support is critical for the successful marketing and sale of our products. If we do not effectively assist our customers in deploying our products, succeed in helping them quickly resolve post-deployment issues or provide effective ongoing support, it could adversely affect our ability to sell our products to existing customers and harm our reputation with potential new customers. As a result, our failure to maintain high quality support and services could result in the loss of customers, which would harm our business.

Our products are highly technical and may contain undetected hardware errors or software bugs, which could harm our reputation and adversely affect our business.

Our products are highly technical and, when deployed, are critical to the operation of many networks. Our products have contained and may contain undetected errors, bugs or security vulnerabilities. Some errors in our products may only be discovered after a product has been installed and used by customers, and may in some cases only be detected under certain circumstances or after extended use. Any errors, bugs, defects or security vulnerabilities discovered in our products after commercial release could result in loss of revenues or delay in revenue recognition, loss of customer goodwill and customers and increased service and warranty cost, any of which could adversely affect our business, operating results and financial condition. In addition, we could face claims for product liability, tort or breach of warranty. Our contracts with customers contain provisions relating to warranty disclaimers and liability limitations, which may not be upheld. Defending a lawsuit, regardless of its merit, is costly and may divert management's attention and adversely affect the market's perception of us and

our products. In addition, if our business liability insurance coverage proves inadequate or future coverage is unavailable on acceptable terms or at all, our business, operating results and financial condition could be adversely impacted.

Our estimates regarding future warranty obligations may change due to product failure rates, shipment volumes, field service obligations and rework costs incurred in correcting product failures. If our estimates change, the liability for warranty obligations may be increased, impacting future cost of revenue.

Our products are highly complex, and our product development, manufacturing and integration testing may not be adequate to detect all defects, errors, failures and quality issues. Quality or performance problems for products covered under warranty could adversely impact our reputation and negatively affect our operating results and financial position. The development and production of new products with high complexity often involves problems with software, components and manufacturing methods. If significant warranty obligations arise due to reliability or quality issues arising from defects in software, faulty components or manufacturing methods, our operating results and financial position could be negatively impacted by:

- cost associated with fixing software or hardware defects;
- · high service and warranty expenses;
- · high inventory obsolescence expense;
- delays in collecting accounts receivable;
- · payment of liquidated damages for performance failures; and
- declining sales to existing customers.

Our use of open source software could impose limitations on our ability to commercialize our products.

We incorporate open source software into our products. Although we closely monitor our use of open source software, the terms of many open source software licenses have not been interpreted by the courts, and there is a risk that such licenses could be construed in a manner that could impose unanticipated conditions or restrictions on our ability to sell our products. In such event, we could be required to make our proprietary software generally available to third parties, including competitors, at no cost, to seek licenses from third parties in order to continue offering our products, to re-engineer our products or to discontinue the sale of our products in the event re-engineering cannot be accomplished on a timely basis or at all, any of which could adversely affect our revenues and operating expenses.

If we are unable to obtain necessary third-party technology licenses, our ability to develop new products or product enhancements may be impaired.

While our current licenses of third-party technology relate to commercially available off-the-shelf technology, we may in the future be required to license additional technology from third parties to develop new products or product enhancements. These third-party licenses may be unavailable to us on commercially reasonable terms, if at all. Our inability to obtain necessary third-party licenses may force us to obtain substitute technology of lower quality or performance standards or at greater cost, any of which could harm the competitiveness of our products and result in lost revenues.

We may pursue acquisitions, which involve a number of risks. If we are unable to address and resolve these risks successfully, such acquisitions could disrupt our business.

On August 20, 2012, we signed a definitive agreement to acquire Ericsson's fiber access assets; the acquisition transaction is expected to close in the fourth quarter of 2012. On February 22, 2011, we acquired Occam Networks. We may in the future acquire other businesses, products or technologies to expand our product offerings and capabilities, customer base and business. We have evaluated, and expect to continue to evaluate, a wide array of potential strategic transactions. We have limited experience making such acquisitions. Any of these transactions could be material to our financial condition and results of operations. The anticipated benefit of acquisitions may never materialize. In addition, the process of integrating acquired businesses, products or technologies may create unforeseen operating difficulties and expenditures. Some of the areas where we may face acquisition-related risks include:

- diversion of management time and potential business disruptions;
- · expenses, distractions and potential claims resulting from acquisitions, whether or not they are completed;
- retaining and integrating employees from any businesses we may acquire;
- issuance of dilutive equity securities or incurrence of debt;
- integrating various accounting, management, information, human resource and other systems to permit effective management;
- · incurring possible write-offs, impairment charges, contingent liabilities, amortization expense or write-offs of

goodwill;

- difficulties integrating and supporting acquired products or technologies;
- unexpected capital expenditure requirements;
- insufficient revenues to offset increased expenses associated with the acquisition;
- · opportunity costs associated with committing capital to such acquisitions; and
- · acquisition-related litigation.

Foreign acquisitions would involve risks in addition to those mentioned above, including those related to integration of operations across different cultures and languages, currency risks and the particular economic, political and regulatory risks associated with specific countries. We may not be able to address these risks successfully, or at all, without incurring significant costs, delays or other operating problems. Our inability to address successfully such risks could disrupt our business.

Our obligation to issue performance bonds to satisfy requirements under RUS and ARRA-related contracts may negatively impact our working capital and financial condition.

We are sometimes required to issue performance bonds to satisfy requirements under our RUS contracts, and expect that we may also be required to issue such bonds under the terms of contracts required by Broadband Stimulus programs under ARRA. The performance bonds generally cover the full amount of the RUS contract, and may be the same for ARRA contracts. Upon our performance under the contract and acceptance by the customer, the performance bond is released. The time period between issuing the performance bond and its release can be lengthy. We issue letters of credit under our existing credit facility to support these performance bonds. In the event we do not have sufficient capacity under our credit facility to support these bonds, we will have to provide certificates of deposit or other security, which could materially impact our working capital or limit our ability to satisfy such contract requirements. In the event that we are unable to issue such bonds, we may lose business and customers who purchase under RUS and ARRA contracts. In addition, if we exhaust our credit facility or working capital reserves in issuing such bonds, we may be required to eliminate or curtail expenditures to mitigate the impact on our working capital or financial condition.

Our use of and reliance upon development resources in China may expose us to unanticipated costs or liabilities.

We operate a wholly foreign owned enterprise, in Nanjing, China, where a dedicated team of engineers performs quality assurance, cost reduction and other engineering work. We also outsource a portion of our software development to a team of software engineers based in Shenyang, China. Our reliance upon development resources in China may not enable us to achieve meaningful product cost reductions or greater resource efficiency. Further, our development efforts and other operations in China involve significant risks, including:

- · difficulty hiring and retaining appropriate engineering resources due to intense competition for such resources and resulting wage inflation;
- the knowledge transfer related to our technology and exposure to misappropriation of intellectual property or confidential information, including information that is proprietary to us, our customers and third parties;
- heightened exposure to changes in the economic, security and political conditions of China;
- fluctuation in currency exchange rates and tax risks associated with international operations; and
- development efforts that do not meet our requirements because of language, cultural or other differences associated with international
 operations, resulting in errors or delays.

Difficulties resulting from the factors above and other risks related to our operations in China could expose us to increased expense, impair our development efforts, harm our competitive position and damage our reputation.

Our customers are subject to government regulation, and changes in current or future laws or regulations that negatively impact our customers could harm our business.

The FCC has jurisdiction over all of our U.S. customers. FCC regulatory policies that create disincentives for investment in access network infrastructure or impact the competitive environment in which our customers operate may harm our business. For example, future FCC regulation affecting providers of broadband Internet access services could impede the penetration of our customers into certain markets or affect the prices they may charge in such markets. Furthermore, many of our customers are subject to FCC rate regulation of interstate telecommunications services, and are recipients of federal universal service fund payments, which are intended to subsidize telecommunications services in areas that are expensive to serve. In early October 2011, the chairman of the FCC outlined a plan to transform the Universal Service Fund, an \$8 billion fund that is paid for by the nation's telephone customers and used to subsidize basic telephone service in rural areas, into one that will help expand broadband Internet service to 18 million Americans who lack high-speed access. Changes to these programs could change the

ability of IOCs to access capital and reduce our revenue opportunities. In addition, many of our customers are subject to state regulation of intrastate telecommunications services, including rates for such services, and may also receive funding from state universal service funds. Changes in rate regulations or universal service funding rules, either at the federal or state level, could adversely affect our customers' revenues and capital spending plans. In addition, various international regulatory bodies have jurisdiction over certain of our non-U.S. customers. Changes in these domestic and international standards, laws and regulations, or judgments in favor of plaintiffs in lawsuits against CSPs based on changed standards, laws and regulations could adversely affect the development of broadband networks and services. This, in turn, could directly or indirectly adversely impact the communications industry in which our customers operate. To the extent our customers are adversely affected by laws or regulations regarding their business, products or service offerings, our business, financial condition and results of operations would suffer.

We engage resellers, including Ericsson, to promote, sell, install and support our products to some customers in North America, and internationally. Their failure to do so or our inability to recruit or retain resellers may reduce our sales and thus harm our business.

We engage some value added resellers, or VARs, who provide sales and support services for our products. In particular, we expect the reseller agreement signed with Ericsson on August 22, 2012 to provide us with an extensive new global reseller channel. We compete with other telecommunications systems providers for our VARs' business and many of our VARs, including Ericsson, are free to market competing products. If Ericsson or any other VAR promotes a competitor's products to the detriment of our products or otherwise fails to market our products and services effectively, we could lose market share. In addition, the loss of a key VAR or the failure of VARs to provide adequate customer service could have a negative effect on customer satisfaction and could cause harm to our business. If we do not properly train our VARs to sell, install and service our products, our business, financial condition and results of operations may suffer. Our use of VARs and other third party support partners, and the associated risks, are likely to increase as we expand sales outside of North America.

We may be subject to governmental export and import controls that could subject us to liability or impair our ability to compete in additional international markets

Our products may be or become subject to U.S. export controls that will restrict our ability to export them outside of the free-trade zones covered by the North American Free Trade Agreement, Central American Free Trade Agreement and other treaties and laws. Therefore, future international shipments of our products may require export licenses or export license exceptions. In addition, the import laws of other countries may limit our ability to distribute our products, or our customers' ability to buy and use our products, in those countries. Changes in our products or changes in export and import regulations may create delays in the introduction of our products in international markets, prevent our customers with international operations from deploying our products or, in some cases, prevent the export or import of our products to certain countries altogether. Any change in export or import regulations or related legislation, shift in approach to the enforcement or scope of existing regulations, or change in the countries, persons or technologies targeted by such regulations, could negatively impact our ability to sell our products to existing or potential international customers.

If we lose any of our key personnel, or are unable to attract, train and retain qualified personnel, our ability to manage our business and continue our growth would be negatively impacted.

Our success depends, in large part, on the continued contributions of our key management, engineering, sales and marketing personnel, many of whom are highly skilled and would be difficult to replace. None of our senior management or key technical or sales personnel is bound by a written employment contract to remain with us for a specified period. In addition, we do not currently maintain key man life insurance covering our key personnel. If we lose the services of any key personnel, our business, financial condition and results of operations may suffer.

Competition for skilled personnel, particularly those specializing in engineering and sales, is intense. We cannot be certain that we will be successful in attracting and retaining qualified personnel, or that newly hired personnel will function effectively, both individually and as a group. In particular, we must continue to expand our direct sales force, including hiring additional sales managers, to grow our customer base and increase sales. In addition, if we offer employment to personnel employed by competitors, we may become subject to claims of unfair hiring practices, and incur substantial costs in defending ourselves against these claims, regardless of their merits. If we are unable to effectively recruit, hire and utilize new employees, execution of our business strategy and our ability to react to changing market conditions may be impeded, and our business, financial condition and results of operations may suffer.

Volatility or lack of performance in our stock price may also affect our ability to attract and retain our key personnel. Our executive officers and employees hold a substantial number of shares of our common stock and vested stock options. Employees may be more likely to leave us if the shares they own or the shares underlying their vested options have significantly appreciated in value relative to the original purchase prices of the shares or the exercise prices of the options, or if

the exercise prices of the options that they hold are significantly above the market price of our common stock. If we are unable to retain our employees, our business, operating results and financial condition will be harmed.

If we fail to maintain proper and effective internal controls, our ability to produce accurate financial statements on a timely basis could be impaired, which would adversely affect our operating results, our ability to operate our business and our stock price.

Ensuring that we have adequate internal financial and accounting controls and procedures in place to produce accurate financial statements on a timely basis is a costly and time-consuming effort that needs to be re-evaluated frequently. We have in the past discovered, and may in the future discover, areas of our internal financial and accounting controls and procedures that need improvement.

Our management is responsible for establishing and maintaining adequate internal control over financial reporting to provide reasonable assurance regarding the reliability of our financial reporting and the preparation of financial statements for external purposes in accordance with U.S. generally accepted accounting principles. Our management does not expect that our internal control over financial reporting will prevent or detect all error and all fraud. A control system, no matter how well designed and operated, can provide only reasonable, not absolute, assurance that the control system's objectives will be met. Because of the inherent limitations in all control systems, no evaluation of controls can provide absolute assurance that misstatements due to error or fraud will not occur or that all control issues and instances of fraud, if any, within our company will have been detected.

We have to comply with Section 404 of the Sarbanes-Oxley Act, which requires us to expend significant resources in developing the required documentation and testing procedures. We cannot be certain that the actions we have taken and are taking to improve our internal controls over financial reporting will be sufficient to maintain effective internal controls over financial reporting in subsequent reporting periods, or that we will be able to implement our planned processes and procedures in a timely manner. In addition, new and revised accounting standards and financial reporting requirements may occur in the future, and implementing changes required by new standards, requirements or laws may require a significant expenditure of our management's time, attention and resources and may adversely affect our reported financial results. If we are unable to produce accurate financial statements on a timely basis, investors could lose confidence in the reliability of our financial statements, which could cause the market price of our common stock to decline and make it more difficult for us to finance our operations and growth.

Interruptions, failures or material breaches in our information technology and communications systems could harm our business, customer relations and financial condition.

Information technology helps us operate efficiently, interface with customers, maintain financial accuracy and efficiency and accurately produce our financial statements. If we do not allocate and effectively manage the resources necessary to build and sustain the proper technology infrastructure, we could be subject to transaction errors, processing inefficiencies, the loss of customers, business disruptions or the loss of or damage to intellectual property through security breach. If our data management systems do not effectively collect, store, process and report relevant data for the operation of our business, whether due to equipment malfunction or constraints, software deficiencies or human error, our ability to effectively plan, forecast and execute our business plan and comply with applicable laws and regulations will be impaired, perhaps materially. Any such impairment could materially and adversely affect our financial condition, results of operations, cash flows and the timeliness with which we report our internal and external operating results.

We require user names and passwords in order to access our information technology systems. We also use encryption and authentication technologies to secure the transmission and storage of data. These security measures may be compromised as a result of third-party security breaches, employee error, malfeasance, faulty password management or other irregularity, and result in persons obtaining unauthorized access to our data or accounts. Third parties may attempt to fraudulently induce employees into disclosing user names, passwords or other sensitive information, which may in turn be used to access our information technology systems.

While we devote significant resources to network security, data encryption and other security measures to protect our information technology and communications systems and data, these security measures cannot provide absolute security. We may experience a breach of our systems and may be unable to protect sensitive data. The costs to us to eliminate or alleviate network security problems, bugs, viruses, worms, malicious software programs and security vulnerabilities could be significant, and our efforts to address these problems may not be successful and could result in unexpected interruptions, delays, cessation of service and may harm our business operations.

We incur significant increased costs as a result of operating as a public company, which may adversely affect our operating results and financial condition.

As a public company, we incur significant accounting, legal and other expenses that we did not incur as a private company, including costs associated with our public company reporting requirements. We also anticipate that we will continue to incur costs associated with corporate governance requirements, including requirements under the Sarbanes-Oxley Act and the Dodd-Frank Wall Street Reform and Consumer Protection Act, as well as rules implemented by the Securities Exchange Commission, or SEC, and the New York Stock Exchange, or NYSE. Furthermore, these laws and regulations could make it more difficult or more costly for us to obtain certain types of insurance, including director and officer liability insurance, and we may be forced to accept reduced policy limits and coverage or incur substantially higher costs to obtain the same or similar coverage. The impact of these requirements could also make it more difficult for us to attract and retain qualified persons to serve on our board of directors, our board committees or as executive officers.

New laws and regulations as well as changes to existing laws and regulations affecting public companies, including the provisions of the Sarbanes-Oxley Act and the Dodd-Frank Wall Street Reform and Consumer Protection Act and rules adopted by the SEC and the NYSE, would likely result in increased costs to us as we respond to their requirements. We are investing resources to comply with evolving laws and regulations, and this investment may result in increased general and administrative expense and a diversion of management's time and attention from revenue generating activities to compliance activities.

Risks Related to Our Merger Transaction with Occam and Our Pending Acquisition of the Fiber Access Business of Ericsson

Future results of the combined organization may differ materially from those in our current financial statements and financial forecasts, and the potential benefits of the transactions may not be realized.

The future results of the combined organization may be materially different from those contained in our current financial statements and financial forecasts. In addition, potential growth, expected financial results, perceived synergies and anticipated opportunities may not be realized through the ongoing integration of our business with those of Occam and Ericsson.

The Occam and Ericsson transactions could cause disruptions and materially adversely affect the future business and operations of the combined organization.

In connection with the transactions, it is possible that some customers, suppliers and other persons with whom we, Occam or Ericsson have had a business relationship may delay or defer certain business decisions, or determine to purchase a competitor's products. In particular, customers could be reluctant to purchase products due to uncertainty about the direction of our combined technology and product road map, and uncertainty regarding the willingness of the combined organization to support and service existing products after the transactions. If customers, suppliers or other persons, delay or defer business decisions, or purchase a competitor's products, it could negatively impact revenues, earnings and cash flows of the combined organization, as well as the market price of our common stock.

A class action lawsuit is pending against Occam and its former directors challenging the acquisition of Occam by us, and an unfavorable judgment or ruling in this lawsuit could result in substantial costs.

On September 16, 2010, the Company, Occam, Ocean Sub I, Inc., and two direct, wholly owned subsidiaries of the Company, entered into an Agreement and Plan of Merger and Reorganization, or the Merger Agreement. In response to the announcement of the Merger Agreement, four separate purported class action complaints were filed by purported stockholders of Occam against Occam and the former members of its board of directors, including three complaints in the California Superior Court for Santa Barbara County and one complaint in the Delaware Court of Chancery. Each complaint generally alleged that the former members of the Occam board breached their fiduciary duties in connection with the acquisition of Occam by Calix, by, among other things, engaging in an allegedly unfair process and agreeing to an allegedly unfair price for the merger transaction, and each complaint sought rescission of the merger transaction as well as other remedies including unspecified damages and costs and fees. The three California class actions have been dismissed without prejudice in favor of the Delaware class action. On January 6, 2012, the Delaware court found that certain lead plaintiffs, Michael Steinhardt, Steinhardt Overseas Management, L.P. and Ilex Partners, L.L.C., collectively the "Steinhardt Plaintiffs", had engaged in improper trading of Calix shares and dismissed the Steinhardt Plaintiffs from the case with prejudice. The remaining lead plaintiffs are expected to continue to seek an award of damages in an unspecified amount.

We believe that the allegations in the pending action are without merit and intend to vigorously contest the action. However, there can be no assurance that we will be successful in our defense. In addition, under the Merger Agreement and Delaware law, we have obligations, under certain circumstances, to hold harmless and indemnify each of the former directors of Occam against judgments, fines, settlements and expenses related to claims against such directors and otherwise to the

fullest extent permitted under Delaware law and Occam's bylaws and certificate of incorporation. Such obligations may apply to this lawsuit, and therefore an unfavorable outcome in this lawsuit could result in substantial costs for us.

Risks Related to Ownership of Our Common Stock

Our stock price may be volatile, and the value of an investment in our common stock may decline.

The trading price of our common stock has been, and is likely to continue to be, volatile, which means that it could decline substantially within a short period of time and could be subject to wide fluctuations in response to various factors, some of which are beyond our control. These factors include those discussed in this "Risk Factors" section of this Form 10-Q and others such as:

- quarterly variations in our results of operations or those of our competitors;
- · failures by us to meet any guidance regarding our anticipated results that we have previously provided;
- changes in earnings estimates or recommendations by securities analysts;
- announcements by us or our competitors of new products, significant contracts, commercial relationships, acquisitions or capital commitments:
- developments with respect to intellectual property rights;
- our ability to develop and market new and enhanced products on a timely basis;
- our commencement of, or involvement in, litigation;
- changes in governmental regulations or in the status of our regulatory approvals; and
- a slowdown in the communications industry or the general economy.

In recent years, the stock market in general, and the market for technology companies in particular, has experienced extreme price and volume fluctuations that have often been unrelated or disproportionate to the operating performance of those companies. Broad market and industry factors may seriously affect the market price of our common stock, regardless of our actual operating performance. In addition, in the past, following periods of volatility in the overall market and the market price of a particular company's securities, securities class action litigation has often been instituted against these companies. This litigation, if instituted against us, could result in substantial costs and a diversion of our management's attention and resources.

If securities or industry analysts do not publish research or reports about our business or if they issue an adverse or misleading opinion regarding our stock, our stock price and trading volume could decline.

The trading market for our common stock will be influenced by the research and reports that industry or securities analysts publish about us or our business. If any of the analysts who cover us issue an adverse or misleading opinion regarding our stock, our stock price would likely decline. If one or more of these analysts cease coverage of our company or fail to publish reports on us regularly, we could lose visibility in the financial markets, which in turn could cause our stock price or trading volume to decline.

Provisions in our charter documents and under Delaware law could discourage a takeover that stockholders may consider favorable and may lead to entrenchment of management.

Our amended and restated certificate of incorporation and amended and restated bylaws contain provisions that could have the effect of delaying or preventing changes in control or changes in our management without the consent of our board of directors. These provisions include:

- a classified board of directors with three-year staggered terms, which may delay the ability of stockholders to change the membership of a majority of our board of directors;
- · no cumulative voting in the election of directors, which limits the ability of minority stockholders to elect director candidates;
- the exclusive right of our board of directors to elect a director to fill a vacancy created by the expansion of the board of directors or the
 resignation, death or removal of a director, which prevents stockholders from being able to fill vacancies on our board of directors;
- the ability of our board of directors to issue shares of preferred stock and to determine the price and other terms of those shares, including preferences and voting rights, without stockholder approval, which could be used to significantly dilute the ownership of a hostile acquirer;
- a prohibition on stockholder action by written consent, which forces stockholder action to be taken at an annual or special meeting of our stockholders;
- the requirement that a special meeting of stockholders may be called only by the chairman of the board of

- directors, the chief executive officer or the board of directors, which may delay the ability of our stockholders to force consideration of a proposal or to take action, including the removal of directors; and
- advance notice procedures that stockholders must comply with in order to nominate candidates to our board of directors or to propose matters
 to be acted upon at a stockholders' meeting, which may discourage or deter a potential acquirer from conducting a solicitation of proxies to
 elect the acquirer's own slate of directors or otherwise attempting to obtain control of us.

We are also subject to certain anti-takeover provisions under Delaware law. Under Delaware law, a corporation may not, in general, engage in a business combination with any holder of 15% or more of its capital stock unless the holder has held the stock for three years or, among other things, the board of directors has approved the transaction.

We may need additional capital in the future to finance our business.

We may need to raise additional capital to fund operations in the future. Although we believe that, based on our current level of operations and anticipated growth, our existing cash, cash equivalents will provide adequate funds for ongoing operations, planned capital expenditures and working capital requirements for at least the next 12 months, we may need additional capital if our current plans and assumptions change. If future financings involve the issuance of equity securities, our then-existing stockholders would suffer dilution. If we raised additional debt financing, we may be subject to restrictive covenants that limit our ability to conduct our business. We may not be able to raise sufficient additional funds on terms that are favorable to us, if at all. If we fail to raise sufficient funds and continue to incur losses, our ability to fund our operations, take advantage of strategic opportunities, develop products or technologies or otherwise respond to competitive pressures could be significantly limited. Any failure to obtain financing when and as required could force us to curtail our operations, which would harm our business.

We do not currently intend to pay dividends on our common stock and, consequently, our stockholder's ability to achieve a return on their investment will depend on appreciation in the price of our common stock.

We do not currently intend to pay any cash dividends on our common stock for the foreseeable future. We currently intend to invest our future earnings, if any, to fund our growth. Additionally, the terms of our credit facility restrict our ability to pay dividends. Therefore, our stockholders are not likely to receive any dividends on our common stock for the foreseeable future.

ITEM 2. Unregistered Sales of Equity Securities and Use of Proceeds.

Defaults Upon Senior Securities.

There were no unregistered sales, or purchases made by or on behalf of us or by any affiliated purchaser, of our equity securities during the nine months ended September 29, 2012 .

None.		
ITEM 4.	Mine Safety Disclosures.	
Not applicable.		

ITEM 5. Other Information.

None.

ITEM 3.

ITEM 6. Exhibits.

Exhibit Number	Description
10.1	Asset Purchase Agreement between Ericsson Inc. and Calix, Inc., dated August 20, 2012.
10.2*	Calix, Inc. Non-Employee Director Cash Compensation Policy, effective January 1, 2012.
31.1	Certification of Chief Executive Officer of Calix, Inc. Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
31.2	Certification of Chief Financial Officer of Calix, Inc. Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
32.1	Certification of Chief Executive Officer and Chief Financial Officer of Calix, Inc. Pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.
101.INS **	XBRL Instance Document
101.SCH **	XBRL Taxonomy Extension Schema Document
101.CAL **	Taxonomy Extension Calculation Linkbase Document
101.DEF **	Taxonomy Extension Definition Linkbase Document
101.LAB **	XBRL Taxonomy Extension Label Linkbase Document
101.PRE **	XBRL Taxonomy Extension Presentation Linkbase Document

^{*} Indicates management compensatory plan, contract or arrangement.

^{**} In accordance with Rule 406T of Regulation S-T, the XBRL information is furnished and not filed herewith, is not a part of a registration statement or Prospectus for purposes of sections 11 or 12 of the Securities Act of 1933, is deemed not filed for purposes of section 18 of the Securities Exchange Act of 1934, and otherwise is not subject to liability under these sections.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

		CALIX, INC. (Registrant)	
Dated:	November 1, 2012	By:	/s/ Carl Russo
			Carl Russo
			Chief Executive Officer
			(Principal Executive Officer)
Dated:	November 1, 2012	By:	/s/ Michael Ashby
			Michael Ashby
			Chief Financial Officer
			(Principal Financial Officer)

[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.		
=	A GOVER DVID CHA COLOR DVID MENTE	
	ASSET PURCHASE AGREEMENT between:	
	[ERICSSON],	
	and CALIX, INC.	
	Dated August 20, 2012	
=		
	1	

TABLE OF CONTENTS

ARTICLE I : SALE OF ASSETS; RELATED TRANSACTIONS		11
1.1	Sale of Assets	11
1.2	Purchase Price	11
1.3	Closing	11
1.4	Closing Deliveries	11
1.5	Allocation	12
1.6	Withholding of Taxes	13
1.7	Consents	13
1.8	Ericsson Trademarks Excluded	13
ARTICLE II : R	EPRESENTATIONS AND WARRANTIES OF ERICSSON	13
2.1	Organization, Standing and Power	13
2.2	Ownership of Purchased Assets	14
2.3	Authority and Enforceability	14
2.4	Non-Contravention	14
2.5	Governmental Consents; Approvals; Permits	14
2.6	Material Contracts	14
2.7	Title to Purchased Assets; Encumbrances	16
2.8	Litigation	16
2.9	Restrictions on Business Activities	16
2.10	Compliance with Laws	17
2.11	Intellectual Property	17
2.12	Patents and Patent Contracts	19
2.13	Tax Matters	20
2.14	Employee Matters	20
2.15	Books and Records	21
2.16	Customers and Suppliers	21
2.17	Products; Warranties; Purchase Commitments	21
2.18	Finders' Fees; Transaction Expenses	22
ARTICLE III : I	REPRESENTATIONS AND WARRANTIES OF CALIX	22
3.1	Organization and Standing	22
3.2	Authority and Enforceability	22
3.3	Non-Contravention	22
3.4	Government Consent	22
3.5	Finders' Fee; Transaction Expenses	23
ARTICLE IV : 0	CONDUCT PRIOR TO THE CLOSING	23
4.1	Conduct of Business of Ericsson	23
4.2	Restrictions on Conduct of Business of Ericsson	23

ARTICLE V : 0	CERTAIN COVENANTS OF THE PARTIES	24
5.1	Confidentiality; Public Disclosure	25
5.2	Regulatory Approvals	25
5.3	Reasonable Efforts	25
	2	

5	5.4	Third Party Consents; Notices	25
5	5.5	Access to Information	25
5	5.6	Employees and Contractors	26
5	5.7	Tax Matters	27
5	5.8	Assets Held by Affiliates	28
5	5.9	Further Actions	28
5	5.10	Further Assurances	28
5	5.11	Non-Compete; Non-Solicit	28
5	5.12	Transition Services	29
ARTICI	LE VI : CON	NDITIONS TO THE TRANSACTIONS	29
6	6.1	Conditions to Obligations of Each Party to Consummate the Transactions	29
6	6.2	Additional Conditions to Obligations of Ericsson	29
6	6.3	Additional Conditions to the Obligations of Calix	30
6	6.4	Employees	30
6	6.5	Proceedings	30
Ć	6.6	Release of Liens	30
ARTICI	LE VII : TE	RMINATION	30
7	7.1	Termination	30
7	7.2	Effect of Termination	31
7	7.3	Extension; Waiver	31
7	7.4	Non-exclusivity of Termination Rights	31
ARTICI	LE VIII : IN	DEMNIFICATION	32
8	8.1	Survival of Representations and Warranties	32
8	8.2	Indemnification	32
8	8.3	Indemnification Procedure	33
ARTICI	LE IX : GEN	NERAL PROVISIONS	33
ç	9.1	Fees and Expenses	33
ç	9.2	Bulk Sales Compliance	33
Ģ	9.3	Attorneys' Fees	34
Ģ	9.4	Governing Law; Venue	34
ç	9.5	Notices	34
Ģ	9.6	Successors and Assigns; Parties In Interest	34
Ģ	9.7	Remedies Cumulative; Specific Performance	35
Ģ	9.8	Severability	35
Ģ	9.9	Entire Agreement	35
Ģ	9.10	English Language	35
9	9.11	Execution	35

Exhibit B -	Reseller Agreement
Exhibit C -	License Agreements
Exhibit D -	Transition Services
Exhibit E -	Disclosure Schedule
Exhibit F -	List of Ericsson Products

ASSET PURCHASE AGREEMENT

THIS ASSET PURCHASE AGREEMENT ("Agreement") is entered into on August 20, 2012 ("Agreement Date"), between Ericsson Inc., a Delaware corporation with offices at 6300 Legacy Drive, Plano, TX 75024 ("Ericsson"), and Calix, Inc., a Delaware corporation with offices at 1035 N. McDowell Blvd., Petaluma, CA 94954 ("Calix"). Ericsson and Calix may be referred to individually as a "Party" and collectively as the "Parties."

Calix and Ericsson wish to provide for the sale by Ericsson to Calix of certain of Ericsson's assets relating to Ericsson Products (as defined below) and certain other related transactions between the Parties, on the terms set forth below.

DEFINITIONS

Capitalized terms used in this Agreement are defined above and as follows:

- "Affiliate" has the meaning set forth in Rule 144 under the Securities Act.
- "Allocation" has the meaning set forth in Section 1.5(a).
- "Assets" means, with respect to any Person, any and all of such Person's right, title and ownership interest in and to all properties, rights, claims, goodwill, interests and assets of every kind and description, real, personal or mixed, tangible and intangible, whether accrued, contingent or otherwise, and wherever located (including in the possession of vendors or other third parties or elsewhere), in each case whether or not recorded or reflected on the books and records or financial statements of any Person.
- "Assumed Liabilities" means all Liabilities arising out of (i) Calix's operation and ownership of the Purchased Assets and (ii) manufacture, use, sale, offer to sell, importation, export and/or distribution of the Ericsson Products, in each case solely to the extent such Liabilities arise or accrue after the Closing Date.
- "Bill of Sale" has the meaning set forth in Section 1.4(a)(ii).
- "Books and Records" means original or true copies of: (A) all product, business and marketing plans and studies, sales and advertising and promotional literature, creative materials and artwork, (B) all books, records (including customer, supplier and purchasing records), invoices, lists (including price, customer and supplier lists), documents, ledgers, financial data, files, reports, product and design manuals, plans, drawings, technical manuals, management information systems (including related computer software) and operating records of every kind, (C) all telephone numbers, fax numbers and email addresses (D) all personnel records and (E) all books and records or documents relating to Taxes imposed on the Purchased Assets, in each case whether maintained as hard copy or stored in computer memory and whether owned by Ericsson or its Affiliates. Books and Records exclude any of the foregoing related to Ericsson Patents.
- "Business" means the business operations and activities of Ericsson that relate primarily to the development, licensing, marketing and sale of Ericsson Products, and related Technology and services. Business shall exclude any activities regarding the filing, prosecution, maintenance, licensing or enforcement of the Ericsson Patents.
- "Business Day" means any day other than a Saturday, a Sunday or other day on which banking institutions in San Francisco, California are not required to be open.

- "Business Employee" means the employees of Ericsson or its Affiliates, as set forth in Section 5.6 of the Disclosure Schedule, including any such employees on temporary leave, vacation or a regularly scheduled day off, but, unless otherwise required under local employment Legal Requirements, excluding any such employees on long term disability or whose employment with Ericsson or its Affiliates has terminated prior to the Closing.
- "Calix Closing Certificate" has the meaning set forth in Section 1.4(a)(vi).
- "Closing" has the meaning set forth in Section 1.3.
- "Closing Date" has the meaning set forth in Section 1.3.
- "Code" means the Internal Revenue Code of 1986, as amended.
- "Confidentiality Agreement" has the meaning set forth in Section 5.1.
- "Contract" means any written or oral legally binding contract, agreement, instrument, commitment, obligation, understanding, or undertaking of any nature. Contract shall exclude Patent Contracts and Third Party Patent Contracts.
- "Disclosure Schedule" means the schedule (dated as of the Agreement Date) delivered to Calix on behalf of Ericsson, a copy of which is attached to this Agreement and incorporated by reference.
- "Encumbrance" means any lien, pledge, charge, mortgage, security interest, encumbrance, equity, trust, equitable interest, claim, preference, right of possession, lease, tenancy, license, encroachment, covenant, infringement, interference, Order, proxy, option, right of first refusal, preemptive right, community property interest, legend, defect, impediment, exception, reservation, limitation, impairment, imperfection of title, condition or restriction of any nature (including any restriction on the transfer of any asset, any restriction on the receipt of any income derived from any asset, any restriction on the use of any asset and any restriction on the possession, exercise or transfer of any other attribute of ownership of any asset). Encumbrance shall exclude Patent Contracts.
- "Ericsson Authorizations" has the meaning set forth in Section 2.5(b).
- "Ericsson Closing Certificate" has the meaning set forth in Section 1.4(b)(v).
- "Ericsson Employee Plans" means (A) each of the "employee benefit plans" (as such term is defined in Section 3(3) of ERISA), of which Ericsson or any trade or business (whether or not incorporated) which is treated as a single employer with Ericsson within the meaning of Section 414(b), (c), (m) or (o) of the Code (an "ERISA Affiliate") is or ever was a sponsor or participating employer or as to which Ericsson or any of its ERISA Affiliates makes contributions or is required to make contributions, and (B) any employment, severance or other agreement, arrangement or policy of Ericsson or any of its ERISA Affiliates (whether written or oral) providing for compensation, health, life, vision or dental insurance coverage (including self-insured arrangements), workers' compensation, disability benefits, supplemental unemployment benefits, vacation benefits or retirement benefits, fringe benefits, or for profit sharing, deferred compensation, bonuses, stock options, stock appreciation or other forms of incentive compensation or post-retirement insurance, compensation or benefits, in each case as applicable to any Business Employee (or his or her dependents, spouse or beneficiary).
- "Ericsson Licensed Intellectual Property" means all Technology and Intellectual Property Rights owned or purported to be owned by Ericsson or its Affiliates that are used by the Company in the Business or for

the design, development, manufacturing, commercialization, distribution and support of the Ericsson Products but are not incorporated in the Ericsson Products and are not primarily related to the design, development, manufacturing, commercialization, distribution and support of the Ericsson Products. Ericsson Licensed Intellectual Property shall exclude Ericsson Patents.

- "Ericsson Patents" means any Patent owned or purported to be owned or controlled by Ericsson or its Affiliates.
- "Ericsson Material Adverse Effect" means any change, event, development, circumstance or effect (each, an "Effect") that, individually or taken together with all other Effects, and regardless of whether or not such Effect constitutes a breach of the representations or warranties made by Ericsson in this Agreement, is, or could reasonably be expected to become, (A) materially adverse to the Business or the Purchased Assets, or (B) materially impede or delay Ericsson's ability to consummate the Transactions under the terms of this Agreement and Legal Requirements, except to the extent any such Effect directly results from or directly arises out of: (i) changes in general economic conditions (provided that such changes do not affect the Business or the Purchased Assets in a disproportionate manner compared to other comparable businesses in the industry in which Ericsson operates the Business), (ii) changes affecting the industry generally in which the Business operates (provided that such changes do not affect the Business or the Purchased Assets in a disproportionate manner compared to other comparable businesses in such industry), or (iii) the announcement of the transactions contemplated by this Agreement.
- "Ericsson-Owned Intellectual Property" means all Technology and Intellectual Property Rights owned or purported to be owned by Ericsson or its Affiliates that is incorporated in the Ericsson Products or is otherwise primarily related to the design, development, manufacturing, commercialization, distribution and support of the Ericsson Products. Ericsson-Owned Intellectual Property shall exclude Ericsson Patents.
- "Ericsson Products" means those products from Ericsson's GPON 1500 portfolio (OLT, ONT), and Entriview management system listed on Exhibit F.
- "Ericsson Product Patents" "means the Ericsson Patents scheduled in the attached Patent License Agreement to be entered at Closing.
- "Ericsson Proprietary Software" means Software owned or purported to be owned by Ericsson or its Affiliates that is incorporated in the Ericsson Products or is otherwise primarily related to the design, development, manufacturing, commercialization, distribution and support of the Ericsson Products.
- "ERISA" means the Employee Retirement Income Security Act of 1974, as amended.
- "ERISA Affiliate" has the meaning set forth in the definition of Ericsson Employee Plans.
- "Excluded Assets" means all Assets of Ericsson that are not specifically included within the Purchased Assets.
- "Excluded Liabilities" means any Liability of Ericsson or its Affiliates other than the Assumed Liabilities, including without limitation (i) any Liability for Transaction Expenses, (ii) any Liability arising out of Ericsson's operation and ownership of the Purchased Assets on or before the Closing Date, (iii) any Liability for Taxes of Ericsson or its Affiliates except as specifically contemplated by Section 5.7(b); and (iv) any Liability arising out of Ericsson's manufacture, use, sale, offer to sell, importation, export and/or distribution of the Ericsson Products, in any case of the foregoing arising or accruing on or

before the Closing Date;

- "GAAP" means United States generally accepted accounting principles consistently applied.
- "Governmental Authorization" means any: (A) permit, license, certificate, franchise, concession, approval, consent, ratification, permission, clearance, confirmation, endorsement, waiver, certification, designation, rating, registration, qualification or authorization issued, granted, given or otherwise made available by or under the authority of any Governmental Entity or under any Legal Requirement; or (B) right under any Contract with any Governmental Entity. Governmental Authorization shall exclude any of the foregoing which relate to Patents.
- "Governmental Entity" means any: (A) nation, principality, state, commonwealth, province, territory, county, municipality, district or other jurisdiction of any nature; (B) federal, state, local, municipal, foreign or other government; (C) governmental or quasi-governmental authority of any nature (including any governmental division, subdivision, department, agency, bureau, branch, office, commission, council, board, instrumentality, officer, official, representative, organization, unit, body or entity and any court or other tribunal); (D) multinational organization or body; or (E) individual, entity or body exercising, or entitled to exercise, any executive, legislative, judicial, administrative, regulatory, police, military or taxing authority or power of any nature. Governmental Entity shall exclude any of the foregoing having jurisdiction over any matters relating to Patents.
- "Indemnified Persons" means the following Persons: (A) Calix; (B) Calix's current and future Affiliates; (C) the Representatives of the Persons referred to in clauses "(A)" and "(B)" above; and (D) the successors and assigns of the Persons referred to in clauses "(A)", "(B)" and "(C)" above.
- "Infringing" has the meaning set forth in Section 2.11(f).
- "Intellectual Property Rights" means all past and present rights of the following types, whether registered or unregistered, which may exist under the laws of any jurisdiction: (A) rights associated with works of authorship, including exclusive exploitation rights, copyrights, and moral rights; (B) trade secrets and other rights in know-how and confidential or proprietary information (including any business plans, designs, technical data, , customer data, financial information, pricing and cost information, bills of material, or other similar information); (C) mask work rights; (D) rights in databases and data collections (including knowledge databases, customer lists and customer databases); (E) URL and domain name registrations; (F) any other non-patented proprietary rights applicable to Technology now known or later recognized in any jurisdiction; and (G) all past and present claims and causes of action arising out of or related to infringement or misappropriation of any of the above. Intellectual Property Rights shall exclude Patents.
- "IRS" means the United States Internal Revenue Service.
- "Knowledge" Information shall be deemed to be known to or to the "Knowledge" or "knowledge" of Ericsson if that information is actually known by a senior management (Director level or above) employee of Ericsson connected with the Business, or should be known by any of those Persons after reasonable inquiry.
- "Legal Requirement" means any federal, state, local, municipal, foreign or other law, statute, legislation, constitution, principle of common law, resolution, ordinance, code, edict, decree, proclamation, treaty, convention, rule, regulation, ruling, directive, pronouncement, requirement, specification, determination, decision, opinion or interpretation issued, enacted, adopted, passed, approved, promulgated, made, implemented or otherwise put into effect by or under the authority of any Governmental Entity, specifically excluding any foregoing legal requirement relating to Patents.

- "Liability" means all debts, liabilities and obligations (including with respect to Taxes), whether accrued or fixed, absolute or contingent, matured or unmatured, determined or determinable, asserted or unasserted, known or unknown, including those arising under any Legal Requirements or Proceeding and those arising under any Contract.
- "License Agreements" has the meaning set forth in Section 1.4(a)(iv).
- "Losses" means any loss, claim, Liability, damage, fee, obligation, judgment, settlement, interest, penalty, fee, Proceeding, charge, cost and expense, including costs of investigation and defense and fees and expenses of lawyers, accountants, experts and other professionals.
- "Material Contracts" has the meaning set forth in Section 2.6(a).
- "Offer Letters" means employment offer letters to be delivered by Calix to Business Employees after the Agreement Date.
- "Order" means any: (A) order, judgment, injunction, edict, decree, ruling, pronouncement, determination, decision, opinion, verdict, sentence, subpoena, writ or award issued, made, entered, rendered or otherwise put into effect by or under the authority of any court, administrative agency or other Governmental Entity or any arbitrator or arbitration panel; or (B) Contract with any Governmental Entity entered into in connection with any Proceeding, excluding any of the foregoing related to Patents.
- "Patents" means United States and foreign patents and patent applications, and any renewals, reissues, reexaminations, extensions, continuations, continuations-in-part, divisions and substitutions relating to any of the patents and patent applications.
- "Patent Contract" means any written or oral legally binding contract, agreement, instrument, commitment, obligation, understanding, or undertaking of any nature pursuant to which Ericsson in-licenses any Third Party Patent or Ericsson licenses any Ericsson Patents to any Third Party, where the essential subject of the contract is the licensing of Patents.
- "Patent License Agreement" shall mean the License Agreement pursuant to which Ericsson licenses the Ericsson Product Patents to Calix.
- "Person" means any individual, entity or Governmental Entity.
- "Post-Closing Tax Period" means any Tax Period beginning after the Closing Date and that portion of a Straddle Period beginning after the Closing Date.
- "Pre-Closing Period" means the period from the Agreement Date through the earlier of the termination of this Agreement under Article VII and the Closing Date.
- "Pre-Closing Tax Period" means any Tax Period ending on or before the Closing Date and the portion of any Straddle Period ending on the Closing Date.
- "Proceeding" means any action, suit, litigation, mediation, arbitration, proceeding (including any civil, criminal, administrative, investigative or appellate proceeding and any informal proceeding), prosecution, contest, hearing, inquiry, inquest, audit, examination or investigation commenced, brought, conducted or heard by or before, or otherwise involving, any Governmental Entity or any arbitrator or arbitration panel. Proceeding shall exclude any of the foregoing which relate to Patents.

- "Property Taxes" means all real property, personal property and similar ad valorem Taxes.
- "Purchase Price" has the meaning set forth in Section 1.2.
- "Purchased Assets" means all of the Assets of Ericsson and its Affiliates listed on Exhibit A. For the avoidance of doubt, no current or future Ericsson Patents or rights under any Patent Contracts shall be considered a Purchased Asset.
- "Related Party" Each of the following shall be deemed to be a "Related Party": (A) each individual who is, or who has at any time been, an officer, agent, employee or stockholder of Ericsson; (B) each member of the family of each of the individuals referred to in clause "(A)" above; and (C) any entity (other than Ericsson) in which any one of the individuals referred to in clauses "(A)" and "(B)" above holds or held (or in which more than one of such individuals collectively hold or held), beneficially or otherwise, a controlling interest or a material voting, proprietary or equity interest.
- "Representatives" means officers, directors, employees, agents, attorneys, accountants, advisors and representatives of a Person.
- "Reseller Agreement" has the meaning set forth in Section 1.4(a)(iii).
- "Securities Act" means the Securities Act of 1933, as amended.
- "SEC" means the Securities and Exchange Commission.
- "Software" means, with respect to the Ericsson Products, any computer program, operating system, database, applications system, firmware or software code of any nature, whether operational, under development or inactive, including all object code, source code, data files, rules, definitions or methodology derived from the above and any derivations, updates, enhancements and customization of any of the above, processes, know-how, operating procedures, methods and all other Technology embodied with the above, tools, developers' kits, utilities, developers' notes, technical manuals, user manuals and other documentation, including comments and annotations, whether in machine-readable form, programming language or any other language or symbols and whether stored, encoded, recorded or written on disk, tape, film, memory device, paper or other media of any nature.
- "Specifications" means, with respect to any Ericsson Product, the GDS II, SPICE and Verilog and other Software files used to represent the design, specifications, functionality or performance of such product, including all technical records related to such product.
- "Straddle Period" means any Tax Period beginning before or on and ending after the Closing Date.
- "Tax" or "Taxes" means any federal, state, local or foreign income, gross receipts, license, payroll, employment, excise, severance, stamp, occupation, premium, windfall profits, environmental, customs duties, capital stock, franchise, profits, escheat, withholding, social security, unemployment, disability, real property, personal property, sales, use, transfer, registration, value added, alternative or add-on minimum, estimated, or other tax of any kind whatsoever, including any interest, penalty or addition, whether disputed or not, and including any obligation to indemnify or otherwise assume or succeed to the Tax liability of any other Person by law, by Contract or otherwise.
- "Tax Period" means any period prescribed by any Governmental Entity for which a Tax Return is required to be filed or a Tax is required to be paid.

- "Tax Returns" means any return, declaration, report, claim for refund, transfer pricing report or information return or statement relating to Taxes filed or required to be filed with a Governmental Entity with jurisdiction over Taxes, including any schedule or attachment, and including any amendment.
- "Technology" means, with respect to the Ericsson Products, algorithms, APIs, databases, data collections, diagrams, mask works, methods and processes, know-how, trade secrets, , network configurations and architectures, proprietary information, protocols, layout rules, schematics, packaging and other specifications, Software (in any form, including without limitation source code, executable code, RTL code, Gerber files and GDSII files), techniques, interfaces, verification tools, , works of authorship, technical documentation, designs, bills of material, build instructions, test reports, routines, formulae, test vectors, IP cores, net lists, photomasks, lab notebooks, processes, prototypes, samples, studies, , in each case whether or not embodied in any tangible form.
- "Third Party" means any party other than the parties hereto and their Affiliates.
- "Third Party Intellectual Property" means all Technology and Intellectual Property Rights owned by third parties, including Third Party Software, that is either (A) licensed, offered or provided to customers of Ericsson or its Affiliates as part of or in conjunction with any Ericsson Product, or (B) otherwise used by Ericsson or its Affiliates in connection with the Business as currently conducted. Third Party Intellectual Property shall exclude any Third Party Patent.
- "Third Party Patent" means any Patent owned or controlled by a Third Party.
- "Third Party Patent-Ericsson Product" means any Third Party Patent that the Third Party has alleged in writing to Ericsson at any time in the three-year period prior to the date hereof to be infringed specifically by the Ericsson Products. An standards related intellectual property rights or patent declaration or claim chart that generally maps a Third Party Patent to a standard or industry specification shall not be included within the definition of Third Party Patent-Ericsson Product.
- "Third Party Patent Proceeding" means any unresolved action, suit, litigation, mediation or arbitration, pursuant to which an Ericsson Product has been specifically identified as Infringing a Third Party Patent.
- "Third Party Patent Contract" means any written or oral legally binding contract, agreement, instrument, commitment, obligation, understanding, or undertaking of any nature pursuant to which Ericsson in-licenses any Third Party Patent and which specifically references the Ericsson Products.
- "Third Party Software" means all Software owned by third parties that is either (A) licensed, offered or provided to customers of Ericsson or its Affiliates as part of or in conjunction with any Ericsson Product, or (B) otherwise used by Ericsson or its Affiliates in connection with the Business, excluding generally commercially available, "off-the-shelf" or "shrink-wrapped" Software that is not redistributed with or used in the development or provision of the Ericsson Products and that has an individual acquisition and annual support cost of \$5,000 or less.
- "Transaction Documents" means: (A) this Agreement; (B) the Bill of Sale; (C) the Reseller Agreement; (D) the License Agreements; (E) the Ericsson Closing Certificate; (F) the Calix Closing Certificate; and (G) the Transition Services Exhibit.
- "Transaction Expenses" means: (A) any and all third party fees and expenses of Ericsson in connection with the Transactions and this Agreement whether or not paid, payable, billed or accrued (including any fees and expenses of legal counsel and accountants, fees and expenses payable to financial advisors, investment bankers and brokers of Ericsson not including any contingencies, such as escrows, and all

other miscellaneous expenses or costs incurred by Ericsson), (B) any bonuses, retention bonuses, commissions, accrued wages, vacation accrual or severance payments to be paid by or on behalf of Ericsson to any officer or employee of Ericsson in connection with the Transactions, including any such payments payable upon the termination of any Business Employee's employment with Ericsson or any of its Affiliates, and any payroll taxes incurred by Ericsson and (C) all other miscellaneous expenses or costs, in each case, incurred by Ericsson or any of its Related Parties that are payable by Ericsson, in connection with the Transactions.

"Transactions" means: (A) the execution and delivery of the respective Transaction Documents, and (B) all of the contemplated transactions by the respective Transaction Documents, including: (i) the sale of the Purchased Assets by Ericsson to Calix under the Agreement; (ii) the assumption of the Assumed Liabilities by Calix; and (iii) the performance by Ericsson and Calix of their respective obligations under the Transaction Documents, and the exercise by Ericsson and Calix of their respective rights under the Transaction Documents.

"Transition Services" has the meaning set forth in Section 5.12.

"Transition Services Exhibit" has the meaning set forth in Section 1.4(a)(v).

"Transfer Tax" has the meaning set forth in Section 5.7(c).

"Transferred Employees" has the meaning set forth in Section 5.6(a).

ARTICLE I: SALE OF ASSETS; RELATED TRANSACTIONS

- 1.1 Sale of Assets . At the Closing, Ericsson shall sell and deliver to Calix valid title to the Purchased Assets, free and clear of any Encumbrances. Ericsson and its Affiliates shall retain and be responsible for, and Calix shall not be obligated to and shall not assume, perform or discharge, any of the Excluded Liabilities.
- 1.2 <u>Purchase Price</u>. As consideration for the sale of the Purchased Assets to Calix, Calix shall pay to Ericsson the sum of [***] ("Purchase Price").
- 1.3 Closing. Unless this Agreement is earlier terminated under Section 7.1, the closing of the

Transactions ("Closing") shall take place as soon as reasonably practicable and no later than two Business Days after the satisfaction or waiver of each of the conditions set forth in Article VI (other than those conditions that by their nature are to be satisfied at the Closing, but subject to the fulfillment or waiver of those conditions) or at such other time as may be agreed upon by the Parties. The Closing shall take place at the offices of Calix, or at such other location as the Parties agree. The date on which the Closing occurs is referred to as the "Closing Date."

1.4 Closing Deliveries.

- (a) Calix Deliveries . Calix shall deliver to Ericsson, at or prior to the Closing, each of the following:
 - (i) the Purchase Price;
 - (ii) the bill of sale dated the Closing Date and in the form attached as Exhibit A1 ("Bill of Sale"), executed by Calix;

[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.

- (iii) the reseller agreement dated the Closing Date, in substantially the form attached as Exhibit B ("Reseller Agreement"), executed by Calix;
- (iv) the intellectual property license agreements dated the Closing Date attached as Exhibit C ("License Agreements"), executed by Calix; and
- (v) a certificate executed on behalf of Calix by an officer and dated the Closing Date ("Calix Closing Certificate") confirming that the conditions set forth in Section 6.2(a) have been satisfied in all respects.
- (b) Ericsson Deliveries. Ericsson shall deliver to Calix, at or prior to the Closing, each of the following:
 - (i) the Bill of Sale, executed by Ericsson;
 - (ii) the Reseller Agreement, in substantially the form attached as Exhibit B, executed by Ericsson;
 - (iii) the License Agreements, executed by Ericsson;
 - (iv) evidence satisfactory to Calix of the consent to assignment of any Person whose consent to assignment, as the case may be, may be required in connection with the Transactions under the Contracts listed or described on <u>Schedule 5.4</u>, in the manner described on such Schedule with respect to each such Contract;
 - (v) a certificate executed on behalf of Ericsson by an officer (" *Ericsson Closing Certificate*") confirming that each of the conditions set forth in Sections 6.3(a), 6.3(c), and 6.3(d) has been satisfied in all respects;
 - (vi) a certificate executed by Ericsson prepared in accordance with Treasury Regulations Section 1.1445-2(b), in form and substance satisfactory to Calix, certifying Ericsson's non-foreign status; and
 - (vii) such other documents as Calix may reasonably request from Ericsson in connection with the Transactions.

1.5 Allocation.

- (a) The Parties shall negotiate in good faith to determine an allocation of the Purchase Price (and, to the extent properly taken into account under the applicable Tax laws, the Assumed Liabilities) among the Assets in accordance with the principles of Section 1060 of the Code and the Treasury Regulations promulgated thereunder and other applicable Tax laws. No later than 30 days following the Closing Date, Calix will deliver to Ericsson the allocation of the Purchase Price (plus the Assumed Liabilities and any other amounts paid by Calix under this Agreement to the extent properly taken into account under the Code and the related Treasury regulations) among the Purchased Assets ("Allocation"). If the Purchase Price is adjusted, the Allocation shall be adjusted as determined by the Parties.
- (b) The Parties agree to act in accordance with the Allocation in the preparation and filing of all Tax Returns (including, without limitation, filing Form 8594 with their United States federal income Tax Return for the taxable year that includes the Closing Date). Not later than 30 days prior to the filing

of their respective Forms 8594 relating to this transaction, each of Calix and Ericsson shall deliver to the other a copy of its Form 8594.

- 1.6 Withholding of Taxes. Calix shall be entitled to deduct and withhold from any amounts payable under this Agreement such amounts as may be required to be deducted or withheld under any provision of federal, state, local or foreign Tax law or under any Legal Requirement. To the extent such amounts are deducted and withheld, such amounts shall be treated for all purposes under this Agreement as having been paid to the Party in respect of whom such deduction and withholding was made. Calix shall use commercially reasonable efforts to inform Ericsson in advance of the Closing if Calix believes that any withholding is required and cooperate with Ericsson in good faith to reduce or eliminate any such withholding under applicable Legal Requirements. Ericsson shall indemnify Calix to the extent any Taxes were required to be withheld from payments from Calix to Ericsson and were not withheld, but were subsequently paid by Calix for Ericsson's account. All rights or claims for refund for any indemnified withholding taxes are transferred to Ericsson.
- 1.7 Consents. This Agreement will not constitute an agreement to assign any Contract or any claim, right or benefit if the agreement to assign or attempt to assign, without the consent of a third party, would constitute a breach of or in any way adversely affect the rights of Calix under that contract. Until such consent is obtained, or if an attempted assignment would be ineffective or would adversely affect the rights of Ericsson or Calix so that Calix would not in fact receive all such rights, Calix and Ericsson will cooperate with each other in any arrangement designed to provide for Calix the benefits of any such Contract. Any transfer or assignment to Calix by Ericsson of any Contract that requires the consent or approval of any third party will, without reducing or adversely affecting the obligations of Ericsson to transfer or assign such Contract set forth in this Article I or the representations and warranties of Ericsson set forth in Article II, be made subject to such consent or approval being obtained, and shall become fully effective when such consent or approval is obtained.
- 1.8 <u>Ericsson Trademarks Excluded</u>. The sale of the Business Assets and any other transaction contemplated hereby does not include any right of Calix or any of its Affiliates to use the word "Ericsson" or the trademark "Ericsson," either standing alone or in combination with the 3 bar logo, or the 3 bar logo standing alone, or any derivative thereof or any combinations of words, trade names, trademarks or company names to the extent it includes such word or trademark. But after the Closing Date and as long as used in accordance with Ericsson's reasonable instructions and guidelines, Calix:
 - (a) shall have the right to use existing packaging, labeling, containers, supplies, advertising materials, technical data sheets and any similar production or office materials bearing such names and trademarks for 90 days after the Closing Date; and
 - (b) shall have the right to use such names and trademarks in existing signs in each case for a period not to exceed 90 days after the Closing Date.

ARTICLE II: REPRESENTATIONS AND WARRANTIES OF ERICSSON

Subject to the disclosures set forth in the Disclosure Schedule (each of which disclosures, in order to be effective, shall clearly indicate the Section and, if applicable, the Subsection of this Article II to which it relates, and each of which shall also be deemed to be representations and warranties made by Ericsson to Calix under this Article II), Ericsson represents and warrants to Calix as follows:

2.1 Organization, Standing and Power. Ericsson is a corporation duly organized, validly existing and in good standing and has full power and authority and possesses all permits necessary or desirable to enable it to own, lease or otherwise hold its properties and to conduct the Business as now being conducted.

Ericsson is qualified to do business and is in good standing in each jurisdiction in which it conducts the Business.

- 2.2 Ownership of Purchased Assets . All Purchased Assets are owned and held by Ericsson directly.
- 2.3 <u>Authority and Enforceability</u>. Ericsson has all requisite power and authority to enter into this Agreement, the Transaction Documents and each of the other contemplated agreements and to consummate the Transactions. The execution and delivery of this Agreement, the Transaction Documents and each of the other contemplated agreements and the consummation of the Transactions, have been authorized by all necessary action on the part of Ericsson. This Agreement has been executed and delivered by Ericsson and constitutes the valid and binding obligation of Ericsson enforceable against Ericsson under its terms, and the Transaction Documents and each other contemplated agreement, after being executed and delivered by Ericsson, will constitute a binding obligation of Ericsson.
- 2.4 Non-Contravention. The execution and delivery of this Agreement by Ericsson does not, the execution and delivery of the Transaction Documents and each of the other contemplated agreements do not and will not, the consummation of the Transactions will not, and the performance by Ericsson of its obligations do not and will not: (a) result in the creation of any Encumbrance on any of the Purchased Assets; or (b) conflict with, or result in any violation of or default under (with or without notice or lapse of time, or both), or give rise to a right of termination, cancellation or acceleration of any obligation or loss of any benefit under, or to increased, additional, accelerated or guaranteed rights or entitlements of any Person under, or require any consent, approval or waiver from any Person under, (i) any provision of the Certificate of Incorporation or other governing documents of Ericsson, as amended to date, (ii) any Contract of Ericsson related to the Business or applicable to any of the Purchased Assets or (iii) any Legal Requirements applicable to Ericsson or to any of the Purchased Assets.

2.5 Governmental Consents; Approvals; Permits.

- (a) The execution, delivery and performance by Ericsson of this Agreement, the Transaction Documents, each of the other contemplated agreements and the consummation of the Transactions do not and will not require any notices, reports, filings, approvals, orders, authorizations, consents, licenses, permits, qualifications or registrations, or qualifications, registrations or applications for new approvals, orders, authorizations, consents, licenses, permits, qualifications or registrations, or waivers of any of the above, to be obtained from or made with, or any notice, statement or other communications to be filed with or delivered to, any Governmental Entity.
- (b) <u>Section 2.5(b)</u> of the Disclosure Schedule sets forth a list of each Governmental Authorization and other third party license, approval, consent or other permission (excluding those related to Patents) (i) under which Ericsson or any of its Affiliates currently operates the Business or holds any interest in any of the Purchased Assets, or (ii) that is required for the operation of the Business by Ericsson or its Affiliates (all of the above approvals, collectively, the "*Ericsson Authorizations*"). All of the Ericsson Authorizations are in full force and effect and Ericsson and its Affiliates are, and have at all times been, in compliance with the requirements of such Ericsson Authorizations.

2.6 Material Contracts.

- (a) Except for this Agreement and the Contracts specifically identified in <u>Section 2.6(a)</u> of the Disclosure Schedule, Ericsson and its Affiliates are neither party to nor bound by any of the following Contracts relating to the Business or the Purchased Assets (each a "*Material Contract*"):
 - (i) other than "shrink wrap" and similar generally available commercial end-user licenses to

Software that is not redistributed with or used in the development or provision of the Ericsson Products that have an individual acquisition and annual support costs of \$10,000 or less, any licenses, sublicenses and other Contracts to which Ericsson is a party and under which Ericsson acquired or is authorized to use any Third Party Software;

- (ii) any continuing Contract for the purchase, sale or license of materials, supplies or equipment, or any services, Technology (other than Software), Intellectual Property Rights, or other Purchased Assets;
- (iii) any Contract, other than the Contracts for the sale of Ericsson Products in the ordinary course of business, under which Ericsson granted or is required to grant to any Person any right or license to make, have made, manufacture, use, sell, offer to sell, import, export, or otherwise distribute any Ericsson Product, with or without the right to sublicense the same;
- (iv) any Contract providing for the development of any Software, content, technology or other Technology, independently or jointly, by or for Ericsson;
- (v) any Contract to license or authorize any third party to manufacture or reproduce any of the Ericsson Products;
- (vi) any Contract relating to the creation of any Encumbrance with respect to any of the Purchased Assets;
- (vii) to the Knowledge of Ericsson, any Contract, imposing any restriction on Ericsson's right or ability, or after the Closing Date, the right or ability of Calix or any of its Affiliates (A) to engage or participate, to compete in any line of business or with any Person or in any area, or which would so limit the freedom of Calix or any of its Affiliates after the Closing Date (including granting exclusive rights or rights of first refusal to license, market, sell or deliver any products or services offered by Ericsson or make use of any Technology or Intellectual Property Right), (B) to acquire any product or other asset or any services from any other Person, to sell any product or other asset to or perform any services for any other Person, or to transact business or deal in any other manner with any other Person, or (C) develop or distribute any Technology or Intellectual Property Rights;
- (viii) to the Knowledge of Ericsson, any Contract granting any exclusive rights of any type or scope to any Person;
- (ix) any Contract granting rights of refusal, rights of first negotiation or similar rights or terms to any Person;
- (x) any Contract for the employment of any Business Employee or consultant of Ericsson or its Affiliates who provides services to the Business or any other type of Contract with any Business Employee or consultant of Ericsson or its Affiliates who provides service to the Business, in each case, that is not immediately terminable by Ericsson or any such Affiliate without cost or Liability, including any Contract requiring it to make a payment to any officer, employee or consultant on account of the Transactions, or any Contract that is entered into in connection with this Agreement;
- (xi) any Contract with any labor union or any collective bargaining agreement or similar contract which covers or affects Business Employees;

- (xii) any confidentiality, secrecy or non-disclosure Contract, other than any such Contract entered into with customers and distributors in the ordinary course of business;
- (xiii) any settlement agreement entered into by Ericsson Inc. that would impact Calix;
- (xiv) any Contract under which rights of any third party are triggered or become exercisable, or under which any other consequence, result or effect arises, in connection with or as a result of the execution of this Agreement, the Transaction Documents or the consummation of the Transactions, either alone or in combination with any other event;
- (b) Ericsson or the Affiliate of Ericsson party to such Material Contract has performed all of the obligations required to be performed by it and is entitled to all benefits under, is not alleged to be in default in respect of, any Material Contract. There exists no default or event of default or event, occurrence, condition or act, with respect to Ericsson or such Affiliate, or to Ericsson's Knowledge, with respect to any other contracting party, which, with the giving of notice, the lapse of time or the happening of any other event or condition, would reasonably be expected to (i) become a default or event of default under any Material Contract or (ii) give any third party (A) the right to declare a default or exercise any remedy under any Material Contract, (B) the right to a rebate, chargeback, refund, credit, penalty or change in delivery schedule under any Material Contract, (C) the right to accelerate the maturity or performance of any obligation of Ericsson or the Affiliate of Ericsson under any Material Contract, or (D) the right to cancel, terminate or modify any Material Contract. To the Knowledge of Ericsson, neither Ericsson nor any Affiliate has received any notice or other communication regarding any actual or possible violation or breach of, default under, or intention to cancel or modify any Material Contract. Ericsson has not terminated or waived any rights, benefits or claims of Ericsson or its Affiliates under any Material Contract.
- (c) Each of the Material Contracts is in full force and effect and constitutes a legal, valid and binding agreement of Ericsson or any Affiliate of Ericsson, and Ericsson has no knowledge that any Material Contract is not a legal, valid and binding agreement of any other party, subject only to the effect, if any, of (i) bankruptcy, insolvency, reorganization, fraudulent transfer, moratorium or other similar laws relating to or affecting the rights or remedies of creditors or (ii) general principles of equity, whether considered in a proceeding in equity or at law (including the possible unavailability of specific performance or injunctive relief).
- 2.7 <u>Title to Purchased Assets</u>; <u>Encumbrances</u>. Ericsson has good title to the Purchased Assets, free and clear of all Encumbrances. Ericsson has the complete and unrestricted power and unqualified right to sell the Purchased Assets to Calix.
- 2.8 <u>Litigation</u>. There is no private or governmental Proceeding pending, or, to the Knowledge of Ericsson, threatened against Ericsson, its Affiliates or any of their respective officers or employees relating to the current conduct of the Business, nor, to the Knowledge of Ericsson, is there any reasonable basis for any such action, suit, Proceeding, claim, mediation, arbitration or investigation; (b) there is no Order against or applicable to Ericsson, its Affiliates, the Business, the Purchased Assets or, to the Knowledge of Ericsson, any officers or employees of Ericsson or its Affiliates related to the Business; (c) there is no private or governmental Proceeding pending, or, to the Knowledge of Ericsson, threatened based upon Ericsson entering into this Agreement, the Transaction Documents, or any of the other Transactions, nor, to the Knowledge of Ericsson, is there any reasonable basis for any Person to assert a claim against Ericsson based upon Ericsson entering into this Agreement, the Transaction Documents or any of the other Transactions.
- 2.9 Restrictions on Business Activities . To the Knowledge of Ericsson, there is no Contract or Order

binding upon Ericsson or its Affiliates which has or would reasonably be expected to have, whether before or after consummation of the Transactions, the effect of prohibiting or impairing the conduct of the Business by Ericsson or its Affiliates or, after the Closing, Calix and its Affiliates, as currently conducted by Ericsson or as presently proposed to be conducted by Calix.

- 2.10 <u>Compliance with Laws</u>. To the Knowledge of Ericsson, Ericsson and its Affiliates have complied in all material respects with and are not in violation of any Legal Requirement, and neither Ericsson nor any Affiliate of Ericsson has received any notice of, or other communication regarding, any pending or threatened Proceeding or violation relating to any Legal Requirement, in each case with respect to the conduct of the Business, or the ownership or operation of the Business or the Purchased Assets.
- 2.11 <u>Intellectual Property</u>. With respect to Intellectual Property Rights used in connection with the Business and the Purchased Assets:
 - (a) Section 2.11(a)(i) of the Disclosure Schedule contains, to the Knowledge of Ericsson, an accurate and complete list of all Ericsson Proprietary Software incorporated in, provided with, used in or otherwise necessary for the use, support, manufacture and maintenance of the Ericsson Products, including all Software that Ericsson and its Affiliates provide or make available to their customers. Section 2.11 (a)(ii) of the Disclosure Schedule contains, to the Knowledge of Ericsson, an accurate and complete list of all third party components of the Ericsson Products (including Third Party Software), which is distributed or made available to Third Parties as incorporated in or provided with the Ericsson Products.
 - (b) Section 2.11(b) of the Disclosure Schedule contains to the knowledge of Ericsson, a list of all unregistered trademarks related to the Business.
 - (c) To the Knowledge of Ericsson, Ericsson has sufficient right, title and interest in and to all the Ericsson-Owned Intellectual Property and to the Ericsson Licensed Intellectual Property to grant the licenses set forth in the Copyright License Agreement.
 - (d) By the Closing Date or within 90 days thereof, Ericsson shall cooperate with Calix to facilitate the transfer of rights in and to the Third Party Intellectual Property (other than Third Party Intellectual Property not already in the possession of Ericsson or not sublicensable or transferable by Ericsson to Calix) necessary to enable Calix to design, manufacture, license and support of all Ericsson Products currently in production.
 - (e) To the Knowledge of Ericsson, the Ericsson-Owned Intellectual Property, Third Party Intellectual Property and Ericsson Licensed Intellectual Property constitutes all of the Technology and Intellectual Property Rights necessary for Ericsson to manufacture, distribute and sell the Ericsson Products as of the Closing Date.
 - (f) Neither Ericsson nor any Affiliate of Ericsson is, with respect to the Ericsson Products, infringing, misappropriating, interfering with, or otherwise violating (collectively, "Infringing"), any Intellectual Property Right of any Person, and no claim of such infringement, misappropriation, or violation is pending against Ericsson, or any Affiliate of Ericsson. No Ericsson-Owned Intellectual Property and no Ericsson Licensed Intellectual Property used in connection with the Business or the Ericsson Products is subject to any Proceeding or outstanding Order that restricts the use, transfer or licensing by Ericsson or may affect the validity, use or enforceability of the Ericsson-Owned Intellectual Property or Ericsson Licensed Intellectual Property.

 Neither Ericsson nor any Affiliate of Ericsson has received any notice alleging, or otherwise has Knowledge of (i) the invalidity, or limitation on Ericsson's right to use, with respect to any of the Ericsson-Owned Intellectual Property

or the Ericsson Licensed Intellectual Property directly related to the Ericsson Products or of (ii) the alleged infringement, misappropriation or breach of any rights of others by Ericsson or any Affiliate of Ericsson which directly relate to the Ericsson Products. No Person has notified Ericsson or any of its Affiliates that it is claiming any ownership of or right to use any Ericsson-Owned Intellectual Property or any Ericsson Licensed Technology related directly to the Ericsson Products.

- (g) Ericsson has taken all reasonable steps necessary or appropriate (including, entering into written confidentiality and nondisclosure agreements with officers, directors, subcontractors, employees, licensees and customers in connection with the Purchased Assets or the Business) to safeguard, maintain the confidentiality of, and otherwise protect and enforce its rights in all proprietary information that Ericsson or its Affiliates hold, or purport to hold, as a trade secret. In particular, to the Knowledge of Ericsson, (i) there has been no misappropriation of any trade secrets or other material confidential Intellectual Property Rights or Technology directly related to the Ericsson Products; (ii) to the Knowledge of Ericsson, no employee, independent contractor or agent of Ericsson or an Affiliate of Ericsson has misappropriated any trade secrets of any other person in the course of performance as an employee, independent contractor or agent of the Business; and (iii) to the Knowledge of Ericsson, no employee, independent contractor or agent of Ericsson is in default or breach of any term of any employment agreement, nondisclosure agreement, assignment of invention agreement or similar agreement or contract directly related to the Ericsson Products.
- (h) Except as set forth on Section 2.11(h) of the Disclosure Schedule, Ericsson maintains (i) machine-readable copies of Ericsson Proprietary Software and the Third Party Software used in connection with, or otherwise related to, the Business and (ii) reasonably complete technical documentation or user manuals for all releases or versions currently in use by Ericsson, its Affiliates or their respective customers. Such machine-readable copy conforms to the corresponding source code listing in all material respects.
- (i) Except as set forth on Section 2.11(i) of the Disclosure Schedule, to the Knowledge of Ericsson, Ericsson Proprietary Software and the Third Party Software used in connection with, or otherwise related to, the Business do not contain and are not in any manner derived from any open-source Software and do not contain any virus, worm, Trojan horse, or similar malicious code. No source code for any Ericsson Proprietary Software used in connection with, or otherwise related to, the Business has been delivered, licensed, or made available to any escrow agent or other Person other than those Persons listed on Section 2.11(i) of the Disclosure Schedule. Ericsson has no obligation (whether present, contingent, or otherwise) to deliver, license, or make available the source code for any Software to any escrow agent or other Person. No event has occurred, and no circumstance or condition exists, that (with or without notice or lapse of time) will, or could reasonably be expected to, result in the delivery, license, or disclosure of any source code for any Ericsson Proprietary Software used in connection with, or otherwise related to, the Business to any other Person.
- (j) Except as set forth on Section 2.11(j) of the Disclosure Schedule, to the Knowledge of Ericsson, all Ericsson Products comply in all material respects with all representations, warranties and other requirements applicable to such products, including those specified in all customer contracts and all specifications published by Ericsson or its Affiliates. To the Knowledge of Ericsson, the Ericsson Products may be manufactured in accordance with their specifications in substantially the same manner as currently conducted in the Business. The need for the provision of bug fixes under a support agreement shall not constitute a violation of this Representation.
- (k) The Third Party Software (including, for purposes of this clause, all commercially available "off- the-shelf" or "shrink-wrapped" software) that constitutes a component of any Ericsson Product is

generally commercially available for license at commercially reasonable rates and on commercially reasonable terms.

- (1) To the Knowledge of Ericsson, personnel, including employees, agents, consultants and contractors, who have contributed to or participated in the conception and development of Ericsson-Owned Intellectual Property have executed and are legally bound by a nondisclosure agreement applicable to Ericsson's confidential information to which Ericsson is the beneficiary either directly or indirectly (for example, in the case of an employee of a consultant or independent contractor who has entered into such a nondisclosure agreement with Ericsson); and have executed appropriate instruments of assignment in favor of Ericsson as assignee that have conveyed to Ericsson complete and exclusive ownership of all Technology developed for Ericsson.
- (m) The fact (if applicable) that any funding, facilities, or personnel of any Governmental Entity were used, directly or indirectly, to develop or create, in whole or in part, any Ericsson-Owned Intellectual Property within the Purchased Assets used in connection with, or otherwise related to, the Business, shall not have any material effect on Ericsson's right, title and interest in and to all the Ericsson-Owned Intellectual Property within the Purchased Assets and to the Ericsson Licensed Intellectual Property.
- (n) The fact (if applicable) that Ericsson or any Affiliate of Ericsson is or ever was a member or promoter of, or a contributor to, any industry standards body or similar organization that could require or obligate Ericsson or any such Affiliate to grant or offer to any other Person any license or right to or otherwise impair Ericsson's control of any Ericsson-Owned Intellectual Property within the Purchased Assets shall not have any material effect on Ericsson's right, title and interest in and to all the Ericsson-Owned Intellectual Property within the Purchased Assets and to the Ericsson Licensed Intellectual Property,
- (o) To the Knowledge of Ericsson, neither the execution, delivery, or performance of this Agreement or any Transaction Documents, nor the consummation of any of the Transactions will, with or without notice or the lapse of time, result in, or give any other Person the right or option to cause or declare, (A) a loss of rights in, or Encumbrance on, or the acceleration of any rights with respect to any Ericsson-Owned Intellectual Property within the Purchased Assets, Ericsson Licensed Intellectual Property or Third Party Intellectual Property used in connection with, or otherwise related to, the Products (B) a breach of, or termination or other right arising under, any Contract required to be listed on Section 2.6(a)(iii) of the Disclosure Schedule; (C) the release, disclosure, or delivery of any Ericsson-Owned Intellectual Property within the Purchased Assets by or to any escrow agent or other Person; or (D) the grant, assignment, or transfer to any other Person of any license or other right or interest under, to, or in any of Ericsson-Owned Intellectual Property within the Purchased Assets. All Ericsson-Owned Intellectual Property will as of the Closing Date be transferable (to the extent they are within the definition of Purchased Assets) or licensable by Calix, as may be applicable.

2.12 Patents and Patent Contracts.

- (a) Except as disclosed on Schedule 2.12, to the Knowledge of Ericsson, there is no (i) Third Party Patent-Ericsson Product; (ii) Third Party Patent Contract; or (iii) Third Party Patent Proceeding. Other than as set forth in the immediately preceding sentence, Ericsson makes no representation or warranty hereunder with respect to any Patent or any Patent Contract.
- (b) Ericsson, makes no representation or warranty that the Ericsson Product Patents constitutes all of the rights necessary for the manufacture, distribution and sale of the Ericsson Products or that Ericsson has all of the rights under any Patents or under any Patent Contracts rights necessary for the

manufacture, distribution and sale of the Ericsson Products.

- (c) Ericsson makes no representation or warranty that the design, manufacture, distribution, sale or use of the Ericsson Products or Technology embodied in the Ericsson Products are Infringing or might infringe any Third Party Patent.
- (d) To the Knowledge of Ericsson, personnel, including employees, agents, consultants and contractors, who have contributed to or participated in the conception and development of Ericsson Product Patents have executed appropriate instruments of assignment in favor of Ericsson as assignee that have conveyed to Ericsson complete and exclusive ownership of the Ericsson Product Patents.
- 2.13 <u>Tax Matters</u>. To the Knowledge of Ericsson there are no existing or potential Tax-related claims, liabilities, liens, penalties, agreements or other Tax issues that would preclude the execution and delivery of the Transaction Documents and the consummation of the contemplated Transactions or Ericsson's right to grant the licenses set forth in the Patent License Agreement, or that are expected to result in any Encumbrance for Taxes on the Assets or any liability of Calix therefor.
- 2.14 Employee Matters. None of the execution and delivery of this Agreement and the Transaction Documents, the consummation of the Transactions or any related termination of employment or service will, individually or together with the occurrence of some other event, (i) result in any payment (including severance, unemployment compensation, golden parachute, bonus or otherwise) becoming due to any Business Employee, (ii) materially increase or otherwise enhance any benefits otherwise payable by Ericsson or an Affiliate of Ericsson to a Business Employee, (iii) result in the acceleration of the time of payment or vesting of any such benefits to a Business Employee, except as required under Section 411(d)(3) of the Code, (iv) increase the amount of compensation due to any Business Employee, or (v) result in the forgiveness in whole or in part of any outstanding loans made by Ericsson or an Affiliate of Ericsson to any Business Employee.
 - (a) Section 2.13(a) of the Disclosure Schedule sets forth a complete list as of the Agreement Date of all severance Contracts and employment Contracts with Business Employees to which Ericsson is a party or by which Ericsson is bound. Except as set forth in Section 2.13(a) of the Disclosure Schedule, with respect to the Business, neither Ericsson nor any Affiliate of Ericsson has been a party to or bound by any collective bargaining agreement or other labor union Contract, no collective bargaining agreement is being negotiated by Ericsson or its Affiliates and neither Ericsson nor any Affiliate of Ericsson has any duty to bargain with any labor organization. There is no pending demand for recognition or any other request or demand from a labor organization for representative status with respect to any Business Employee. Ericsson has no Knowledge of any activities or proceedings of any labor union or to organize the Business Employees. There is no labor dispute, strike or work stoppage against Ericsson or its Affiliates pending or, to the Knowledge of Ericsson, threatened which may interfere with the Business.
 - (b) No Business Employee is in violation of any term of any employment agreement, patent disclosure agreement, non-competition agreement, or any restrictive covenant to a former employer relating to the right of any such Business Employee to be employed by Ericsson because of the nature of the business conducted or presently proposed to be conducted by Ericsson or to the use of trade secrets or proprietary information of others. The employment of each of the Business Employees is "at will" and neither Ericsson nor any Affiliate of Ericsson has any obligation to provide any particular form or period of notice prior to terminating the employment of any of the Business Employee, except as may be required by applicable law. As of the Agreement Date, Ericsson has not, (i) entered into any Contract that obligates or purports to obligate Ericsson or its Affiliates to make an offer of employment to any present or former Business Employee or consultant of Ericsson or its

Affiliates who provides service to the Business or (ii) promised or otherwise provided any assurances (contingent or otherwise) to any such present or former Business Employee or consultant of Ericsson or its Affiliates of any terms or conditions of employment with Ericsson or its Affiliates following the Closing Date.

- 2.15 <u>Books and Records</u>. Ericsson has provided to Calix complete copies of each document that has been requested by Calix or its counsel in connection with their legal and accounting review of the Business (other than any such document that does not exist or is not in Ericsson's possession or subject to its control). Ericsson has provided to Calix complete and correct copies of all documents identified on the Disclosure Schedule, and all permits, orders, authorizations and consents issued by any Governmental Entity with respect to the Business or the Purchased Assets, and all applications for such permits, orders, authorizations and consents. The books, records and accounts of Ericsson (a) are complete in all material respects, (b) have been maintained in accordance with reasonable business practices on a basis consistent with prior years and (c) are stated in reasonable detail and accurately and fairly reflect the transactions and dispositions of the assets and properties of the Business.
- 2.16 <u>Customers and Suppliers.</u> To the Knowledge of Ericsson, neither Ericsson nor any Affiliate of Ericsson has any outstanding material disputes with any customer or distributor relating to the Business or Purchased Assets. Ericsson has no Knowledge (i) of any material dissatisfaction on the part of any such customer or distributor, (ii) that any such customer or distributor intends to cease or materially diminish the use of the Ericsson Products, or (iii) of any fact or circumstance that could reasonably be expected to cause such customers or distributors to cease or materially diminish the use of the Ericsson Products. Neither Ericsson nor any Affiliate of Ericsson has received any information from any customer or distributor that such customer or distributor shall not continue as a customer or distributor of Calix after the Closing or that such customer or distributor intends to terminate or materially modify existing Contracts with Ericsson or its Affiliates (or Calix). Neither Ericsson nor any Affiliate of Ericsson has had any of the Ericsson Products returned by a purchaser except for normal warranty returns consistent with past history and those returns that would not result in a reversal of any revenue by Ericsson.
 - (a) To the Knowledge of Ericsson, neither Ericsson nor any Affiliate of Ericsson has any outstanding material dispute concerning products or services provided by any supplier relating to the Business or Purchased Assets. Ericsson has no knowledge (i) of any material dissatisfaction on the part of any such supplier, (ii) that any such supplier intends to cease or materially diminish the provision of products or services to Ericsson or its Affiliates, or (iii) of any fact or circumstance that could reasonably be expected to cause such supplier to cease or materially diminish the provision of products or services to Ericsson or its Affiliates. Neither Ericsson nor any Affiliate of Ericsson has received any information from any supplier that such supplier shall not continue as a supplier to Calix after the Closing or that such supplier intends to terminate or materially modify existing Contracts with Ericsson or its Affiliates (or Calix). Ericsson has access, on commercially reasonable terms, to all products and services reasonably necessary to carry on the Business, and Ericsson has no Knowledge of any reason why it will not continue to have such access on commercially reasonable terms.
- 2.17 <u>Products; Warranties; Purchase Commitments.</u> There are no pending or, to Ericsson's Knowledge, threatened claims, and neither Ericsson nor any Affiliate of Ericsson have been notified in writing of any claims, relating to any warranty obligations, failure to meet warranties or material product returns, other than normal returns in the ordinary course of business. There are no pending, or, to Ericsson's Knowledge, threatened, and neither Ericsson nor any Affiliate of Ericsson has been notified in writing of, any material claims relating to product liability against or involving Ericsson or its Affiliates or any Ericsson Product.

- (a) All of the Ericsson Products comply in all material respects with all authorizations, permits or licenses of any Governmental Entity and all Legal Requirements.
- (b) No outstanding purchase or outstanding lease commitment of Ericsson related to the Business presently is in excess of the normal, ordinary and usual requirements of the Business. There is no outstanding bid, proposal, Contract or unfilled order which relates to the Purchased Assets which will or would, if accepted, have an Ericsson Material Adverse Effect.
- 2.18 <u>Finders' Fees; Transaction Expenses</u>. Neither Ericsson nor any Affiliate of Ericsson is obligated for the payment of any fees or expenses of any investment banker, broker, advisor, finder or similar party in connection with the origin, negotiation or execution of this Agreement, any of the other agreements contemplated to which Ericsson is or will be a party, or in connection with the Transactions.
- 2.19 <u>Representations Complete</u>. To the Knowledge of Ericsson, no representation, warranty or statement contained in this Agreement and its schedules, appendices and exhibits, contains or shall contain any untrue statement or a material fact or omits or shall omit any material fact necessary to make the information contained not misleading in light of the circumstances in which they were made.

ARTICLE III: REPRESENTATIONS AND WARRANTIES OF CALIX

Calix represents and warrants, to and for the benefit of Ericsson, as follows:

- 3.1 <u>Organization and Standing</u>. Calix is a corporation organized, validly existing and in good standing under the laws of the State of Delaware and has full power and authority and possesses all permits necessary or desirable to enable it to own, lease or otherwise hold its properties and to conduct its business as now being conducted and as currently proposed by it to be conducted.
- 3.2 <u>Authority and Enforceability</u>. Calix has all requisite entity power and authority to enter into this Agreement, the Transaction Documents and each of the other contemplated agreements and to consummate the contemplated transactions. The execution and delivery of this Agreement, the Transaction Documents and each of the other contemplated agreements and the consummation of the contemplated transactions and have been authorized by all necessary corporate action on the part of Calix. This Agreement has been executed and delivered by Calix and constitutes the valid and binding obligation of Calix enforceable against it in accordance with its terms, and the Transaction Documents and each other contemplated agreement, after being executed and delivered by Calix will constitute a valid and binding obligation of Calix, enforceable against it in accordance with its terms.
- 3.3 Non-Contravention. The execution and delivery of this Agreement, the Transaction Documents, and each of the other contemplated agreements, the consummation of the contemplated transactions, and the performance by Calix of its obligations do not and will not conflict with, or result in any violation of or default under (with or without notice or lapse of time, or both), or give rise to a right of termination, cancellation or acceleration of any obligation or loss of any benefit under, or to increased, additional, accelerated or guaranteed rights or entitlements of any Person under, or require any consent, approval or waiver from any Person under, (i) any provision of the Certificate of Incorporation or other equivalent organizational or governing documents of Calix, or (ii) any Legal Requirements applicable to Calix or to any of its material properties or assets.
- 3.4 Government Consent. No Approval of any Governmental Entity is required by or with respect to Calix in connection with the execution and delivery of this Agreement or the consummation of the Transactions, except for such consents, authorizations, filings, approvals, notices and registrations which, if not obtained or made, would not be material to Calix's ability to consummate the Transactions or to

perform its obligations under this Agreement.

3.5 <u>Finders' Fee; Transaction Expenses</u>. Neither Calix nor any Affiliate of Calix is obligated for the payment of any fees or expenses of any investment banker, broker, advisor, finder or similar party in connection with the origin, negotiation or execution of this Agreement, any of the other agreements contemplated to which Calix is or will be a party, or in connection with the Transactions.

ARTICLE IV: CONDUCT PRIOR TO THE CLOSING

- 4.1 Conduct of Business of Ericsson. During the Pre-Closing Period:
 - (a) Ericsson shall, and shall cause its Affiliates to, conduct the Business in the ordinary course of business in substantially the same manner as previously conducted and in compliance with all Legal Requirements;
 - (b) Ericsson shall, and shall cause its Affiliates to, (i) as related to the Business or the Purchased Assets, pay all of its debts and Taxes when due, subject to good faith disputes over such debts, (ii) as related to the Purchased Assets, pay or perform its other obligations when due, (iii) as related to the Business, use commercially reasonable efforts consistent with past practice and policies to collect accounts receivable when due and not extend credit outside of the ordinary course of business, (iv) sell Ericsson Products consistent with past practices as to license, service and maintenance terms, incentive programs, and in accordance with GAAP requirements as to revenue recognition, and (v) use its commercially reasonable efforts consistent with past practice and policies to preserve intact the Purchased Assets, keep available the services of the Business Employees and preserve its relationships with customers, suppliers, distributors, licensors, licensees, and others having business dealings with the Business, to the end that the Purchased Assets, including goodwill, shall be unimpaired at the Closing;
 - (c) Ericsson shall promptly notify Calix of any change, occurrence or event not in the ordinary course of business, or of any change, occurrence or event which, individually or in the aggregate with any other changes, occurrences and events, would reasonably be expected to be materially adverse to the Business or the Purchased Assets or cause any of the conditions to Closing set forth in <u>Article VI</u> not to be satisfied; and
 - (d) Ericsson shall assure that each of its Contracts related to the Business entered into after the Agreement Date will not require the procurement of any consent, waiver or novation or provide for any change in the obligations of any party in connection with, or terminate as a result of the consummation of, the Transactions, and shall give reasonable advance notice to Calix prior to allowing any Material Contract or right to lapse or terminate by its terms.
- 4.2 <u>Restrictions on Conduct of Business of Ericsson</u>. During the Pre-Closing Period, Ericsson shall not, and shall not permit any of its Affiliates to, do, cause or permit any of the following:
 - (a) <u>Material Contracts</u>. Enter into any Contract that would constitute a Material Contract, other material Contract or a Contract requiring a novation or consent in connection with the Transactions, or violate, terminate, amend, or otherwise modify (including by entering into a new Contract with such party or otherwise) or waive any of the terms of any of its Material Contracts;
 - (b) Employees; Consultants; Independent Contractors. (i) Hire any additional Business Employees, or any consultants or independent contractors to provide services related to the Business, (ii) terminate without cause the employment, change the title, office or position, or materially reduce the

responsibilities of any management, supervisory or other key personnel of Ericsson who are Business Employees, (iii) enter into, amend or extend the term of any employment or consulting agreement with any Business Employee, or consultant or independent contractor who provides services to the Business, or (iv) enter into any Contract with a labor union or collective bargaining agreement which affects a Business Employee. Ericsson shall notify Calix of the termination of an Employee pursuant to (ii) above;

- (c) Intellectual Property; Ericsson Patents. (i) Transfer or license from any Person any rights to any Technology within the Purchased Assets, or transfer or license to any Person any rights to any Ericsson-Owned Intellectual Property within the Purchased Assets, or transfer or provide a copy of any Ericsson Proprietary Software, or exclusively license, assign or transfer any Ericsson Licensed Intellectual Property used in the Business, to any Person (including any current or former employee or consultant of Ericsson or any contractor or commercial partner of Ericsson) (other than providing access to Ericsson Proprietary Software to current employees and consultants of Ericsson involved in the development of the Ericsson Products on a need to know basis, consistent with past practices); (ii) dispose of or permit to lapse any rights in, to or for the use of any Technology within the Purchased Assets or Intellectual Property Rights within the Purchased Assets used in connection with, or otherwise related to, the Business; or (iii) transfer or exclusively license or sublicense to any Person any rights to any Third Party Intellectual Property necessary for the conduct of or primarily related to the Business; or (iv) assign any rights in, or exclusively license, the Ericsson Product Patents, unless such transfer is made subject to Calix's rights to be granted in the attached License Agreement.
- (d) Exclusive Rights and Most Favored Party Provisions. Enter into or amend any agreement, excluding Patent Contracts, under which any other party is granted exclusive rights or "most favored party" rights of any type or scope with respect to any of the Ericsson Products, the Business, Technology within the Purchased Assets, or containing any non-competition covenants or other restrictions relating to its conduct of the Business or that could potentially impact Calix's business activities;
- (e) <u>Dispositions</u>. Sell, lease, license or otherwise dispose of or encumber any of the Purchased Assets, other than sales of Ericsson Products in the ordinary course of business and consistent with past practice, or enter into any Contract with respect to the above;
- (f) <u>Insurance</u>. Materially change the amount of any insurance coverage related to the Purchased Assets;
- (g) <u>Termination or Waiver</u>. Terminate or waive any right of substantial value related to the Purchased Assets;
- (h) <u>Severance Agreement</u>. Enter into or amend any Contract providing for the granting of any severance, retention or termination pay, or the acceleration of vesting or other benefits, to any Business Employee or consultant who provides service to the Business;
- (i) <u>Lawsuits</u>; <u>Settlements</u>. (i) Commence a lawsuit or (ii) settle or agree to settle any pending or threatened Proceeding, lawsuit or other dispute, in each case related to the Purchased Assets;
- (j) <u>Encumbrances</u>. Place or allow the creation of any Encumbrance (other than a Permitted Encumbrance) on any of the Purchased Assets;

ARTICLE V: CERTAIN COVENANTS OF THE PARTIES

- 5.1 Confidentiality; Public Disclosure. The Parties acknowledge that Calix, Inc. and Ericsson Inc. have previously executed a Non-Disclosure Agreement dated January 6, 2012 ("Confidentiality Agreement") which shall continue in full force and effect and shall apply to all Confidential Information (as defined in that Confidentiality Agreement) disclosed by either Party in the course of the transactions contemplated by this Agreement. From and after the Agreement Date, neither Party or its Representatives shall, directly or indirectly, issue any press release or other public statement relating to the terms of this Agreement or the Transactions or use the other Party's name or refer to the other Party directly or indirectly in connection with the contemplated Transaction in any media interview, advertisement, news release, press release or professional or trade publication, or in any print media, whether or not in response to an inquiry, without the prior written approval of the other Party, which shall not be unreasonably withheld, conditioned or delayed; provided, however, that either Party may make any public disclosure as required by Legal Requirements or any listing agreement with any national or regional securities exchange or market upon reasonable prior notice to the other Party.
- 5.2 <u>Regulatory Approvals</u>. Either Party shall promptly execute and file, or join in the execution and filing of, any application, notification or other document that may be necessary in order to obtain the authorization, approval or consent of any Governmental Entity, whether federal, state, local or foreign, which may be required, or which the other Party may reasonably request, in connection with the consummation of the Transactions. Each Party shall use commercially reasonable efforts to obtain, and to cooperate with the other Party to promptly obtain, all such authorizations, approvals and consents and shall pay any associated filing fees payable with respect to such authorizations, approvals and consents. Each Party shall promptly inform the other Party of any material communication with any Governmental Entity regarding any of the Transactions. If either Party or its Affiliates receives any formal or informal request for supplemental information or documentary material from any Governmental Entity with respect to the Transactions, then the other Party shall make, or cause to be made, as soon as reasonably practicable, a response in compliance with such request
- 5.3 <u>Reasonable Efforts</u>. Each Party agrees to use its commercially reasonable efforts, and to cooperate with the other Party, to take, or cause to be taken, all actions, and to do, or cause to be done, all things necessary, appropriate or desirable to consummate and make effective, in the most expeditious manner practicable, the Transactions, including the satisfaction of the respective conditions set forth in <u>Article VI</u>, and including to execute and deliver such other instruments and do and perform such other acts and things as may be necessary or reasonably desirable for effecting completely the consummation of the Transactions.

5.4 Third Party Consents; Notices.

- (a) Ericsson shall obtain prior to the Closing, and deliver to Calix at or prior to the Closing, all notices, consents, waivers and approvals under each Contract listed or described on Schedule 5.4.
- (b) Ericsson shall give all notices and other information required to be given to the employees of Ericsson, any collective bargaining unit representing any group of employees of Ericsson, and any government authority under the WARN Act, the National Labor Relations Act, as amended, the Code, COBRA and other Legal Requirements in connection with the Transactions.

5.5 Access to Information .

(a) During the Pre-Closing Period, Ericsson shall afford Calix and its Representatives, reasonable access during business hours to (A) all of Ericsson's properties, books, Contracts, intellectual property, employees and records related to the Business or the Purchased Assets, and (B) all other information concerning the Business or the Purchased Assets as Calix may reasonably request, and

Ericsson shall respond promptly to Calix's requests concerning the above.

- (b) Subject to compliance with Legal Requirements, during the Pre-Closing Period, Ericsson shall confer from time to time as requested by Calix with one or more Representatives of Calix to discuss any material changes or developments in the operational matters of the Business.
- (c) No information or knowledge obtained in any investigation under this <u>Section 5.5</u> shall affect or be deemed to modify any representation or warranty or the conditions to the obligations of the Parties to consummate the Transactions.

5.6 Employees and Contractors.

- (a) Calix intends to extend offers of employment to the Business Employees listed on Schedule 5.6 (such employees who accept Calix's offers of employment and commence employment with Calix or an Affiliate of Calix on the Closing Date are referred to as the "Transferred Employees"), which offers shall be on terms and conditions which Calix shall determine in its sole discretion. Ericsson shall accept the resignation of the employment of all Transferred Employees immediately prior to the Closing and shall cooperate with and use its reasonable efforts to assist Calix in its efforts to secure satisfactory employment arrangements with those employees of Ericsson to whom Calix makes offers of employment. In the event that one or more Employees listed in Schedule 6.4 elect not to become Calix employees effective at the Closing Date, Ericsson shall follow the remedies and actions outlined in Appendix D of this agreement.
- (b) Ericsson shall be solely responsible for all obligations and liabilities of the Ericsson Employee [***].
- (c) Nothing contained in this Agreement shall confer upon any Transferred Employee any right with respect to continuance of employment by Calix, nor interfere with the right of Calix to terminate the employment of any Transferred Employee at any time, with or without cause.
- (d) Calix, as the successor-in-interest entity to Ericsson for purposes of immigration-related issues for the Transferred Employees, will acquire the immigration-related liabilities of such employees and will defend, indemnify and hold Ericsson harmless from any U.S. immigration claims, damages, or liability that may accrue after the Closing Date with the exception of liabilities related to fraud and misrepresentation that occurred prior to the transfer.
- (e) This Section 5.6 shall not confer any rights, benefits or remedies upon, and shall not be enforceable by any Transferred Employee or any other Person, other than Calix and its successors and assigns. Nothing contained in this Section 5.6 shall in any way alter the at-will employment status or any other terms of employment of any Transferred Employee. [***] The provisions of this Section 5.6 are for the sole benefit of Calix and Ericsson and nothing in this Section 5.6, expressed or implied, is intended or shall be construed to constitute an amendment of any Ericsson Employee Plan or Calix plan, arrangement, policy or agreement (or an undertaking to amend any such plan, arrangement, policy or agreement) or other compensation and benefits plan maintained for or provided to Business Employees, including Transferred Employees, prior to, on or following the Closing Date.

[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.

5.7 Tax Matters.

- (a) Each Party agrees to furnish to the other, upon request, as promptly as practicable, such information and assistance relating to the Purchased Assets, including, without limitation, access to Books and Records, as is reasonably necessary for the filing of all Tax Returns by either Party, the making of any election relating to Taxes, the preparation for any audit by any taxing authority and the prosecution or defense of any claim, suit or proceeding relating to any Tax. Ericsson shall retain all Books and Records with respect to Taxes pertaining to the Purchased Assets for a period of at least six years following the Closing Date. Calix and Ericsson further agree, upon request, to use their best efforts to obtain any certificate or other document from any Governmental Entity or any other Person as may be necessary to mitigate, reduce or eliminate any Tax that could be imposed (including, but not limited to, with respect to the contemplated transactions).
- (b) <u>Allocation of Straddle Period Property Taxes</u>. Ericsson shall be responsible for and shall pay when due all Property Taxes levied with respect to the Purchased Assets attributable to the Pre-Closing Tax Period, and Calix shall be responsible for and shall pay when due all Property Taxes levied with respect to the Purchased Assets attributable to the Post-Closing Tax Period. All Property Taxes levied with respect to the Purchased Assets for the Straddle Period shall be apportioned between the Pre-Closing Tax Period and the Post-Closing Tax Period, as follows: the portion allocable to the Pre-Closing Tax Period shall be deemed to be the amount of such Tax for the entire Straddle Period multiplied by a fraction the numerator of which is the number of days in the Tax period ending on the Closing Date and the denominator of which is the number of days in the entire Straddle Period. Upon receipt of any bill for such Taxes relating to the Purchased Assets, Each Party shall present a statement to the other setting forth the amount of reimbursement to which each is entitled under this <u>Section 5.7(b)</u> together with such supporting evidence as is reasonably necessary to calculate the proration amount. The proration amount shall be paid by the Party owing it to the other within 10 days after delivery of such statement. In the event that either Party makes any payment for which it is entitled to reimbursement under this <u>Section 5.7(b)</u>, the other Party shall make such reimbursement no later than 10 days after the presentation of a statement setting forth the amount of reimbursement to which the presenting Party is entitled along with such supporting evidence as is reasonably necessary to calculate the amount of reimbursement. Calix shall, at its election, be entitled to recover any Property Taxes for which Ericsson is liable under this Section 5.7(b).
- (c) <u>Certain Taxes</u>. The Parties agree that the Purchase Price is exclusive of any transfer, documentary, sales, use, stamp, registration, value added and other substantially similar Taxes and fees (including any penalties and interest) incurred in connection with this Agreement (collectively, "*Transfer Taxes*"). Each Party shall be responsible for the payment of 50% of any Transfer Taxes that may be imposed upon or payable or collectible or incurred in connection with this Agreement or the transactions contemplated herein, or that may be imposed upon or payable or collectible or incurred in connection with the execution of any other Transaction Document; provided, that if any such Transfer Taxes are required to be collected, remitted or paid by a seller, 50% of such Transfer Taxes shall be paid by Calix to Ericsson at the Closing or thereafter, as required. All other Closing expenses will be paid by the Party incurring such expenses. Upon request, the Party taking

responsibility for the remittance of a jointly paid amount of Transfer Tax to a Tax Authority shall provide the other Party with an original receipt (or such other evidence as shall be reasonably satisfactory) evidencing the payment of such Transfer Taxes.

- (d) Each Transfer Tax Return with respect to Transfer Taxes imposed in respect of this Agreement and the transactions contemplated herein or in respect of the execution of any other Transaction Document shall be prepared by the Party that customarily has primary responsibility for filing such Transfer Tax Return pursuant to the applicable Tax Laws. Any Transfer Tax Returns prepared by pursuant to this Section shall be made available to the other Party at least five Business Days before such Tax Returns are due to be filed. The Party not filing such return shall pay to the filing Party 50% of the Transfer Taxes payable at least one Business Day before such Transfer Tax becomes due and payable.
- (e) Calix, with the reasonable assistance of Ericsson, shall be responsible for the preparation and timely filing (taking into account any extensions received from the relevant Tax Authorities) of all Tax Returns with respect to the Assets or the Business for all Straddle Periods.
- (f) <u>Audits</u>. Ericsson shall promptly notify Calix in writing upon receipt by Ericsson of notice of any pending or threatened federal, state, local or foreign Tax audits or assessments that reasonably may be expected to relate to the Purchased Assets.
- (g) <u>Characterization of Payments</u>. Any payments made under this <u>Section 5.7</u> or <u>Article VIII</u> shall constitute an adjustment to the Purchase Price for Tax purposes and shall be treated as such by the Parties on their Tax Returns to the extent permitted by law.
- 5.8 <u>Assets Held by Affiliates</u>. To the extent that any of the Purchased Assets are held or owned by Affiliates of Ericsson as of the Agreement Date, Ericsson shall cause each such Affiliate to transfer the Purchased Assets held by such Affiliate to Ericsson prior to the Closing. Notwithstanding the foregoing, the Parties recognize and agree that the Patent License Agreement and assignment of copyrights will be done by agreements with Affiliates of Ericsson.
- 5.9 <u>Further Actions</u>. From and after the Closing Date, Ericsson shall cooperate with Calix and Calix's Affiliates and Representatives, and shall execute and deliver such documents and take such other actions as Calix may reasonably request, for the purpose of evidencing the Transactions and putting Calix in possession and control of all of the Purchased Assets.
- 5.10 <u>Further Assurances</u>. Ericsson agrees to, and shall cause its respective Subsidiaries to, without further consideration, execute and deliver, following the Closing, such other instruments of transfer and take such other action as Calix or its counsel may reasonably request in order to put Calix in possession of, and to vest in Calix, good, valid and unencumbered title to the Purchased Assets under this Agreement and to consummate the contemplated acquisition.

5.11 Non-Compete; Non-Solicit.

(a) Ericsson agrees that neither Ericsson nor any Affiliate of Ericsson will at any time during the one- year period from and immediately following the Closing Date, solicit, seek to employ or otherwise attempt to induce any Transferred Employee to terminate his or her employment with Calix or any Affiliate of Calix, unless Calix gives its written consent to such employment or offer of employment. The restrictions on solicitation set forth in this Section 5.11(a) shall not apply to general bona fide public advertisements for employment placed by Ericsson and not specifically targeted at the employees of Calix or its Affiliates.

- (b) Ericsson acknowledges that the provisions of this Section 5.11 are reasonable and necessary to protect the interests of Calix, that any violation of this Section 5.11 may result in an irreparable injury to Calix and that damages at Law may not be reasonable or adequate compensation to Calix for violation of this Section 5.11 and that, in addition to any other available remedies, Calix shall be entitled to seek to have the provisions of this Section 5.11 specifically enforced by preliminary and permanent injunctive relief without the necessity of proving actual damages or posting a bond or other security to an equitable accounting of all earnings, profits and other benefits arising out of any violation of this Section 5.11. In the event that the provision of this Section 5.11 shall ever be deemed to exceed the time, geographic scope or other limitations permitted by Law, then the provisions shall be deemed reformed to the maximum extent permitted by Law.
- 5.12 <u>Transition Services</u>. For a period of 90 days following the Closing Date, Ericsson shall provide Calix with services to assist in the transition of the operation of the Purchased Assets, including the sale and marketing of the Business Products, by Calix after the Closing Date ("*Transition Services*"). Such Transition Services shall be provided on terms set forth in the attached Exhibit C, Transition Services.

ARTICLE VI: CONDITIONS TO THE TRANSACTIONS

- 6.1 <u>Conditions to Obligations of Each Party to Consummate the Transactions</u>. The respective obligations of each Party to consummate the Transactions shall be subject to the satisfaction at or prior to the Closing of each of the following conditions:
 - (a) Illegality. No temporary restraining order, preliminary or permanent injunction or other Order issued by any court of competent jurisdiction or other legal or regulatory restraint or prohibition preventing the consummation of the Transactions shall be in effect, nor shall any action have been taken by any Governmental Entity seeking any of the above, and no statute, rule, regulation or Order shall have been enacted, entered, enforced or deemed applicable to the Transactions, which makes the consummation of the Transactions illegal.
 - (b) <u>Governmental Approvals</u>. The Parties shall have timely obtained from each Governmental Entity all approvals, waivers and consents, if any, necessary for consummation of, or in connection with, the Transactions.
- 6.2 <u>Additional Conditions to Obligations of Ericsson</u>. The obligations of Ericsson to consummate the Transactions shall be subject to the satisfaction at or prior to the Closing of each of the following conditions (it being understood that each such condition is solely for the benefit of Ericsson and may be waived by Ericsson in writing in its sole discretion without notice or Liability to any Person):
 - (a) Representations, Warranties and Covenants. The representations and warranties of Calix in this Agreement in Section 3.1 and Section 3.2 and any representations and warranties of Calix in this Agreement that are qualified as to materiality shall have been true and correct in all respects on and as of the Agreement Date and shall be true and correct in all respects on and as of the Closing Date as though such representations and warranties were made on and as of such date, and all other representations and warranties of Calix in this Agreement shall have been true and correct in all material respects on and as of the Agreement Date and shall be true and correct in all material respects on and as of the Closing Date as though such representations and warranties were made on and as of such date. Calix shall have performed and complied in all material respects with all covenants, obligations and conditions of this Agreement required to be performed and complied with by it at or prior to the Closing.
 - (b) Receipt of Closing Deliveries. Ericsson shall have received each of the agreements, instruments

and other deliveries set forth in Section 1.4(a).

- 6.3 Additional Conditions to the Obligations of Calix. The obligations of Calix to consummate the Transactions shall be subject to the satisfaction at or prior to the Closing of each of the following conditions (it being understood that each such condition is solely for the benefit of Calix and may be waived by Calix in writing in its sole discretion without notice or Liability to any Person):
 - (a) Representations, Warranties and Covenants. The representations and warranties of Ericsson in this Agreement in Section 2.1, Section 2.2 and Section 2.3 and any representations and warranties of Ericsson in this Agreement that are qualified as to materiality shall have been true and correct in all respects on and as of the Agreement Date and shall be true and correct in all respects on and as of the Closing Date as though such representations and warranties were made on and as of such date, and all other representations and warranties of Ericsson in this Agreement shall have been true and correct in all material respects on and as of the Closing Date as though such representations and warranties were made on and as of such date. Ericsson shall have performed and complied in all material respects with all covenants, obligations and conditions of this Agreement required to be performed and complied with by it at or prior to the Closing.
 - (b) Receipt of Closing Deliveries. Calix shall have received each of the agreements, instruments and other deliveries set forth in Section 1.4(b).
 - (c) <u>Injunctions or Restraints on Conduct of Business</u>. No temporary restraining order, preliminary or permanent injunction or other order issued by any court of competent jurisdiction or other legal or regulatory restraint provision limiting or restricting Calix's ownership, conduct or operation of the Purchased Assets or the Business, following the Closing shall be in effect. Nor shall there be pending or threatened any Proceeding seeking (i) any of the above, any restraint relating to antitrust or any other injunction or restraint, or (ii) material damages in connection with the Transactions.
 - (d) No Ericsson Material Adverse Effect. There shall not have occurred and still be occurring an Ericsson Material Adverse Effect.
- 6.4 Employees. (i) Ericsson shall not have terminated the employment (unless for cause) of any of the employees identified on Schedule 5.6.
- 6.5 <u>Proceedings</u>. Since the Agreement Date, there shall not have been commenced or threatened against Calix, or against any Person Affiliated with Calix, any Proceeding (a) involving any material challenge to, or seeking material damages or other material relief in connection with, any of the Transactions, or (b) that may have the effect of preventing, delaying, making illegal or otherwise interfering with any of the Transactions.
- 6.6 <u>Release of Liens</u>. Calix shall have received evidence satisfactory to it of the release by any Person who held any Encumbrance on any of the Purchased Assets of all such Encumbrances on the Purchased Assets effective upon or concurrent with the Closing.

ARTICLE VII: TERMINATION

- 7.1 <u>Termination</u>. At any time prior to the Closing, this Agreement may be terminated and the Transactions abandoned by authorized action taken by the terminating Party:
 - (a) by mutual written consent of the Parties;

- (b) by either Party, if the Closing shall not have occurred on or before the date six months after the Agreement Date or such other date that Calix and Ericsson may agree upon in writing; <u>provided</u>, that the right to terminate this Agreement under this <u>Section 7.1(b)</u> shall not be available to any Party whose failure to fulfill any obligation under this Agreement has been the cause of, or resulted in, the failure of the Closing to occur on or before such date:
- (c) by either Party, if any permanent injunction or other Order of a Governmental Entity of competent authority preventing the consummation of the Transactions shall have become final and non-appealable;
- (d) by Calix, if (i) Ericsson shall have breached any representation, warranty, covenant or agreement and such breach shall not have been cured within ten Business Days after receipt by Ericsson of written notice of such breach (<u>provided</u>, <u>however</u>, that no such cure period shall be available or applicable to any such breach which by its nature cannot be cured) and if not cured within the timeframe above and at or prior to the Closing, such breach would result in the failure of any of the conditions set forth in <u>Section 6.1</u> or <u>Section 6.3</u> to be satisfied, or (ii) there shall have been an Ericsson Material Adverse Effect; or
- (e) by Ericsson, if Calix shall have breached any representation, warranty, covenant or agreement and such breach shall not have been cured within ten Business Days after receipt by Calix of written notice of such breach (provided, however, that no such cure period shall be available or applicable to any such breach which by its nature cannot be cured) and if not cured within the timeframe above and at or prior to the Closing, such breach would result in the failure of any of the conditions set forth in Section 6.1 or Section 6.2 to be satisfied.
- 7.2 Effect of Termination. In the event of termination of this Agreement as provided in Section 7.1, this Agreement shall immediately become void and there shall be no liability or obligation on the part of either Party or its respective officers, directors, stockholders or Affiliates; provided, however, that (a) the provisions of this Section 7.2 (Effect of Termination), Article IX (General Provisions), Section 5.1 (Confidentiality; Public Disclosure) and the Confidentiality Agreement shall remain in full effect and survive any termination of this Agreement and (b) nothing shall relieve any Party from liability in connection with any breach of such Party's representations, warranties or covenants prior to the date of such termination.
- 7.3 Extension; Waiver . At any time at or prior to the Closing, any Party may, to the extent legally allowed, (a) extend the time for the performance of any of the obligations or other acts of the other Parties, (b) waive any inaccuracies in the representations and warranties made to such Party or in any document delivered, and (c) waive compliance with any of the agreements or conditions for the benefit of such Party. At any time after the Closing, Ericsson on the one hand, and Calix on the other hand, may, to the extent legally allowed, (i) extend the time for the performance of any of the obligations or other acts of the other, (ii) waive any inaccuracies in the representations and warranties made to such Party, and (iii) waive compliance with any of the agreements or conditions for the benefit of such Person. Any agreement on the part of a Party to any such extension or waiver shall be valid only if set forth in an instrument in writing signed on behalf of such Party. Without limiting the generality or effect of the preceding sentence, no delay in exercising any right under this Agreement shall constitute a waiver of such right, and no waiver of any breach or default shall be deemed a waiver of any other breach or default of the same or any other provision in this Agreement.
- 7.4 Non-exclusivity of Termination Rights. The termination rights provided in Section 7.1 shall not be deemed to be exclusive. Accordingly, the exercise by any Party of its right to terminate this Agreement under Section 7.1 shall not be deemed to be an election of remedies and shall not be deemed to prejudice,

or to constitute or operate as a waiver of, any other right or remedy that such Party may be entitled to exercise (whether under this Agreement, under any other Contract, under any statute, rule or other Legal Requirement, at common law, in equity or otherwise).

ARTICLE VIII: INDEMNIFICATION

8.1 <u>Survival of Representations and Warranties</u>. Each Party shall have the right to rely fully upon the representations, warranties, covenants and agreements of the other Parties contained in this Agreement, the Disclosure Schedule (including any exhibit or schedule of the Disclosure Schedule), and the other agreements and certificates contemplated without regard to investigation or knowledge. If the Transactions are consummated, all of the representations and warranties of Ericsson contained in this Agreement, the Disclosure Schedule, and the other agreements and certificates shall survive the Closing and remain in effect, regardless of any investigation or disclosure made by or on behalf of any of the Parties, until two years following the Closing; provided, however, that such expiration shall not affect the rights of any Indemnified Person under <u>Article VIII</u> or otherwise to seek recovery of Losses arising out of any fraud, intentional misrepresentation or intentional omission of Ericsson, its Affiliates or their respective Representatives until the expiration of the applicable statute of limitations. All Disclosures made by or on behalf of any of the Parties, shall be deemed to be within the Knowledge of the receiving Party. If the Transactions are consummated, the representations and warranties of Calix contained in this Agreement and the other certificates shall expire as of the Closing. If the Transactions are consummated, all covenants of the Parties (including the covenants set forth in <u>Article IV</u> and <u>Article V</u>) shall expire as of the Closing, except to the extent such covenants provide that they are to be performed after the Closing; provided, however, that no right to indemnification under <u>Article VIII</u> in respect of any claim based upon any breach of a covenant shall be affected by the expiration of such covenant.

8.2 Indemnification.

- (a) Subject to the limitations set forth in this Article VIII, Ericsson shall indemnify and hold harmless the Indemnified Persons from and against, and shall compensate and reimburse the Indemnified Persons for all Losses arising out of, resulting from, related to or in connection with (i) any failure of any certification, representation or warranty made by Ericsson in this Agreement, or (ii) any Excluded Liability or (iii) any breach of any covenant in this Agreement, but only to the extent such covenant provides that it is to be performed after the Closing.
- (b) The above obligation of Ericsson to indemnify the Indemnified Persons shall be subject to each of the following principles or qualifications:
 - (i) No claim for the recovery of Losses under Section 8.2(a)(i) may be asserted by any Indemnified Person against Ericsson or its successors in interest after such representations and warranties have expired under Section 8.1; provided, however, that claims first asserted in writing prior to such expiration shall not be barred.
 - (ii) The maximum aggregate amount of indemnifiable Losses that may be recovered from Ericsson for claims made under <u>Section 8.2(a)(i)</u> shall not exceed the Purchase Price. Claims arising out of, resulting from, related to or in connection with fraud, intentional misrepresentation or intentional omission of Ericsson, its Affiliates or any of their respective Representatives, and all Losses related to Excluded Liabilities, shall not be subject to this <u>Section 8.2(b)(ii)</u> or any other contractual limitation.
 - (iii) The Liability of Ericsson for indemnifiable Losses under Section 8.2(a)(i) shall not be payable unless and until the aggregate amount of Losses suffered or incurred by the Indemnified Persons exceeds [***], following which, the Indemnified Persons shall be entitled to seek compensation for all indemnifiable Losses, and Ericsson shall be

responsible for the payment of indemnifiable Losses, without regard to such limitation; provided, that the limitation set forth in this Section 8.2(b)(iii) shall not apply to any Losses due to Claims arising out of, resulting from, related to or in connection with fraud, intentional misrepresentation or intentional omission of Ericsson, its Affiliates or any of their respective Representatives.

- (iv) The right to indemnification based on any representations or warranties will not be affected by (A) any investigation conducted with respect to, or any knowledge acquired (or capable of being acquired) at any time, whether before or after the Closing Date, with respect to the accuracy or inaccuracy of or compliance with, any such representation or warranty or (B) the breach of, or knowledge of the inaccuracy of any, such representations or warranties by Ericsson.
- (v) Ericsson (A) waives any right to require any Indemnified Person to (1) proceed against any Person including any other Indemnified Person, (2) proceed against or exhaust any collateral or security or (3) pursue any other remedy in its power, and (B) waives any defense arising by reason of any inability of any other obligor to pay or any defense based on bankruptcy or insolvency or other similar limitations on creditors' remedies with respect to any other Person.
- (vi) This Section 8.2 states Ericsson's, including its Affiliates or any of their respective Representatives, sole liability for claims arising out of this Agreement, the Transactions, and related agreements.
- 8.3 <u>Indemnification Procedure</u>. In the event Calix becomes aware of a third-party claim which Calix believes may result in a claim for indemnification under this <u>Article VIII</u> by or on behalf of an Indemnified Person, Calix shall promptly notify Ericsson of such third-party claim. No delay in providing such notice shall affect an Indemnified Person's rights, unless (and then only to the extent that) Ericsson is materially prejudiced. Calix shall have the right in its sole discretion to conduct the defense of and to settle or resolve any such claim (and the costs and expenses incurred by Calix in connection with such defense, settlement or resolution (including reasonable attorneys' fees, other professionals' and experts' fees and court or arbitration costs) shall be included in the Losses for which Calix may seek indemnification via a claim). Ericsson shall have the right to receive copies of all pleadings, notices and communications with respect to the third-party claim to the extent that receipt of such documents does not affect any privilege relating to any Indemnified Person and shall be entitled, at its expense, to participate in, but not to determine or conduct, any defense of the third-party claim or settlement negotiations with respect to the third-party claim. However, except with the consent of Ericsson, which consent shall not be unreasonably withheld, conditioned or delayed and which shall be deemed to have been given unless Ericsson shall have objected within 15 days after a written request for such consent by Calix, no settlement or resolution of any such claim with any third-party claimant shall be determinative of the existence of or amount of Losses relating to such matter. In the event that Ericsson has consented to any such settlement or resolution, Ericsson shall not have any power or authority to object under <u>Section 8.3</u> or any other provision of this <u>Article VIII</u> to the amount of any claim by or on behalf of any Indemnified Person for indemnity with respect to such settlement or reso

ARTICLE IX: GENERAL PROVISIONS

- 9.1 <u>Fees and Expenses</u>. Whether or not the Transactions are consummated, all costs and expenses incurred in connection with the Transaction Documents and the Transactions shall be paid by the Party incurring such expense, and all Transaction Expenses shall be borne by Ericsson.
- 9.2 Bulk Sales Compliance. Calix hereby waives compliance by Ericsson with any applicable bulk sale

or bulk transfer laws of any jurisdiction in connection with Ericsson's sale of the Assets hereunder.

- 9.3 Attorneys' Fees. If any legal action or other legal proceeding relating to any of the Transaction Documents or the enforcement of any provision of any of the Transaction Documents is brought against any Party, the prevailing Party shall be entitled to recover its reasonable attorneys' fees, costs and disbursements (in addition to any other relief to which the prevailing Party may be entitled).
- 9.4 <u>Governing Law; Venue</u>. The Transaction Documents shall be construed under the laws of the State of New York, US, without regard to its conflicts of laws provisions. The Parties consent to the exclusive jurisdiction of, and venue in, the state and federal courts within the state of New York for the resolution of any disputes arising out of or related to the Transaction Documents.
- 9.5 <u>Notices</u>. Any notice or other communication required or permitted to be delivered to either Party under the Transaction Documents shall be in writing and shall be deemed given upon receipt when delivered (by hand, mail, courier or email with confirmed receipt) to the addresses set forth above (or to such other address as such Party shall have specified in a written notice given to the other Party), and directed to the following individuals:

if to Calix, to: Carl Russo

President and CEO

Email: [***]

with a copy to: Denis J. Quinlan

General Counsel

Email: [***]

if to Ericsson, to: John Moore

General Counsel

Email: [***]

with a copy to: Kyle McElroy

Associate General Counsel

Email: [***]

9.6 Successors and Assigns; Parties In Interest .

- (a) This Agreement shall be binding upon and inure to the benefit of the Parties and their respective successors and assigns (by merger, consolidation, acquisition of stock or assets or otherwise).
- (b) Except as set forth below, neither Party may assign its rights under this Agreement (including indemnification rights under Article VIII), in whole or in part, to any other Person without first obtaining the written consent or approval of the other Party. After the Closing Date, Calix may transfer and assign this Agreement and the rights granted under it upon the sale of all or substantially all of Calix's assets. Provided, however, that this Section 9.6(b) shall not apply to the assignment of Calix's rights under the Patent License Agreement between Calix and Telefonaktiebolaget LM Ericsson (as set forth in Exhibit C). Calix agrees to use commercially reasonable efforts to notify Ericsson not less than 30 days prior

to the sale of all or substantially all of its assets. Prior to the Closing Date, upon the sale of all or substantially all of Calix's assets, Calix may assign this Agreement with the prior written consent of Ericsson, which shall not be unreasonably withheld.

- (c) Except for the provisions of Article VIII, none of the provisions of this Agreement is intended to provide any rights or remedies to any Person other than the Parties and their respective successors and assigns. However, (i) no employee of Ericsson shall have any rights under this Agreement or under any of the other Transaction Documents, and (ii) no creditor of Ericsson shall have any rights under this Agreement or any of the other Transaction Documents.
- 9.7 <u>Remedies Cumulative</u>; <u>Specific Performance</u>. Ericsson agrees that: (a) in the event of any breach or threatened breach by Ericsson of any covenant, obligation or other provision set forth in this Agreement, Calix shall be entitled (in addition to any other remedy that may be available to it) to seek (i) a decree or order of specific performance or mandamus to enforce the observance and performance of such covenant, obligation or other provision, and (ii) an injunction restraining such breach or threatened breach; and (b) neither Calix nor any other Indemnified Person shall be required to provide any bond or other security in connection with any such decree, order or injunction or in connection with any related action or Proceeding.
- 9.8 Severability. In the event any one or more of the provisions of the Transaction Documents is held to be unenforceable under any law, such unenforceability shall not affect any other provision of the Transaction Documents. The Transaction Documents shall be construed as if the unenforceable provision(s) had not been contained in it, and the Parties shall negotiate in good faith to replace the unenforceable provisions by such modified provisions as have the effect nearest to that of the provisions being replaced.
- 9.9 Entire Agreement . The Transaction Documents and the Confidentiality Agreement constitutes the entire understanding between the Parties concerning its subject matter and supersedes all prior discussions, agreements, understandings, and representations, including the conflicting terms of any purchase order or other document, whether oral or written and whether or not executed by Calix or Ericsson. No modification, amendment or other change may be made to this Agreement unless in writing and executed by authorized representatives of both Parties.
- 9.10 English Language. The Transaction Documents are in the English language only, which language will be controlling in all respects, and any versions in any other language will not be binding on the Parties. All communications and notices to be made or given under the Transaction Documents will be in English.
- 9.11 Execution. The Transaction Documents may be executed in two or more counterparts, each of which when executed shall be deemed to be an original but all of which together shall constitute one agreement. The Parties agree that signatures delivered by scan, email or other electronic means shall be considered original signatures for all purposes under the Transaction Documents.

The Parties have caused this Agreement to be executed and delivered as of the Agreement Date.

By: Printed Name: Title: Ericsson, Inc. By: Printed Name: Title:	Calix, Inc.	
Ericsson, Inc. By: Printed Name: Title:		
By: Printed Name: Title:	Title:	
By: Printed Name: Title:		
Printed Name: Title:	Ericsson, Inc.	
Title:	Ву:	
	Printed Name:	
	Title:	
36	36	

Exhibit A

BILL OF SALE

Delaware Corpora		12, by and between Ericsson Inc. A Delaware Corporation, ("Seller"), and Calix Inc. A a, CA 94954 ("Buyer"). Additional capitalized terms are defined in the Asset Purchas reement").
of Sale, [***] of I		ourchase of the following assets: [***] of capital assets listed as Appendix A of this Bil onal [***] of Inventory to be transferred on or before January 15, 2013, and all Ericsson
inventories will co		y will transfer on the Closing Date, and another [***] on or before January 15, 2013 til the Closing Date. Therefore upon the Closing Date Seller will provide Buyer a list of
	REAS, the Seller and Buyer have agreed that the price yould be [***] and that the Equipment would be made	for the Purchased Assets transferred under the Agreement (including without limitation available for Buyer on the Closing Date.
NOW,	THEREFORE, for good and valuable consideration,	the parties hereto agree as follows:
1.	Seller hereby sells, transfers, and conveys to Buye payment of [***], payable to Ericsson, Inc. in cert	er all of Seller's rights, title, and interest in and to the Equipment in exchange for the tified funds.
2.	disclaims all warranties (express, implied or statut merchantability, (b) the implied warranty of fitnes	IS AND WITH ALL FAULTS. Except as set forth in the Agreement, Seller expressly tory) of any type including but not limited to (a) the implied warranty of ss for a particular purpose and (c) any warranty regarding the use of the Equipment, its manship, the conformity of the Equipment to any law, rule, regulation, specification or ent infringement.
("Seller")		("Buyer")
By:		By:
Name:		Name:
Title:		Title:
	nformation in this document has been omitted and file atment has been requested with respect to the omitted	od separately with the Securities and Exchange Commission.

Appendix A Capital Assets

[***]		
[***]	Inventory Summary as of August 17, 2012	
[***] Certain information in this docu Confidential treatment has been reque	ument has been omitted and filed separately with the Securities and Exchange Commission. ested with respect to the omitted portions.	

Exhibit B

Reseller Agreement

This Reseller Agreement ("Reseller Agreement") is entered into on August __2012 ("Effective Date") between Ericsson AB ("Ericsson"), a Swedish corporation with offices at Torshamngsgatan 23, 164 80 Stockholm, Sweden and Calix, Inc. ("Calix"), a Delaware corporation with offices at 1035 N. McDowell Blvd., Petaluma, CA 94954. Ericsson and Calix may be referred to individually as a "Party" or collectively as the "Parties."

Calix appoints Ericsson as a Reseller of Calix products on the following terms and conditions:

1. Terms of Appointment.

- 1.1 <u>Appointment</u>. Subject to the terms and conditions of this Reseller Agreement, Calix appoints Ericsson, as a non-exclusive reseller of those Calix products listed in Exhibit A ("Products"), and as a reseller of Calix Services and Support ("Services") as listed in the Support and Service Addendum, attached as Exhibit B. Ericsson agrees to purchase the Products and Services exclusively from Calix, and shall not serve as a reseller of new or used Products obtained from other sources.
- Preferred Global Partner. During the Term (as defined below) Ericsson will be the prime contractor for the resale of Products and Services to those customers of Ericsson listed in Exhibit C ("Customers"), except as set forth below. It is Ericsson's intention to sell the Calix Products as Ericsson's preferred solution for broadband access applications and Ericsson Business Networks shall position Calix as its preferred global partner. Calix and Ericsson wish to work cooperatively to develop the broadband access market for application solutions such as GPON across Ericsson and Customer's networks. Accordingly, Ericsson agrees to introduce Calix to all Ericsson Customers listed in Exhibit C. Calix will have the responsibility to coordinate with Ericsson for introductions to such Customers. Upon reviewing each Customer, and with Ericsson's consent (which will not be unreasonably withheld), Calix may offer and conclude completely separate agreements directly with Customers at Calix's discretion.
- 1.3 Marketing and Sales Program . With Calix's input and mutual approval of the Parties, Ericsson shall prepare a marketing and sales program to promote the Calix Products widely among Ericsson's and its affiliates' marketing organization and sales force. This program shall include but is not limited to marketing launch, marketing materials, business solution materials, support documentation, sales launch program, road shows and training sessions. Ericsson shall launch this program as part of a global access solution and Calix will provide the required material and resources to support Ericsson in this effort. Ericsson will also place these materials in the Ericsson Global product Catalog for access by Ericsson regions and Customer units. Both Parties shall support the above programs with product management, project management, program management, and technical support, and Ericsson shall maintain a support organization to fulfill these requirements. The Parties agree to evaluate the progress of this Section 1.3 through Quarterly Business Reviews ("QBRs").
- 1.4 <u>Amendment of Exhibit C</u>. Exhibit C may be amended from time to time by via the following process: Ericsson shall notify Calix in writing each time it wishes to add a "Customer" to Exhibit C. Within five business days of receipt of such notice, Calix must accept or decline the addition. If Calix fails to decline such designation within five business days of receipt of notice from Ericsson, the notice shall be deemed accepted and such Customer shall be added to the list in Exhibit C.
- 1.5 <u>Direct Sales</u>. Calix shall have the right to appoint other resellers of the Products and Services. Calix shall at all times retain the right to sell, market and service the Products and Services directly to any customer (including those listed in Exhibit C). In the event Ericsson and Calix do not agree on a direct Calix customer

relationship as permitted in this Section 1.2, except as set forth in Section 1.6 below, if Calix intends to sell directly to a Customer listed in Exhibit C, Calix may proceed to sell to such Customer after providing 30 days notice to Ericsson.

- 1.6 Focus on Calix Products and Services. The intent of the Parties is to develop a market for, promote, and sell the Calix Products and Services. Ericsson BNET will work in good faith, both for new and existing Customers to position Calix as the vendor of choice for all broadband access opportunities. If to the knowledge of director level and above employees of the Ericsson PAID organization during the Term of this Reseller Agreement, and upon the request of an Ericsson Customer, an Ericsson Region, affiliate or distributor intends to market or offer an alternative solution of non-Calix products or services that may reasonably be considered competitive with the Products or Services, Ericsson BNET will work with Calix on a competitive offering to meet the Customer requirement. Ericsson will notify Calix at least 30 days in advance of offering such competitive products or services and offer Calix the immediate ability to sell directly to that customer.
- 1.7 <u>Resale Restrictions</u>. Except with Calix's prior consent, Ericsson shall not resell Products to any other reseller, to any other similar resale vendor or channel partner, or to any Calix competitor, including but not limited to Adtran, Alcatel-Lucent, Enablence, Huawei, Tellabs, Zhone, and ZTE.
- 1.8 <u>Customer Service</u>. Ericsson agrees to provide its Customers with technical service in accordance to the standards and terms set forth in Exhibit B.
- 1.9 Forecasts and Reports. Ericsson shall furnish Calix with:
 - 1.9.1 within 15 business days of the end of each quarter, a statement of shipments of Products, with detail by Customer; including, Customer name, ship-to-address, part number, quantity, and ship date.
 - 1.9.2 a quarterly, non-binding, rolling 12-month forecast which sets forth projected demand for the Products and Services for each of the succeeding 12 months;

2. Term and Termination.

- 2.1. <u>Term.</u> This Reseller Agreement is effective as of the Effective Date, and shall continue in force, unless terminated under Section 2.2, below, for an initial period of three years ("Term"). The Term will then automatically renew for a succession of one-year periods unless either Party gives 60 days' written notice of termination prior to the end of any such one-year extension of the Term.
- 2.2. <u>Termination</u>. This Reseller Agreement may also be terminated:
 - 2.2.1 By Calix, effective immediately upon Ericsson's receipt of Calix's written notice of termination, upon the occurrence of any of the following events:
 - 2.2.1.1. a breach of Section 11 of this Reseller Agreement;
 - 2.2.1.2. Ericsson becoming the subject of any proceeding under bankruptcy, receivership, insolvency or similar laws instituted by or against Ericsson, which proceedings are not dismissed as to Ericsson within 30 days after being instituted;
 - 2.2.1.3. Ericsson engaging in conduct which, in the sole judgment of Calix, adversely affects Calix's reputation or its interests in the promotion, marketing or distribution of Products or Services; or
 - 2.2.1.4. any change of control of Ericsson; or
 - 2.2.2 By either Party, if the other fails to cure a material breach of this Reseller Agreement, other than of

- Section 11, within 30 days after receipt of notice of such breach.
- 2.2.3 By Ericsson, effective immediately upon Calix's receipt of Ericsson's written notice of termination, upon the occurrence of any of the following events:
 - 2.2.3.1 A change of control event in which Calix is acquired by or merged into any entity which may reasonably be considered to be a direct competitor of Ericsson
 - 2.2.3.2 Calix becoming the subject of any proceeding under bankruptcy, receivership, insolvency or similar laws instituted by or against Calix, which proceedings are not dismissed as to Calix within 30 days after being instituted; or
 - 2.2.3.3 Calix engaging in conduct which, in the sole judgment of Ericsson, adversely affects Ericsson's reputation or its interests in the promotion, marketing or distribution of Products or Services;
- 2.3. <u>Obligations Upon Termination</u>. Upon the expiration or termination of this Reseller Agreement for any reason ("End Date") Ericsson shall, within 90days of the End Date:
 - 2.3.1 Make all payments due to Calix incurred prior to the End Date;
 - 2.3.2 Return to Calix any and all Calix Confidential Information, sales promotion materials such as catalogues, price lists, technical manuals, signs, displays, etc., and Ericsson shall discontinue any and all use of Calix's name, and shall cease in any manner to represent itself to be a Reseller of the Products and Services;
 - 2.3.3 Provide Calix with complete information regarding any outstanding commitments to supply Products or Services to Customers that had not yet been reduced to orders, and assist Calix in good faith to fulfill any such commitments; and
 - 2.3.4 Cooperate with Calix to transfer all existing warranty obligations and agreements for technical support services to Calix.
- 2.4. Ongoing Rights and Obligations. From and after the End Date:
 - 2.4.1 End user licenses to software contained in Products will survive termination or expiration of this Reseller Agreement in accordance with their terms;
 - 2.4.2 Ericsson may resell Products and Services for which orders have been accepted prior to the End Date; and
 - 2.4.3 Each Party shall remain obligated to provide any indemnification due to the other Party under this Reseller Agreement, and each Party will retain all rights and entitlement to remedies with respect to any prior breach of this Reseller Agreement;
- 2.5. The rights and remedies provided for in this Section are in addition to any other remedy provided by law, and do not release either Party from any liability for damages for which it might otherwise be responsible. Rights and obligations that by their nature would continue beyond the termination of this Reseller Agreement shall survive the termination of this Reseller Agreement.
- 2.6. The termination of this Reseller Agreement for any reason whatsoever shall be without prejudice to any right or obligation of any Party in respect of this Reseller Agreement, which has arisen prior to such termination, including but not limited to that all sublicenses granted to end users (or required to be granted under binding quotes, contracts or purchase orders) before termination shall remain in full force and effect and end users' continued use of the Software shall in no way be affected. Further, Ericsson shall be entitled to continue to use the software and the support materials for the sole purpose of providing support to such end users.
- 3. Pricing and Sale.

- 3.1. Prices. Calix shall sell Products and Services to Ericsson at the prices in Exhibit A, as that Exhibit is amended from time to time by good faith agreement of the Parties ("Prices"). During the initial year of the Term, the Prices shall be adjusted by agreement of the Parties on a quarterly basis (at the end of each 90-day period following the Effective Date. Thereafter during the Term, the Prices shall be adjusted once annually, on or after each anniversary of the Effective Date. Prices are Calix Confidential Information, and shall not be disclosed to any third party, shall not be advertised, and shall not be posted to any website other than Ericsson or Customer intranet sites not available to the general public, or made available via any other publicly available resource.
- 3.2. <u>Taxes</u>. Calix shall bear at its own cost any taxes, duties, levies, social charges, and other fiscal charges imposed on it or its employees, agents and subcontractors and keep Ericsson indemnified from any such duties, taxes, levies, and other fiscal charges if imposed on Ericsson in connection with the delivery of any Product, or the execution of the Services. Upon request by Ericsson, Calix shall show evidence that the applicable taxes have been paid. Value added, excise, use and sales tax, and any other taxes assessed as a result of the sale of Products and Services (apart from taxes assessed on the revenues of Calix) shall be the responsibility of Ericsson. If in accordance with present or future laws, any taxes are required to be withheld on payments made by Ericsson, Ericsson shall deduct such taxes from the amount owed to Calix and pay the taxes to the appropriate tax authority. Promptly after each such payment, the official tax receipt issued by the tax authority will be forwarded by Ericsson to Calix to enable it to support a claim for tax credit.
- 3.3. Resale Prices. Ericsson will determine its own resale prices to Customers. Calix may, however, from time to time provide Ericsson with suggested retail price lists. Resale prices shall not be shared with any third party other than Customers, and shall not be advertised publicly, posted to any website other than Ericsson or Customer intranet sites not available to the general public, or made available via any other publicly available resource. However, Ericsson may disclose pricing to Customers via a secure, password-protected website or online system controlled by Ericsson and made available solely to Customers.
- 3.4. <u>Invoicing/Payment</u>. Each invoice shall refer to one purchase order only and shall be submitted to Ericsson's location designated in the relevant purchase order. Invoices shall refer to the purchase order number and Ericsson's product or service numbers for the Services. All invoices are payable [***]. All prices are set forth in, and payments must be made in, U.S. dollars. Late payments will accrue interest at the lesser of 1.5% per month and the highest rate allowed by applicable law.
 - 3.4.1 <u>Products.</u> Calix shall invoice Ericsson for Products and installation materials associated with Services upon shipment as per article 4.
 - 3.4.2 <u>Services</u>. Services shall be invoiced quarterly in arrears. Installation Services may be invoiced only after acceptance of such Services by Ericsson and the applicable Customer on a per-site basis as defined within the applicable statement of work, and training Services may be invoiced only after performance of the specific training Services. Provided, however, that if Ericsson or Customer fail to accept or reject any Services within ten business days of receipt of notice from Calix that such Services have been completed, such Services shall be deemed accepted.
 - 3.4.3 Products and Services acquired by Ericsson under any contract shall be regarded as part of Ericsson Company volumes in relation to staggered prices and other volume calculations.
 - 3.4.4 Upon Calix's consent, Ericsson may deduct from any amounts payable to Calix any liquidated damages or other damages, costs and expenses owed to Ericsson as a result of this Agreement.

3.5. Ordering Process.

- 3.5.1. Ericsson (or one of its affiliates or authorized distributors (each an "Ericsson Company")) shall send Calix orders for Products and Services (each an "Order"), which shall include: A) Order ID numbers in an agreed format; B) All necessary billing and shipping instructions; C) Product part numbers, descriptions, quantities and prices; and D) Complete Customer name and contact information. Orders may be sent to Calix by EDI, email, courier or mail. Calix may, at its discretion, accept Orders by phone, subject to confirmation via one of the above methods. Orders shall be signed or electronically authenticated by an authorized representative of Ericsson.
- 3.5.2. Within two business days of receipt of an Order, Calix shall send an "Order Acknowledgment" accepting or rejecting Ericsson's order and confirming shipment date ("Shipment Date"). If Calix rejects an Order because it contains errors or omissions, Calix shall notify Ericsson of the particulars of the discrepancy within an additional three business days of sending the rejecting Order Acknowledgment, and request that Ericsson send Calix a corrected Order prior to acknowledgment.
- 3.5.3. At the request of an Ericsson Company, Calix shall provide a quotation for Products and Services ("Quotation"). The Quotation shall be binding on Calix for a period of no less than 180 days. Ericsson Companies may issue Orders to the Calix by referring to the Quotation.

3.6. Cancellation and Return.

- 3.6.1. Ericsson may cancel an Order by giving written notice at least two weeks prior to Shipment Date.
- 3.6.2. Any unopened Products may only be returned to Calix [***].
- 3.7. Rescheduling. Ericsson may, without charge, reschedule the shipment of any Order by giving written notice to Calix no less than two weeks prior to the Shipment Date. The requested rescheduled shipment date shall be no more than 120 days after the original Shipment Date. In the event that Ericsson (a) gives notice of rescheduling less than two weeks prior to the Shipment Date, (b) requests a new shipment date greater than 120 days after the Shipment Date, or (c) attempts to request a second rescheduling of any Order, such Order shall be considered canceled and the terms of Section 3.6.1 shall apply.

3.8. Delivery Delay.

- 3.8.1. Calix will endeavor to provide Ericsson with 15 days notice in advance of any anticipated delay in the Shipment Date of a Product Order. In the event of any such delay, Calix shall use its best commercially reasonable efforts to:
 - 3.8.1.1. Upon mutual consent, make a partial shipment of the Products under such Order;
 - 3.8.1.2. Ship to other generally available Products providing substantially similar form, fit and function to the ordered Products; or
 - 3.8.1.3. Take such other reasonable actions as may be agreeable to Ericsson and Customer.

- 3.8.2. In the event that a Product delivery delay is greater than 30 days from the Shipment Date, and Calix is unable to implement solutions under Section 3.8.1 that are reasonably acceptable to Ericsson and the Customer, Ericsson may cancel any undelivered portion of such Order without cost to Ericsson. Further, if the reason for delay was solely due to Calix performance, Calix will reimburse Ericsson for any penalties charged by end customers that are specific to aforementioned delay, provided that Ericsson has given Calix notice of Customer agreements that have such penalty terms in advance of the placement of any Order to which such penalties may apply.
- 3.9. Product Changes. Calix may make modifications in Specifications (as defined in Section 0, below), construction, or design of any Product at any time, and any Product so modified shall be accepted by Ericsson in fulfillment of existing Orders, provided that such Product, as modified, meets or exceeds the original Specifications. Calix shall endeavor to provide immediate notice to Ericsson of any such changes that affect Product form, fit and function.
- 3.10. <u>Obligation to Pay</u>. Ericsson bears sole responsibility to pay for accepted Orders of Products and Services regardless of any non-payments by any of its Customers.
- 4. <u>Shipping, Title and Risk of Loss</u>. All Orders shall be [***] Freight carrier shall be determined and shipping arrangements made by Calix based on Customer location and requirements, and title and risk of loss will pass to Ericsson when Calix places the Products at the disposal of the carrier.
- 4.2 In the event Calix's ability to ship accepted Orders is constrained due to an inability to procure sufficient raw materials necessary to satisfy all of its accepted customer orders, Calix agrees to, at a minimum, fulfill Ericsson's orders in the proportion equal to Ericsson's share of Calix's total customer orders for the same product.

5. Warranty.

- 5.1. Calix warrants that it has and will maintain sufficient rights and interests in licensed programs and Specifications (as defined below), including licensor's and any third party intellectual property rights (as applicable), to grant the rights and licenses under this Reseller Agreement with respect to the Products listed on Exhibit A as "Calix Product". Calix makes no such warranties with respect to Products listed on Exhibit A as "1500 Legacy" and "ONT Legacy" Products.
- 5.2. <u>Warranty Period</u>. Calix warrants that all Products shall operate in accordance with Calix's published specifications and documentation ("Specifications"), when used in accordance with those Specifications, for [***] from the date of shipment of Products ("Warranty Period"). Provided, however, that the Warranty Period shall <u>not</u> apply to any Ericsson ONT products listed in Exhibit A. The Warranty period may be extended only with purchase of extended warranty program coverage.

5.3. Warranty Limitations.

5.3.1. The above warranty extends only to new Products sold directly to Customers by Ericsson, and is not transferable to subsequent purchasers.

- 5.3.2. The above warranty shall not apply if the Product (i) has been subjected to abuse, misuse, accident, improper testing, improper storage, or other use contrary to Calix instructions, (ii) has been repaired or modified by persons other than Calix, (iii) has not been installed, operated, repaired and maintained in accordance with its documentation (iv) has failed due to an act of God, including without limitation fire, flood, tornado, earthquake, hurricane or lightning or (v) has been used with any third-party software or hardware not previously approved in writing by Calix. In addition, the above warranty shall not apply to Products (i) marked or identified as "sample," (ii) loaned or provided at no cost, (iii) which have had their serial numbers or other identifying marks removed or altered, or (iv) sold "as is".
- 5.3.3. In the event that Ericsson provides any additional warranties to its Customers beyond those set forth in this Section 5: (i) Ericsson shall advise Calix in writing of such additional warranties; and (ii) any additional warranties are made at Eriksson's risk and will not bind Calix.
- 5.4. Repair or Replacement. If during the Warranty Period (i) Calix is notified promptly in writing upon discovery of a breach of the above warranty, including a detailed description of the alleged breach, (ii) the Product is returned, shipping prepaid, to a designated manufacturing facility under Calix's then-current return procedures, and (iii) Calix determines that the Product is defective and not subject to any Section 5.3 exception, then, as Customer's sole remedy and Calix's sole obligation, Calix will, at its option, repair or replace the Product without charge. Any Product repaired or replaced under warranty shall have warranty coverage for the longer of [***] from return or the remaining Warranty Period. Replacement parts used in the repair of Products may be new or used. The replacement Products shall be delivered to Customer at Calix's expense and risk.
- 5.5. <u>Software</u>. With respect to software defects, Calix's sole obligation is to use commercially reasonable efforts to: provide corrected software, communicate a workaround, or provide another remedy which enables Customer to achieve substantially the same functionality as could be obtained without the defect. The software warranty applies solely to software within two major releases of the most current for that Product.
- 5.6. Specifications. Should Calix become aware of any error with regard to the content of the Specifications which is not (alone or accumulated) trivial or insignificant, Calix shall promptly inform Ericsson in writing about the error and provide Ericsson with new corrected Specifications. If Ericsson will suffer expenses as consequence of the error, the parties shall in good faith discuss to what extent Calix shall compensate Ericsson.

5.7. Systemic Defects.

- 5.7.1. By mutual consent, in addition to the warranties above, Calix shall be obligated to remedy any systemic defects in the Products. A systematic defect is a defect of the same or substantially the same type, that appears or is likely to appear during the reasonably expected life time of the concerned Products in more than 3% of such Products delivered during any relevant time period, such time period to be determined by Ericsson in its reasonable discretion, however not less than a time period of three consecutive months. In case of a systematic defect in the Products, Calix shall promptly:
 - 5.7.1.1. at no charge to Ericsson, replace all units of the respective Products delivered up to the time that the systematic defect has been remedied by Calix (<u>provided, however</u> that this Section 5.7.1.1 shall not apply to any EDA 1500 products previously shipped by Ericsson or shipped by Calix within the first nine months after the Effective Date); and

- 5.7.1.2. reimburse Ericsson for the actual costs for the replacement of such defective Products.
- 5.7.2. Calix's obligations regarding systemic defects shall be applicable for the full expected life time of the relevant Products, which time, however, never shall be shorter than [***] from the date of delivery of the relevant Products, or [***] in the case of ONT Products.
- 5.8. <u>Disclaimer</u>. Calix makes no additional warranty, express, implied or statutory, for any Product or Service, including without limitation any implied warranty of merchantability, fitness for a particular purpose, non-infringement, or arising from usage of trade, course of performance, or dealing.
- **6. Road Maps**. Under no circumstances shall provision to Ericsson of any product road maps, or other documents or oral presentations of any kind concerning Calix's future product or service development plans, be construed as a binding commitment to provide any such new products, services, features or enhancements to Ericsson or any Customer on any specific timetable or at all.
- 7. <u>Calix Advantage Program.</u> Annual purchase of Calix Advantage Program services by Customers is mandatory and is required in order to receive the following with regard to Calix Products listed in Exhibit A only [***]:
- 7.1. CMS. A license (under Section 8) to use Calix Management Software ("CMS");
- 7.2. <u>Upgrades</u>. All upgrades, updates and new releases of Calix Software ("Software"), including CMS, made generally available to other Software licensees; and
- 7.3. <u>Support</u>. Toll-free telephone access to the Technical Assistance Center ("TAC"), for resolution of any Product issues, in accordance with the terms of Exhibit B.

8. IPR and Software License.

- 8.1. Calix and its licensors own and retain all intellectual property rights (including any patent, copyright, trade secret and other intellectual property rights and interests collectively "IPR") embodied in the Products or Services. Calix at no time waives or assigns any rights or ownership to any IPR and Ericsson shall not contest the ownership of any IPR.
- 8.2. Calix grants to Ericsson and to its Customers, a personal, non-transferable and non-exclusive right to use, in object code form, all software and related documentation constituting or included in any Product or Service sold under this Reseller Agreement. Title and ownership of all software shall remain with Calix or its suppliers. Ericsson and its Customers may use the software only with Products and Services and only as specified in the Specifications. Calix may request specific site and workstation information for registration. Ericsson is granted no license to transfer, sublicense, or re-sell software, except as specifically set forth in this Reseller Agreement and Exhibit B.
- 8.3. Neither Ericsson nor any Customer may alter, amend, decompile, duplicate or customize any Calix software or other IP. Any such act shall constitute a breach of and shall void all warranties, and authorize Calix to revoke all further rights to such Ericsson and demand the return of the Products and software and the immediate cessation of its use.

9. Source Code Escrow.

- 9.1. Calix agrees to enter into and execute an escrow agreement ("Escrow Agreement") covering the software source code for the GPON 1500 portfolio and any derivative products (the "Source Code") within two weeks from such request by Ericsson. The Escrow Agreement shall be entered into by the Parties with a mutually agreeable third-party escrow agent ("Escrow Agent").
- 9.2. The Escrow Agreement to be entered into shall be the Escrow Agent's model escrow agreement (as amended from time to time by the Escrow Agent) with the amendments set out below incorporated into the said agreement. The Escrow Agreement shall be valid for a minimum period of five years from its execution unless terminated as stated in the Escrow Agreement.
- 9.3. Upon the execution of the Escrow Agreement, Calix agrees to deliver, within 30 days, a sealed package containing the Source Code (i.e. the human readable embodiment of software, whether such embodiment is contained on paper or other form of media) and related documentation for deposit with the Escrow Agent in accordance with the Escrow Agreement. Calix shall, from time to time, deposit with the Escrow Agent any and all updates and upgrades of the Source Code, and related documentation. The Source Code, including updates and upgrades thereto deposited with the Escrow Agent under the Escrow Agreement shall be considered the "Material."
- 9.4. Calix represents and warrants that the Material will at all times set forth the true and complete Source Code.
- 9.5. <u>Events Of Release</u>. Ericsson will be entitled to request release and delivery of the Material if one or more of the following circumstances ("Events of Release") have occurred:
 - 9.5.1. if Calix becomes the subject of any proceeding under any bankruptcy, receivership, insolvency or similar laws instituted by or against Calix, which proceedings are not dismissed as to Calix within 60 days of being instituted; or
 - 9.5.2. if Calix is in material breach of the Maintenance and Support Appendix (Exhibit B) which is not cured within 60 days of Calix's receipt of notice of such material breach from Ericsson.
- 9.6. Delivery Of The Material To Ericsson . Escrow Agent shall distribute the Material to Ericsson no sooner than 15 days after receipt of the request of release of the Material. Ericsson shall deliver notice to Calix of any such release request on or before the date on which Ericsson deliver's such request to Escrow Agent, and should Calix, within ten days of its receipt of a copy of the Ericsson request for release, submit a written statement to the Escrow Agent stating that an Event of Release has not occurred or has been cured, the Escrow Agent shall not release the Material to Ericsson until Ericsson submits a written statement to the Escrow Agent from Skandinaviska Enskilda Banken in Stockholm (the "Bank") stating that an amount of [***] the "Security Amount") has been deposited with the Bank to secure Ericsson's undertaking in accordance with this Article. Following submission by Ericsson of the above mentioned statement from the Bank, the Escrow Agent shall promptly, on the request of Ericsson, release the Material to Ericsson.
 - 9.6.1. Unless Calix initiates legal action in accordance with the Escrow Agreement within 30 days of the forwarding of its statement in accordance with Section 9.6, above to the Escrow Agent, Ericsson is entitled, after a request to the Bank, to re-payment of the Security Amount.

- 9.6.2. Should Calix initiate legal action, the sole question to be determined is whether or not an Event of Release had occurred, and had not been cured within the applicable time period, at the time when Ericsson forwarded its request for release of the Material to the Escrow Agent. If the decision of the trier of fact is in favor of Calix, Ericsson shall promptly return the Material to Calix and pay to Calix an amount equal to the Security Amount, which shall constitute full compensation for the temporary use of the Material by Ericsson. If the decision of the arbitral tribunal is in favor of Ericsson, Ericsson is entitled, after a request to the Bank, to repayment of the Security Amount.
- 9.7. <u>Use Of The Material Released</u>. Upon the release of the Material to Ericsson, Ericsson is automatically granted, free of charge, a limited, non-exclusive, non-sublicensable, irrevocable, non-transferable license to use, copy, sublicense, modify and develop the Material, software and support materials for the limited purposes of the manner in which such Material was used by Calix prior to release. Ericsson shall not disclose the Material to any person or entity, except Ericsson employees to the extent necessary for such support and maintenance as provided in this Reseller Agreement. If the Materials have been properly released to Ericsson prior to termination of this Reseller Agreement, the above license shall survive termination so long as any Ericsson Customer continues to use the support services related to the Material, and for a period of 12 months thereafter. Calix undertakes, at the request of Ericsson, to provide Ericsson with reasonable support in order to enable Ericsson to successfully use the Material for the purposes stated in this Section 9.
- 9.8. <u>Termination By The Escrow Agent</u>. Should the Escrow Agreement be terminated by the Escrow Agent, Escrow Agent shall promptly deliver the Material to such successor escrow agent as is designated by Calix and Ericsson.

10. Advertising and Promotion.

- 10.1. Calix may supply sales literature for the Products and Services to Ericsson in reasonable quantities and at no charge. Ericsson shall use only such sales literature and promotional materials for the Products and Services as supplied by Calix or which have been previously approved in writing by Calix.
- 10.2. Calix may assist Ericsson in the promotion of the sale of the Products and Services by participation or assistance in seminars, trade shows and advertising, as agreed by the Parties in writing.
- 10.3. Neither Party shall use any trade names, trademarks or logos owned by the other Party in any advertising, sales literature or promotional material without the specific advance written consent of the Party owning such trade names, trademarks or logos. Ericsson's use of Calix trade names, trademarks or logos shall be limited to its activities in support of the sale of Products and Services under this Reseller Agreement.

11. Confidentiality.

- 11.1. Each Party agrees to hold in confidence information disclosed by the other Party which is designated in writing as confidential ("Confidential Information"). Each Party agrees to: (i) protect the confidentiality of the Confidential Information; (ii) refrain from using the Confidential Information except as contemplated by this Reseller Agreement; and (iii) not disclose Confidential Information to any third party.
- 11.2. Confidential Information shall not include information which: (i) was in the Party's possession or known to it, without restriction, at the time of disclosure, as proven by files then in existence; (ii) is or becomes public knowledge through a source other than, and without fault of, the Party; (iii) is independently developed without reference to any Confidential Information; (iv) is or becomes available from a third-party source without breach of any obligation of confidentiality; (v) is disclosed with the other Party's prior written approval; or (vi) is disclosed under order or requirement of a court, administrative agency, or governmental body, provided however, that the disclosing party shall (A) provide prompt notice to the other Party to enable that Party to seek to prevent or restrict such disclosure, and (B) disclose only that portion of Confidential

Information required to comply with the order or requirement.

11.3. Neither Party shall disclose any terms of this Reseller Agreement to any third party without prior written consent of the other Party; however, either Party may disclose the terms of this Reseller Agreement: (i) to legal counsel; (ii) in confidence, to accountants, banks, and finance sources and their advisors solely for purposes of securing financing; (iii) in connection with enforcement of this Reseller Agreement; or (iv) in confidence, in connection with an actual or proposed merger or acquisition, solely for use in the due diligence investigation in connection with such transaction.

12. Indemnification.

12.1. General. Each Party shall indemnify, defend and hold harmless the other, its directors, shareholders, employees and representatives against any claim, suit or proceeding ("Claim") brought against the other for damages resulting from the negligence or willful misconduct of the defending Party, or from any actual or third party claim arising out of or based upon defending Party's performance or failure to perform under this Reseller Agreement.

12.2. IPR Infringement.

- 12.2.1. Calix shall indemnify, defend and hold Ericsson and its customers harmless against any Claim brought by a third party against Ericsson alleging that any Product listed on Exhibit A as a "Calix Product" or any Service infringes any IPR of that third party. In the event that such claim is upheld in a court of law, or as part of a settlement of such claim, Calix shall, at its option, a) remove the infringing portion from the Product or Service without substantially reducing functionality, b) replace or modify the infringing Product or Service such that it is no longer infringing, or c) procure for Ericsson the right to continue using the infringing Product or Service. Calix agrees to use commercially reasonable efforts to implement either (a), (b), or (c), however, in the event that none of these options is commercially reasonable, Calix's total liability shall be to refund the amount paid to Calix for the Product or Service.
- 12.2.2. Calix shall have no indemnification obligations with respect to any Product listed on Exhibit A as a "1500 Legacy" or "ONT Legacy" Product, so long as the claim is based on the Specifications of the Product at issue as of the Closing Date of the Asset Purchase Agreement.
- 12.2.3. Calix shall have no indemnification obligations with respect to any third party action alleging that the use of any Product or Service in combination with products not supplied by Calix infringes any third party IPR where the Calix Product or Service standing alone does not infringe.
- 12.3. Each Party's indemnification obligations are subject to the indemnified Party (i) notifying the indemnifying party promptly in writing of a Claim, (ii) giving the indemnifying Party exclusive control over the defense or settlement of the Claim, and (iii) providing all reasonable assistance to the indemnifying Party (at indemnifying Party's expense).

13. Limitation of Liability.

13.1. Except as expressly provided in this article or elsewhere in this Reseller Agreement, neither Party shall in any event be liable to the other Party under this Reseller Agreement for loss of production, loss of business, loss of data or revenue or for any special, indirect, incidental, punitive, or consequential damages, whether or not the possibility of such damages could have been reasonable foreseen.

- 13.2. Except for claims arising under Section 12(Indemnification) or Section 11 (Confidentiality) of this Reseller Agreement (for which each Party accepts unlimited liability for direct damages): (a) in no event shall either party's aggregate liability under this agreement exceed the sum of [***] or the aggregate amount paid to Calix pursuant to the agreement during the preceding [***], whichever is the greater.
- 13.3. The financial limit mentioned in Section 13.2, shall not apply to damages for bodily injury (including death) and damage to real property and tangible personal property.
- **14.** <u>No Joint Venture or Agency</u>. Nothing in this Reseller Agreement shall create a joint venture, partnership, or similar arrangement between the Parties. Neither Party is authorized to act as agent for the other Party except as expressly stated in this Reseller Agreement.

15. Export Obligations.

- 15.1. The export, resale, use or re-export of Products or Services may require prior approval of the US Government. Ericsson agrees to abide by all US and other laws and regulations pertaining to export, resale, use or re-export, violation of which constitutes a breach of this Reseller Agreement, voids Calix warranties, and automatically terminates Calix's obligations to support the Products or perform under this Reseller Agreement. Furthermore, Calix shall be excused from performance, and shall not be held liable, for failure to deliver Products resulting from US Government delay, denial or withdrawal of any export approval of any Products or Services. These export control obligations shall survive termination of this Reseller Agreement.
- 15.2. Calix is responsible for obtaining and maintaining any export license(s) required for delivery of the Products or Services to Customer.
- 15.3. If Calix is unable to obtain or maintain the export license(s), Ericsson may terminate any related contract or part thereof, which may be affected by the license.
- 15.4. Calix shall inform of and issue all documentation which may be required by law, regulation or reasonably requested by Ericsson regarding the export, import or re-export of the Products and Services. In particular, Calix is responsible to continuously provide detailed technical documentation, certificate of origin and relevant data including but not limited to HST (harmonized system tariff), CCATS, (commodity classification automated tracking system), COO (country of origin), and ECCN code (export control classification number) for the Products and Services according to the export administration regulations of the member countries of the European Union and the United States of America, and/or corresponding data according to other applicable regulations. The documentation shall be updated by Calix on an ongoing basis when updates of the Software are provided or when new regulations come into effect.
- 15.5. Calix guarantees that the documentation provided to Ericsson is correct. Calix shall provide the documentation requested by Ericsson within 30 days from request.
- **16.** <u>Force Majeure</u>. Calix shall have no liability for delays, failure in performance or damage due to fire, flood, explosion, power failure, civil disturbance, acts of civil or military authorities or the public enemy, strikes or labor disputes or acts of God.
- 17. <u>Assignment</u>. This Reseller Agreement shall accrue to the benefit of and be binding upon the Parties and any successor entity into which either Party shall have been merged or consolidated. Neither Party may assign this

Reseller Agreement without the prior written consent of the other Party, which consent shall not be unreasonably withheld or delayed. Any attempted assignment in contravention of this clause shall be void and ineffective.

- 18. Governing Law and Venue. This Reseller Agreement and any Purchase Orders, as well as any legal matters connected therewith, shall be governed by the laws of England and Wales. The parties shall use their good faith to settle by amicable negotiations any difference which may occur between them in connection with this Agreement within 30 business days of notice of such dispute. Such efforts shall include escalation of such dispute to the corporate officer level of each party. If the parties fail to reach such an amicable settlement, either party may refer such differences to arbitration as provided below. Any dispute, controversy or claim arising out of or in connection with this Agreement or any Contract, or the breach, termination or invalidity thereof, shall be finally settled by arbitration in accordance with the rules of arbitration of the International Chamber of Commerce (ICC rules) by one arbitrator appointed in accordance with such rules. The place of arbitration shall be London. The language to be used in the arbitral proceedings (including documents submitted) shall be English. All awards may if necessary be enforced by any court having jurisdiction in the same manner as a judgment in such court.
- 19. Anti-Bribery Compliance. Each Party agrees to comply with all applicable law, including, as may be applicable, the United States Foreign Corrupt Practices Act and the UK Bribery Act 2010. In carrying out its duties and responsibilities under this Reseller Agreement, Ericsson shall at all times comply with the Calix Anti-Bribery Policy ("Policy"), which is attached as Exhibit D, and shall also comply with the terms of each of the United States Foreign Corrupt Practices Act and the UK Bribery Act 2010 (collectively the "Acts"). Ericsson shall annually certify its compliance with the Policy by completing the certification attached to Exhibit D. In particular, Ericsson and its employees and agents shall not directly or indirectly offer, promise, authorize or give an advantage or anything of value for the purpose of inducing a person or organization to perform improperly a relevant function (including a decision not to act) or reward a person or organization for an improper function to assist Calix, such as giving an individual a gift to induce them to use their influence to affect any such act or decision in order to assist Calix in obtaining, retaining, or managing any such business. In addition, Ericsson will not request, agree to, receive or accept any advantage or anything of value for a relevant function to be performed improperly. Ericsson agrees to maintain proper internal accounting controls and will not knowingly falsify any book, record or account or attempt to circumvent such accounting controls.

20. Quality and Code of Conduct.

- 20.1. Calix shall comply with the Ericsson's Code of Conduct as described in document number EN/LZT 108 8173 R1A, available from the following website: http://www.ericsson.com/ericsson/corporate_responsibility/suppliers/index.shtml
- 20.2. If Calix does not comply with the Code of Conduct or equivalent code of conduct, Calix shall on or before the execution of this Agreement provide Ericsson with a plan for implementation of the said code.
- 20.3. Calix undertakes to comply with the applicable requirements in the ISO 9000 quality system standards.
- 20.4. When delivering the Products, Calix shall comply at all times with the latest version of Ericsson's directive regarding banned and restricted substances. Appendix 2 (Banned and restricted substances) contains the version valid at the date of the signing of this Agreement. The at all times latest version can be found at: http://www.ericsson.com/ericsson/corporate_responsibility/suppliers/index.shtml
- 21. <u>WEEE Directive Compliance</u>. Ericsson is responsible for ensuring that its resale of Calix Products and Services in each nation within the European Union is in compliance with Directive 2002/96/CE WEEE on Waste of Electrical and Electronic Equipment ("WEEE"). Ericsson is specifically responsible for WEEE product registration of each Calix Product. Ericsson is also responsible for collecting waste from its Customers, or for directing that waste to an authorized recycling center.

- **22.** <u>Notice</u>. Any notices required or permitted shall be given to the appropriate Party at the address specified above, or at such other address as the Party shall specify in writing, and shall be effective upon actual receipt.
- **23.** <u>Waiver</u>. The failure of either Party to insist upon strict performance of any obligation under this Reseller Agreement, regardless of the length of time for which such failure continues, shall not be a waiver of such Party's right to demand strict compliance in the future. No consent or waiver, express or implied, to or of any breach or default in the performance of any obligation shall constitute a consent or waiver to or of any other breach or default in the performance of the same or any other obligation.
- **24.** Severability. In the event any one or more of the provisions of this Reseller Agreement is held to be unenforceable under any law, such unenforceability shall not affect any other provision of this Reseller Agreement. This Reseller Agreement shall be construed as if the unenforceable provision(s) had not been contained in it, and the Parties shall negotiate in good faith to replace the unenforceable provisions by such modified provisions as have the effect nearest to that of the provisions being replaced.
- **25.** Entire Agreement. This Reseller Agreement constitutes the entire understanding between the Parties concerning its subject matter and supersedes all prior discussions, agreements, understandings, and representations, including the conflicting terms of any purchase order or other document, whether oral or written and whether or not executed by Calix or Ericsson. No modification, amendment or other change may be made to this Reseller Agreement unless in writing and executed by authorized representatives of both Parties.
- **26.** English Language. This Reseller Agreement is in the English language only, which language will be controlling in all respects, and any versions in any other language will not be binding on the Parties. All communications and notices to be made or given under this Reseller Agreement will be in English.
- **27.** Execution. This Reseller Agreement may be executed in one or more counterparts, each of which shall be deemed an original, and all of which together shall constitute one agreement. The Parties agree that signatures delivered by fax, scan, email or other electronic means shall be considered original signatures for all purposes under this Reseller Agreement.

The Parties have executed this Reseller Agreement as of the Effective Date.

CALIX, INC.		ERICSSON AB	
By:		Ву:	
	(signature)		(signature)
Name:		Name:	
Title:		Title:	
		Ву:	
			(signature)
		Name:	
		Title:	<u> </u>

EXHIBIT A

PRODUCTS AND PRICE SHEET

Ericsson agrees to purchase and Calix agrees to sell the following Products:

[***] [***]

[***] [***]

This schedule pertains to EDA releases 12A and below:

PON Products Warranty Services

Calix will repair OLT Products shipped by Ericsson for a fee of [***] per unit. Purchasing and invoicing for these services should be sent directly to the Ericsson regions.

PON Products Technical Services

Ericsson Shall pay to Calix for Technical Services for PON Products an annual fee for Calix's creation of a dedicated support team of: The annual fee for 2013 is a fixed amount and will be paid per the MSA, while the amounts for 2014 and beyond will be fixed at the end of the prior year (last quarter) at the CSR/TR annualized run rate, with any overages billed at the variable rate described below.

2013: [***] Including 500 CSR, 200 TR and 4 aggregated correction packages

2014: [***] Including 200 CSR, 100 TR - Number of correction packages negotiated last quarter 2013

2015: [***] Including 100 CSR, 40 TR - Number of correction packages negotiated last quarter 2014

2016+: [***] per year Including 50 CSR, 20 TR - Number of correction packages negotiated last quarter 2015

In addition to the above Ericsson fixed prices, Ericsson shall pay Calix a variable fee of [***]/ SR and [***]/ TR above agreed number of SR, TR and correction packages included in price.

On-Site Support Services

Ericsson shall reimburse Calix for on-site support Services at the rate of [***] per day plus all travel expenses. Ericsson and Calix shall specify an approval process for this On-Site support. Purchasing and invoicing for these services should be sent directly to the Ericsson regions.

For the following [***]Customers, the above pricing will apply, but the Parties will in good faith review on a deal-by-deal basis whether the inclusion of Calix Products may allow for changes in the above pricing. However, Calix will have no obligation to make changes in the above pricing for these Customers.

[***]

EXHIBIT B

MAINTENANCE AND SUPPORT APPENDIX

۲×	< >	κ>	۲1

EXHIBIT C

ERICSSON CUSTOMERS

г	*	*	*	-

EXHIBIT D

CALIX, INC. ANTI-BRIBERY COMPLIANCE POLICY

1.0 INTRODUCTION AND PURPOSE STATEMENT

The Foreign Corrupt Practices Act ("FCPA") is a US federal law that applies to both individuals and businesses. All Calix, Inc. ("Calix") business around the world falls within the scope of the FCPA. Violations of the FCPA subject offending parties to severe criminal and civil penalties.

The Bribery Act 2010 (UK Act) is an Act of the Parliament of the United Kingdom that applies to both individuals and businesses. Due to Calix's presence in the UK, all Calix business around the world falls within the scope of the UK Act. Violations of the UK Act subject the offending parties to severe criminal penalties.

Calix is committed to full compliance with both the letter and the spirit of the FCPA and UK Act (collectively, "the Acts"). The purpose of this policy ("Policy") is to alert employees, officers, directors, agents and contractors of Calix and its subsidiaries to the requirements of the Acts and to establish best practices and recordkeeping procedures that assure that all transactions undertaken by Calix, both internationally and domestically, are in compliance with the Acts.

2.0 THE ACTS IN SUMMARY

The Acts specifically prohibit Calix, and individuals acting on Calix's behalf, from offering, promising, or giving "anything of value," or providing an "advantage" to another person or entity with the intent of inducing that person or entity to take any improper action, or rewarding that person for taking an improper action. Stated more concisely, the Acts strictly prohibit the payment of bribes, kickbacks or other inducements. Receiving a bribe can also violate the Acts.

The Acts prohibit both direct and indirect offers and payments. Therefore, Calix and Calix Representatives (as defined below) are potentially liable for any actions made through an agent, consultant, or other third-party intermediary with the knowledge or belief that it will violate the Acts. "Knowledge" includes a conscious disregard for or deliberate ignorance of facts that indicate a high probability that the payment or offer will occur. In other words, Calix cannot use another party to do indirectly anything which it cannot do directly.

Calix can also be liable if it fails to prevent a bribe. Therefore, it is important that Calix take strong steps to ensure that all Calix employees comply with this Policy.

Although the FCPA prohibits improper payments to "foreign officials," the UK Act is not so limited. Therefore, this policy applies to offers and payments to <u>any</u> individual or entity: there is no difference between bribing a public official, private citizen, government agency or private corporation. The terms "anything of value" and "advantage" are also interpreted very liberally, and can include items such as travel, entertainment, or lavish meals.

The Acts require U.S. companies to maintain books and records that accurately and fairly reflect international corporate transactions, and require that companies establish a system of internal accounting controls to provide reasonable assurance to management of the type of financial transactions undertaken by the company. Under the accounting standards provisions of the Acts, even payments that are legal under the Acts must be properly recorded in the accounts and records of the company. Recording of such payments in any way that would conceal their true nature constitutes an independent violation of the Acts accounting standards.

Statutory penalties for violation of the Acts include unlimited per violation fines for companies, and unlimited fines and imprisonment for up to fifteen years per violation for individuals.

3.0 THE CALIX POLICY AND PROCEDURES

To ensure that Calix remains in compliance with the Acts, all employees, officers, directors, agents and contractors of Calix and its subsidiaries (collectively "Calix Representatives") must comply with this Policy, and the procedural guidelines below.

- 3.1 <u>Scope</u>. This Policy applies to all Calix Representatives. It is designed to help them recognize situations and payments that might raise legal issues under the Acts. It is important that each Calix Representative with responsibilities that might give rise to potential liabilities resulting from the Acts be aware of this Policy, comply with its procedures, and work closely with Calix's Legal and Finance Departments to ensure compliance (See Section 4.2).
- 3.2 <u>Responsibilities</u>. Each Calix Representative is responsible for recognizing, avoiding and reporting any situation involving practices that may be illegal under the Acts. Calix Representatives are further responsible for complying with this Policy and its record-keeping procedures. All managers are responsible for communicating this Policy to Calix Representatives under their supervision. To ensure compliance with this Policy, Calix Representatives involved in retaining agents and consultants are also required to have such agents and consultants certify compliance with this Policy.
- 3.3 Prohibited Payments and requests. No Calix Representative shall offer, promise, or give anything of value or any advantage to any other person or entity with the intent or knowledge that such action may induce that person or entity to take any improper action. This prohibition applies to the use of corporate as well as personal funds or assets. It also applies to indirect contributions, payments, or gifts made through any consultants, agents, advisors, suppliers, customers, or other third parties. No Calix Representative shall use Calix assets to make any charitable or political contributions unless approved in advance by the Calix Legal Department, and Calix Representatives shall take care that any personal charitable or political contributions are not made for purposes that would be improper under this Policy. No Calix employee shall request, agree to receive, or accept an advantage intending that a relevant function or activity should be performed improperly.
- 3.4 <u>Permitted Payments</u>. Certain payments to foreign officials or entities may not be prohibited by the Acts. However, it can be extremely difficult to distinguish between payments that are legal and illegal under the Acts. Consequently, <u>any</u> proposed payment to a foreign official or entity must be approved by the Calix Legal Department or CFO before any such payment is made, and must be accurately reported as a business expense.
- 3.5 <u>General Policies</u>. Calix Representatives are to conduct Calix business in compliance with the written laws of the US, UK, and of all countries in which Calix does business. In particular, no Calix Representative shall:
- 3.5.1 use Calix funds or assets for any unlawful, improper or unethical purpose;
- 3.5.2 establish any undisclosed or unrecorded funds or assets of Calix for any purpose;
- 3.5.3 make any false, inflated, or artificial entries in the books and records of Calix for any reason, or engage in any arrangement that results in such entries;
- 3.5.4 falsify any accounting document relating to any transaction, or otherwise take any action with respect to such a document that may obscure or disguise the true nature of the underlying transaction;
- 3.5.5 approve a payment on behalf of Calix without adequate supporting documentation or with the intention or understanding that any part of such payment is to be used for any purpose other than that described by the documents supporting the payment; or
- 3.5.6 fail to comply with generally accepted accounting principles and established internal audit controls and procedures.

- 3.6 <u>In Case of Doubt</u>. If a Calix Representative has any doubt as to the legality under the Acts or this Policy of any planned payment to any individual or entity, or the accuracy of financial reporting with respect to any transaction, the Calix Representative should immediately contact and discuss the matter with the Calix Legal or Finance Departments prior to making any such payment or recording such financial information.
- 3.7 <u>Duty to Report Potential or Actual Violations</u>. Any Calix Representative who suspects that a violation of either of the Acts is about to occur, shall immediately report the facts supporting that suspicion to his or her manager, to Calix's General Counsel, or anonymously via the online or phone methods used to report concerns regarding violations of the Calix Ethics and Business Conduct Policy. Any Calix manager receiving such report shall immediately report it to Calix's General Counsel. If any Calix Representative believes that an actual violation of this policy has occurred, he or she shall immediately report the violation via the reporting methods detailed in the Calix Whistleblower Policy, which are accessible via the main page of CPoint.
- 3.8 <u>Annual Certification</u>. All Calix Representatives shall certify, on an annual basis, that they have reviewed and understood this Policy, and will comply with it. New Calix employees shall make such certification as part of their orientation process.
- 3.9 <u>Consequences of Noncompliance</u>. In addition to those penalties which may be assessed by the US and UK governments against individuals who violate the Acts, Calix shall take strong disciplinary action, up to and including termination of employment, against any Calix Representative who violates this Policy.

4.0 RETENTION OF INTERNATIONAL AGENTS OR CONSULTANTS

- 4.1 <u>Due Diligence</u>. Calix Representatives shall perform careful due diligence before retaining any foreign agents or consultants, or any agents or consultants retained in the US to provide services outside of the country. The level of due diligence to be performed may increase, depending upon the nature of the proposed relationship, the amount of compensation proposed, the location of the agent and other factors. At a minimum, the due diligence should include a search of the databases available from the U.S. Departments of State, Treasury and Commerce, to ensure that the organization and its principals are not listed as debarred, denied or specially designated nationals. The results of this due diligence should be reviewed with the Calix Legal Department prior to formally retaining such agent or consultant. The Legal Department must also review and approve all contracts with such entities before they are signed.
- 4.2 <u>Certification</u>. If such an agent or consultant is retained, the Calix Representative supervising the agent or consultant shall obtain a signed agreement by which the consultant or agent agrees to comply with the Acts (in substantially the form attached). This language may be included in an overall definitive agreement with the agent or consultant. If such agents or consultants are engaged by Calix for more than one year, they shall be asked to certify their compliance with this Policy, in writing, on an annual basis.
- 4.3 <u>Red Flags</u>. Calix Representatives should be aware of certain "Red Flags," which may indicate a questionable transaction with an agent or consultant, such as:
- 4.3.1 requests for unusual payments or financial arrangements, such as: payments to a numbered bank account; payments to accounts in countries other than where the agent or consultant is located or the business is to be performed; or cash payments.
- 4.3.2 unusually high commission rates or seemingly inflated invoices;
- 4.3.3 a history of or reputation for corruption in the country;
- 4.3.4 a lack of transparency in expense and accounting records; or
- 4.3.5 a close relationship between the agent or consultant and the foreign government or contracting individual.

5.0 CONCLUSION

This Policy contains general guidelines for conducting Calix business consistent with the requirements of the Acts. If you
have any questions about this Policy, please contact your supervisor or Calix's General Counsel or Chief Financial
Officer. Calix expects all employees, officers, directors, agents and contractors to comply with this Policy.

Calix reserves the right to amend or supplement this Policy, without prior notice, at any time.

CERTIFICATION OF COMPLIANCE WITH CALIX ANTI-BRIBERY COMPLIANCE POLICY

Ericsson AB ("Ericsson") certifies that:

- 1. Ericsson has reviewed the attached copy of the Calix, Inc. Anti-Bribery Compliance Policy ("Policy"), and understands and agrees to follow the Policy with respect to all work on behalf of Calix, Inc. or any of its subsidiary or affiliated entities (collectively, "Calix");
- 2. Ericsson is familiar with the requirements of the U.S. Foreign Corrupt Practices Act ("FCPA") and the Bribery Act 2010 ("UK Act") (collectively "the Acts") [The FCPA can be found at http://www.justice.gov/criminal/fraud/fcpa/ statutes/regulations.html, and the UK Act can be found at http://www.legislation.gov.uk/ukpga/2010/23/contents];
- 3. Ericsson shall take no action that might cause Calix to be in violation of the Policy, the Acts, or the laws of any country regarding corrupt payments;
- 4. Ericsson is not affiliated with or employed by any government or governmental agency, that no government or governmental agency has an ownership or financial interest in Ericsson, and that Ericsson does not employ anyone who is also affiliated with or employed by any government or governmental agency;
- Neither Ericsson, nor any other person, including every employee, representative and agent affiliated with Ericsson, has made, offered to make or agreed to make any loan, gift, donation or other payment, directly or indirectly, whether in cash or in kind to or for the benefit of any individual or entity (including government officials, political parties, and candidates) in order to influence performance of an improper action for Calix;
- Ericsson will maintain written books and records in accordance with Generally Accepted Accounting Principles, that reflect all expenditures made by Ericsson on behalf of Calix, and that clearly and accurately identify the persons or entities that receive such payments. Upon request, Ericsson shall makes such books and records available to Calix for review;
- 7. Should Ericsson learn of or have reason to know of any payment, offer, or agreement to make a payment to any individual, entity (including government officials, political parties, and candidates) for the purpose of influencing performance of an improper action for Calix, Ericsson will immediately advise the General Counsel of Calix, Inc., in writing, of such knowledge or suspicion;
- Ericsson agrees to fully cooperate with Calix in any investigation of a potential violation of the Policy; and
- Friesson understands and agrees that any violation of the Policy by Friesson shall constitute a material breach of this

certification, and shall be grounds	•	•		
Ву:				
Company:				
Title:				
Date:				

Exhibit C

License Agreements

Copyright As	ssignment and License Agreemen
	ERICSSON AB
	and
	CALIX, INC.

TABLE OF CONTENTS

- 1. Definitions
- 2. ASSIGNMENT AND LICENSEs
- 3. NO WARRANTIES
- 4. MISCELLANEOUS
- 5. GOVERNING LAW AND DISPUTES

SCHEDULES

Schedule 1 Assigned Copyrights Schedule 2 Licensed Copyrights This **Copyright Assignment and License Agreement** is dated _____ and made between:

- (1) **Ericsson AB**, Reg. No. 556056-6258, a limited liability company incorporated under the laws of Sweden, and having its registered office at 164 83 Stockholm ("**Ericsson**"); and
- Calix, Inc., a Delaware corporation with offices at 1035 N. McDowell Blvd., Petaluma, CA 94954 ("Calix"). Capitalized terms shall have the meaning ascribed to them in Section 1 below, or if not otherwise defined, as set forth in the Asset Purchase Agreement (as defined below).

Background

- A. Ericsson has on August 20, 2012 entered into an Asset Purchase Agreement (the "**Asset Purchase Agreement**") with Calix under which Ericsson has agreed to (i) procure the transfer of the Business to Calix.
- B. Ericsson is the owner of certain Unregistered Intellectual Property Rights (as defined below) which are used within the Business.
- C. Ericsson wishes to assign certain Unregistered Intellectual Property Rights comprising the Assigned Copyrights and license certain other Unregistered Intellectual Property Rights comprising the Licensed Copyrights, to Calix and Calix wishes to acquire the Assigned Copyrights and license the Licensed Copyrights on the following terms and conditions.
- 1. **Definitions**
 - "Affiliate" has the meaning set forth in Rule 144 under the Securities Act.
 - "Agreement" means this Copyright Assignment and License Agreement;
 - "Assets" means as set out in the Asset Purchase Agreement;
 - "Asset Purchase Agreement" means as set out in the Background;
 - "Assigned Copyrights" means the Unregistered Intellectual Property Rights identified in <u>Schedule 1</u> and exclusively used in the Business at the time of Closing Date;
 - "Business" means as set out in the Asset Purchase Agreement;
 - "Business Day" means as set out in the Asset Purchase Agreement;
 - "Calix" means as set out in the heading to this Agreement;
 - "Change of Control" means (i) the acquisition (through an event or series of events whether directly or indirectly, and whether alone or in combination with related entities or persons) of at least 50 per cent of the shares representing at least 50 per cent of the votes in Calix by a third party; (ii) any merger, consolidation, reorganization or other transaction involving Calix (or any series of related transactions) as a result of which any third party thereafter directly or indirectly controls Calix. For the purposes of the foregoing, "control" means having the direct or indirect power to appoint or remove all, or the majority of, the directors of the board of Calix, or to have the direct or indirect power to direct the operating and financial policies of Calix. A change of control of the Business through a business transfer shall for all purposes of this Agreement be deemed to constitute a Change of Control as defined in this Agreement;
 - "Closing Date" means as set out in the Asset Purchase Agreement;

- "Disclosing Party" means as set out in Section 4.1;
- "Ericsson" means as set out in the heading to this Agreement;
- " Ericsson Products" means Ericsson's GPON 1500 portfolio (OLT, ONT), and Entriview management system;
- "Field of Use" means the design, manufacture, marketing, sale, use and support of the Ericsson Products;
- "Know-How" means confidential and proprietary industrial, technical and commercial information and techniques including data, formulae, drawings, test results, reports, specifications, component lists, instructions, manuals, brochures, catalogues and process descriptions;
- "Licensed Copyrights" means the Unregistered Intellectual Property Rights owned or controlled by Ericsson or its Affiliates as of the Closing Date and used in, held for use in or applicable to the operation of the Business, including without limitation the design, development, marketing, manufacture, use, sale, offer for sale, importation and support of Ericsson Products and derivative works of and modifications and improvements to such Ericsson Products. The Licensed Copyrights includes without limitation the Unregistered Intellectual Property Rights identified in Schedule 2;
- "Non-Disclosing Party" means as set out in Section 4.1;
- "Party/ies" means either of Ericsson or Calix or Ericsson and Calix jointly as the case may be;
- "Unregistered Intellectual Property Rights" means all non-patent and non-trademark intellectual property rights, including without limitation copyrights (including copyrights in software), neighbouring rights, database rights, unregistered designs, trade secrets, and Know-How. "Unregistered Intellectual Property Rights" includes registered copyrights (including copyrights in software) and any applications for such registration, but does not include any other registered intellectual property rights, such as patent or trademark rights, or any applications for such registrations.

2. **ASSIGNMENT AND LICENSES**

2.1 **Assignment**

- 2.1.1. Ericsson assigns, transfers, conveys and delivers to Calix all its right, title and interest in and to the Assigned Copyrights. Such assignment shall be deemed to have taken effect on the Closing Date.
- 2.1.2. Calix shall make reasonable efforts to procure that the assignment of any registered copyrights under this Agreement will be registered with the relevant authorities, and Ericsson authorizes such relevant authorities to carry out the foregoing. Each Party shall execute and deliver any additional documents and instruments and shall do any and all such other things as may be reasonably required to perfect or otherwise implement the assignment of the Assigned Copyrights pursuant to this Agreement. Notwithstanding this, any obligation upon Ericsson with reference to this Section 2.1.2 shall not extend beyond 90 days after the Closing Date. Calix shall bear all costs for the work related to the registration referred to in this Section 2.1.2 and shall pay all fees charged by local copyright agents and registration authorities in connection therewith.

2.2 License under Licensed Copyrights

2.2.1. Subject to the terms of this Agreement, Ericsson upon the Closing Date grants to Calix a non-exclusive, irrevocable, perpetual, worldwide, non-assignable (except as set forth in Section 4.4), non-sublicensable (except as set forth in Section 2.2), fully paid-up license under the Licensed Copyrights to make, have made, use, design, have designed, manufacture, repair, support, market, distribute, import, export, offer for sale, offer for lease, sell, lease and otherwise provide the Ericsson Products and related services solely within the Field of Use. This license grant shall extend to include Ericsson Products up to and including those made, used, designed, manufactured, repaired, supported, marketed, distributed,

imported, exported, offered for sale, offered for lease, sold, leased and otherwise provided through the completion of the Calix-announced end-of-support date for the last of the Ericsson Products.

2.2.2. The license granted by Ericsson to Calix under this Section 2.2 may only be sublicensed to (i) one or more Affiliates of Calix, however only for as long as such entities remain Affiliates of Calix, (ii) to Calix's and its Affiliates' end customers to the extent necessary for such customers' own use of services and products provided by Calix and its Affiliates, and (iii) to contractors of Calix or its Affiliates retained to perform any of the activities set forth in Section 2.2.1 on behalf of Calix or its Affiliates (each a "Sub-Licensee"). Calix shall ensure that each Sub-Licensee complies with the terms of this Agreement.

2.3 **Delivery**

- 2.3.1. Upon the effective date of this Agreement, Ericsson shall deliver or have delivered all tangible embodiments, including without limitation applicable software and documentation, and any applicable registrations or correspondence with registering entities regarding registrations, for all Assigned Copyrights. In addition, upon the effective date of this Agreement, Ericsson shall deliver or have delivered copies of tangible embodiments, including without limitation applicable software and documentation, for all Licensed Copyrights.
- 2.3.2. In addition, the parties acknowledge that Ericsson may inadvertently not deliver embodiments of the Assigned Copyrights or the Licensed Copyrights that should have been delivered to Calix as part of the contemplated delivery pursuant to the terms of this Agreement, and Calix may inadvertently receive copies of software or documentation that should not have been delivered. Each party agrees to engage in good faith discussions with the other regarding the delivery or return to the other any such later discovered software or documentation, at the reasonable request of the appropriate party, and deliver or return the applicable software or documentation as applicable.

3. **NO WARRANTIES**

- 3.1. Except as otherwise provided in the Asset Purchase Agreement, no representation, condition or warranty whatsoever is made or given by or on behalf of Ericsson or its Affiliates with respect to the Assigned Copyrights and all conditions and warranties relating to such rights (including as to validity or fitness for any purpose), whether arising by operation of law or otherwise, are expressly excluded.
- 3.2. Except as otherwise provided in the Asset Purchase Agreement, no representation, condition or warranty whatsoever is made or given by or on behalf of Ericsson or its Affiliates with respect to the Licensed Copyrights and all conditions and warranties relating to such rights (including as to validity or fitness for any purpose), whether arising by operation of law or otherwise, are expressly excluded.
- 3.3. The Parties acknowledge and agree that, except as otherwise provided in the Asset Purchase Agreement, the remedies provided for in this Agreement shall be Calix's sole and exclusive remedies with respect to the subject matter of this Agreement. Calix hereby waives any and all other rights, claims and causes of action with respect to the subject matter of this Agreement, whether based on contract or law, statute or legal principle (including the Swedish Sale of Goods Act or International Sale of Goods Act) which it may have against Ericsson or its Affiliates due to any defects or deficiencies in the Assigned Copyrights or the Licensed Copyrights.

4. **MISCELLANEOUS**

4.1. **Confidentiality**

- 4.1.1. Each Party undertakes not to use or disclose the contents of this Agreement or any information relating to the other Party and disclosed in connection with this Agreement, (such as, without limitation, Know How, trade secrets, customer lists) and which is not in the public domain unless:
 - (a) required to do so by law or under any order of court or other competent authority or tribunal;
 - (b) required to do so by any applicable stock exchange regulations or the regulations of any other recognized market place;
 - (c) such disclosure has been consented to by the other Party in writing; or
 - (d) the information is disclosed to its professional advisers who make no other use of the information than for assisting the Party and who are bound to such Party by a duty of confidence which applies to any information disclosed.
- 4.1.2. If a Party becomes required, in circumstances contemplated by (a) or (b) to disclose any information, such Party (the "**Disclosing Party**") shall provide the other Party (the "**Non-Disclosing Party**") with

prompt written notice of that fact so that the Non-Disclosing Party may seek (with the co-operation and reasonable efforts of the Disclosing Party) a protective order, confidential treatment or other appropriate remedy. In such event, the Disclosing Party shall consult with the Non-Disclosing Party regarding such proposed disclosure, shall furnish only that portion of the information which is legally required and shall exercise reasonable efforts to obtain reliable assurance that confidential treatment will be accorded such information to the extent reasonably requested by the Non-Disclosing Party.

4.1.3. The obligations set out above in this Section 4.1 shall survive for 10 years after the signing of this Agreement.

4.2. **Amendments and Waivers**

This Agreement may only be amended or modified by written agreement between the Parties. No failure or delay in exercising any right or power under this Agreement will be effective as a waiver unless it is in writing and signed by the Party granting the waiver, and no such waiver will constitute a waiver of any other right or power under this Agreement.

4.3. **Notices**

4.3.1. All notices and other communications required or permitted under this Agreement must be in writing in the English language, and shall be addressed as set out below or to such other addresses as may be given by written notice in accordance with this Section 4.3.

If to Ericsson:

Group Function Legal Affairs Ericsson AB

Torshamnsgatan 21 SE-164 83 Stockholm

Sweden Email: [***]

If to Calix: Calix, Inc.

1035 N. McDowell Blvd., Petaluma, CA 94954

USA

Attention: President and CEO

Fax:

With a copy to: General Counsel

Email: [***]

All such notices and other communications shall be effective upon actual receipt if delivered by hand, sent by reputable international overnight courier, or sent by email (with confirmation of receipt).

4.4. **Assignments**

This Agreement shall be binding upon and inure to the benefit of the successors of the Parties. Calix may transfer and assign this Agreement and the rights granted to it hereunder upon any Change of Control or the sale of all or substantially all of its assets to which this Agreement relates. Calix agrees to use commercially reasonable efforts to notify Ericsson not less than 30 days prior to any Change of Control or the sale of all or substantially all of its assets to which this Agreement relates. Ericsson may assign its rights and obligations hereunder to an Affiliate provided that the foregoing shall not limit the scope or subject matter of the licenses granted under this Agreement or Ericsson's obligations to fulfill its delivery obligations under this Agreement.

4.5. **Partial Invalidity**

If any provision of this Agreement or the application of it shall be declared or deemed void, invalid or unenforceable in whole or in part for any reason, the Parties shall amend this Agreement in order to give effect to, so far as is possible, the spirit of this Agreement. If the Parties fail to amend this Agreement, the provision, which is void, invalid or unenforceable, shall be deemed deleted and the remaining provisions of this Agreement shall continue in full force and effect.

5. GOVERNING LAW AND DISPUTES

- 5.1. The laws of England and Wales, excluding its conflict of law principles, govern all matters arising under this Agreement.
- All disputes, differences or questions arising out of or relating to the interpretation or performance of this Agreement, between the Parties, shall be finally settled in London, United Kingdom, under the Rules of Arbitration of the International Chamber of Commerce, by three (3) arbitrators, appointed in accordance with said Rules. The arbitration proceedings shall be conducted in the English language. The Parties hereby consent to such dispute being so resolved.
- 5.3. The Parties undertake and agree that all arbitral proceedings conducted or initiated with reference to this Section 5 will be kept strictly confidential. This confidentiality undertaking shall cover the fact that arbitration has been initiated, all information disclosed in the course of such arbitral proceedings, as well as any decision or award that is made or declared during the proceedings. Information covered by this confidentiality undertaking may not, in any form, be disclosed to a third party without the prior written consent of the other Party. This notwithstanding, a Party shall not be prevented from disclosing such information in order to safeguard in the best possible way such Party's rights vis-à-vis the other Party in connection with the dispute, or if such a right exists pursuant to statute, regulation, a decision by an authority, a stock exchange contract or similar.
- 5.4. In case this Agreement or any part of it is assigned or transferred to a third party, such third party shall automatically be bound by the provisions of this arbitration clause.

This Agreement has been duly executed in two identical originals, of which each Party has taken one.

ERICSSON AB	CALIX, INC.	CALIX, INC.	
Ву:	Ву:		
Name:	Name:	<u></u>	
Title:	Title:	<u> </u>	

SCHEDULE 1

ASSIGNED COPYRIGHTS

[***]	
[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission.	
Confidential treatment has been requested with respect to the omitted portions.	

SCHEDULE 2

LICENSED COPYRIGHTS

[*	*	*]

Patent License A	
dated	
TELEFONAKTIEBOLAGET	Γ LM ERICSSON
and	
CALIX, INC	

TABLE OF CONTENTS

- 1. Definitions
- 2. GRANT OF LICENSE
- 3. INFRINGEMENT
- 4. NO WARRANTIES
- 5. TERM AND TERMINATION
- 6. MISCELLANEOUS
- 7. GOVERNING LAW AND DISPUTES

SCHEDULES

Schedule 1 Licensed Patents

Schedule 2 Ericsson Products

This **Patent License Agreement**, dated ______, is and made by and between:

- Telefonaktiebolaget LM Ericsson, a limited liability company incorporated under the laws of Sweden, and having its registered office at 164 83 Stockholm ("**Ericsson**"); and
- (2) **Calix, Inc.**, a Delaware corporation with offices at 1035 N. McDowell Blvd., Petaluma, CA 94954 ("**Calix**").

Capitalized terms shall have the meaning ascribed to them in Section 1 below. Terms not defined herein shall have the meanings ascribed thereto in the Asset Purchase Agreement.

BACKGROUND

- A. Ericsson, Inc. has on August 20, 2012 entered into an asset purchase agreement (the "Asset Purchase Agreement") with Calix under which Ericsson, Inc. has agreed to transfer certain Assets to Calix.
- B. Ericsson's ultimate parent company, LM Ericsson (as defined below), is the owner of certain Licensed Patents (as defined below) which are used within the Business. LM Ericsson has entered into an agreement with Ericsson under which Ericsson is granted a right to sublicense the Licensed Patents to Calix on the terms and conditions set out in this Agreement.
- C. Ericsson wishes to license the Licensed Patents to Calix and Calix wishes to license the Licensed Patents on the following terms and conditions.
- 1. **Definitions**
 - "Affiliate" of any person means, as of any time, (i) any other legal person(s) directly or indirectly controlled by or under the common control of that first-mentioned person and (ii) any other legal person(s) directly or indirectly controlling or jointly controlling such first-mentioned person (whereby "control" means the ownership, directly or indirectly, of more than fifty (50) per cent of the (a) voting power or (b) capital or other securities of a controlled or commonly controlled legal person, and the term "controlling" shall have a meaning correlative to the foregoing);
 - "Agreement" means this Patent License Agreement;
 - "Assets" has the meaning set forth in the Asset Purchase Agreement;
 - "Asset Purchase Agreement" has the meaning as set out in paragraph A of the Background section;
 - "Business" means as defined in the Asset Purchase Agreement;
 - "Business Day" has the meaning set forth in the Asset Purchase Agreement;
 - "Calix" has the meaning as set out in paragraph (2) of the Background section;
 - "Change of Control" means (i) the acquisition (through an event or series of events whether directly or indirectly, and whether alone or in combination with related entities or persons) of at least 50 per cent of the shares representing at least 50 per cent of the votes in Calix by a third party; (ii) any merger, consolidation, reorganization or other transaction involving Calix (or any series of related transactions) as a result of which any third party thereafter directly or indirectly controls Calix. For the purposes of the foregoing, "control" means having the direct or indirect power to appoint or remove all, or the majority of, the directors of the board of Calix, or to have the direct or indirect power to direct the operating and financial policies of Calix. A change of control of the Business through a business transfer shall for all purposes of this Agreement be deemed to constitute a Change of Control as defined in this Agreement;
 - "Closing Date" means as set out in the Asset Purchase Agreement;
 - "**Disclosing Party**" means as set out in Section 6.1;
 - "Ericsson" means as set out in the heading to this Agreement;

- "Ericsson Group" means LM Ericsson and its Affiliates from time to time;
- "Ericsson Products" means Ericsson's GPON 1500 portfolio (OLT, ONT), and Entriview management system, which products are exhaustively listed in Schedule 2;
- "Field of Use" means the design, manufacture, sale, use and customary technical support of the Ericsson Products;
- "Know-How" means confidential and proprietary industrial, technical and commercial information and techniques including data, formulae, drawings, test results, reports, specifications, component lists, instructions, manuals, brochures, catalogues and process descriptions;
- "Licensed Patents" means the patents and patent applications set out in Schedule 1, as may be deemed supplemented pursuant to Section 2.3 hereof;
- "LM Ericsson" or "LME" means Telefonaktiebolaget LM Ericsson, corporate identity number 556016-0680;
- "Non-Disclosing Party" means as set out in Section 6.1;
- "Party/ies" means either of Ericsson or Calix or Ericsson and Calix jointly as the case may be;
- "Sub-Licensee" means as set out in Section 2.2.2;
- "Supplemental Patents" means as set out in Section 2.3.

2. **GRANT OF LICENSE**

- 2.1. Subject to the terms of this Agreement, Ericsson upon the Closing Date grants to Calix a non-exclusive, worldwide, non-assignable, non-assertable, non-sublicensable (subject to Section 2.2) and paid-up license under the Licensed Patents to make, have made, use, design, have designed, manufacture, repair, support, market, distribute, import, export, offer for sale, offer for lease, sell, lease and otherwise provide the Ericsson Products and directly related technical services solely within the Field of Use. This license grant shall extend to include Ericsson Products up to and including those made, used, designed, manufactured, repaired, supported, marketed, distributed, imported, exported, offered for sale, offered for lease, sold, leased and otherwise provided through the completion of the Calix-announced end-of-support date for the last of the Ericsson Products.
- The license granted by Ericsson to Calix under this Section 2 may only be sublicensed to one or more Affiliates of Calix ("Sub-Licensee"), however only for as long as such entities remain Affiliates of Calix. Calix shall ensure that each such sublicensed Affiliate complies with the terms of this Agreement. Each sub-license terminates automatically and simultaneously upon expiration or termination of this Agreement for whatever reason.
- 2.3. From time to time after the Closing Date Schedule 1 shall be deemed supplemented with any patent owned or controlled by Ericsson which meets the following requirements: (i) the application related thereto was filed and owned or controlled by, or such patent was issued to and owned or controlled by, Ericsson as of the Closing Date; and (ii) such patent is owned or controlled by Ericsson on the date it is deemed to supplement Schedule 1 and (iii) such patent is asserted by Ericsson against Calix during the term hereof in any action, suit or litigation as being infringed by any Ericsson Product ("Supplemental Patent"). If and when the foregoing requirements are met with respect to a patent, such Supplemental Patent shall be considered a Licensed Patent hereunder, subject to all of the provisions hereof.
- 2.4. All rights not expressly granted by Ericsson are hereby expressly reserved.

3. **INFRINGEMENT**

3.1. Calix shall notify Ericsson promptly on becoming aware of any action from third parties challenging any

Licensed Patent.

- 3.2. Ericsson and/or LME may, in its sole discretion and at its own expense, bring or cause to be brought any prosecution, lawsuit, action or proceeding for infringement, unauthorized use, or interference with or violation of any Licensed Patent, and Calix shall not be entitled to take any such actions of whatsoever nature against third parties unless it has received Ericsson's prior written approval (and Calix shall not, for example, send letters of notice or cease-and-desist letters referring to any Licensed Patents).
- 3.3. Calix shall not (and shall procure that its Affiliates and Sub-Licensees do not) make any claim or allegation which challenges the right of Ericsson or its Affiliates to, or the validity of, any of the Licensed Patents.

4. **NO WARRANTIES**

- 4.1. No representation, condition or warranty whatsoever, expressed or implied, is made or given by or on behalf of Ericsson or its Affiliates with respect to the Licensed Patents and all conditions and warranties relating to such patents (including as to validity or fitness for any purpose), whether arising by operation of law or otherwise, are expressly excluded.
- 4.2. At all times during the term hereof, and hereafter, Ericsson and/or LME shall have the sole right and discretion to assign, further license or sublicense, prosecute, discontinue prosecution, defend, decline to defend, maintain, decline to maintain and/or abandon any of the Licensed Patents.
- 4.3. Calix undertakes to indemnify Ericsson and its Affiliates and hold them harmless from any liability (including but not limited to liability under any applicable national or international laws or principles) which arises directly from the misuse of the Licensed Patents by Calix or its Affiliates after the Closing Date (including without limitation any mislabeling of the Ericsson Products with patent numbers, or any attempt by Calix or its Affiliates to assert a Licensed Patent which results in actual damages to Ericsson).

5. **TERM AND TERMINATION**

5.1. This Agreement including the license in Section 2 enters into full force and effect on the Closing Date and remains valid until the date of expiration of the last to expire of any Licensed Patent, unless terminated earlier in accordance with Section 6.4 a. and b.

6. **MISCELLANEOUS**

6.1. **Confidentiality**

- 6.1.1. Each Party undertakes not to use or disclose the contents of this Agreement or any information relating to the other Party and disclosed in connection with this Agreement, (such as, without limitation, Know How, trade secrets, customer lists) and which is not in the public domain unless:
 - (e) required to do so by law or under any order of court or other competent authority or tribunal;
 - required to do so by any applicable stock exchange regulations or the regulations of any other recognized market place;
 - (g) such disclosure has been consented to by the other Party in writing; or
 - (h) the information is disclosed to its professional advisers who make no other use of the information than for assisting the Party and who are bound to such Party by a duty of confidence which applies to any information disclosed.
- 6.1.2. If a Party becomes required, in circumstances contemplated by 6.1.1(a) or (b) to disclose any information, such Party (the " **Disclosing Party**") shall provide the other Party (the " **Non-Disclosing Party**") with prompt written notice of that fact so that the Non-Disclosing Party may seek (with the cooperation and reasonable efforts of the Disclosing Party) a protective order, confidential treatment or other appropriate remedy. In such event, the Disclosing Party shall consult with the Non-Disclosing Party regarding such proposed disclosure, shall furnish only that portion of the information which is legally required and shall exercise reasonable efforts to obtain reliable assurance that confidential treatment will be accorded such information to the extent reasonably requested by the Non-Disclosing Party.
- 6.1.3. The obligations set out above in this Section 6.1 shall survive for 6 years after the expiration or termination of this Agreement.
- 6.2. **Amendments and Waivers**

This Agreement may only be amended or modified by written agreement between the Parties. No failure delay in exercising any right or power under this Agreement will be effective as a waiver unless it is in		ilure		

writing and signed by the Party granting the waiver, and no such waiver will constitute a waiver of any other right or power under this Agreement.

6.3. **Notices**

6.3.1. All notices and other communications required or permitted under this Agreement must be in writing in the English language, and shall be addressed as set out below or to such other addresses as may be given by written notice in accordance with this Section 6.3.

If to Ericsson:

Group Function Legal Affairs, Att: Head of IPR Legal Services Telefonaktiebolaget LM Ericsson Torshamnsgatan 21 SE-164 83 Stockholm Sweden

Email: [***]

If to Calix: Calix, Inc. 1035 N. McDowell Blvd. Petaluma, CA 94954 USA

Attention: General Counsel

Email: [***]

All such notices and other communications shall be effective upon actual receipt if delivered by hand, sent by reputable international overnight courier, or sent by email (with confirmation of receipt).

6.4. **Assignments and Change of Control**

a. This Agreement shall be binding upon and inure to the benefit of the successors of the Parties but shall not be assignable by Calix without the prior written consent of Ericsson, which shall not be unreasonably withheld. Calix acknowledges and agrees that it shall be reasonable for Ericsson or its Affiliates to withhold consent for, inter alia, any requested assignment of this Agreement to any competitor of Ericsson or its Affiliates. Ericsson may assign its rights and obligations under this Agreement to an Affiliate and may assign, transfer, sell, further license or sublicense any of the Licensed Patents to any Affiliate or third party, provided that such assignment, transfer, sale, further license or sublicense of this Agreement to an Affiliate is made subject to the rights of Calix as set forth herein, including those under Section 2.3, and provided further that any assignment, transfer, sale, further licensing or sublicensing any of the Licensed Patents to any Affiliate or third party is made subject to the rights of Calix with respect to such Licensed Patents as set forth herein. This Agreement shall automatically terminate (i) in the event Calix attempts to assign the Agreement without Ericsson's prior written consent; or (ii) to sublicense Calix's rights, except as permitted by Section 2.2.

b. Calix shall notify Ericsson not less than 30 days prior to any Change of Control. Such notice shall set out all relevant circumstances triggering the Change of Control. Provided that Ericsson, within 30 days after having obtained Calix's notice, informs Calix that Ericsson does not consent to the Change of Control, Ericsson shall be entitled to terminate this Agreement with immediate effect upon the occurrence of the Change of Control

6.5. **Partial Invalidity**

If any provision of this Agreement or the application of it shall be declared or deemed void, invalid or unenforceable in whole or in part for any reason, the Parties shall amend this Agreement in order to give effect to, so far as is possible, the spirit of this Agreement. If the Parties fail to amend this Agreement, the provision, which is void, invalid or unenforceable, shall be deemed deleted and the remaining provisions of this Agreement shall continue in full force and effect.

7. GOVERNING LAW AND DISPUTES

- 5.5. The laws of England and Wales, excluding its conflict of law principles, govern all matters arising under this Agreement.
- All disputes, differences or questions arising out of or relating to the interpretation or performance of this Agreement, between the Parties, shall be finally settled in London, United Kingdom, under the Rules of Arbitration of the International Chamber of Commerce, by three (3) arbitrators, appointed in accordance with said Rules. The arbitration proceedings shall be conducted in the English language. The Parties hereby consent to such dispute being so resolved.
- 5.7. The Parties undertake and agree that all arbitral proceedings conducted or initiated with reference to this Section 7 will be kept strictly confidential. This confidentiality undertaking shall cover the fact that arbitration has been initiated, all information disclosed in the course of such arbitral proceedings, as well as any decision or award that is made or declared during the proceedings. Information covered by this confidentiality undertaking may not, in any form, be disclosed to a third party without the prior written consent of the other Party. This notwithstanding, a Party shall not be prevented from disclosing such information in order to safeguard in the best possible way such Party's rights vis-à-vis the other Party in connection with the dispute, or if such a right exists pursuant to statute, regulation, a decision by an authority, a stock exchange contract or similar.
- 5.8. In case this Agreement or any part of it is assigned or transferred to a third party, such third party shall automatically be bound by the provisions of this arbitration clause.

This Agreement has been duly executed in two identical originals, of which each Party has taken one.

TELEFONAKTIEBOLAGET	CALIX, INC.
LM ERICSSON	
By:	By:
Name:	Name:
Title:	Title:

SCHEDULE 1

[***] Certain information in this document has been omitted and filed separately with the Securities and Exchange Commission. Confidential treatment has been requested with respect to the omitted portions.

[***]

SCHEDULE 2

List of Ericsson Products

[***]

Exhibit D Transition Services

1.0 "Lights On" Period and Milestones .

- 1.1 This Exhibit details the support required in transitioning the Purchased Assets and Assumed Liabilities from Ericsson to Calix.
- 1.2 The Parties agree that the Transition Services shall address the following activities: (i) transition of development and test related activities, including knowledge transfer, test capability transfer, development and test rigs, and requisite lab facilities from their existing distributed base of global locations to San Jose, thus enabling Calix to carry out the continued 1500 product line and the agreed services as agreed in the MSA; (ii) transition from Ericsson manufacturing to Calix manufacturing.
- 1.3 The Transition period shall be initially be set for 90 days from the Closing Date ("Lights On" Period), however, it may be terminated earlier if all of the described exit Milestones are achieved. If the milestones are not achieved by the 90 day mark, an extension on support of the final remaining items may be mutually agreed to by the Parties.

2.0 <u>Transition of Development and Test Activities</u>

2.1 Ericsson Development Software Support deliverables are as set forth below:

Multi Project Plan Milestones

Prior to or during the Lights On Period, the following releases shall be published by Ericsson:

R12A CP1

R12A CP2

R12A GA

R12A CP3

R11A CP9

Completion of these deliverables will end of this portion of the Lights On requirement for supporting these functions. All packages follow a milestone model that sets clear expectations on scope definition and quality checks before committing the content. The criteria are specified in the maintenance work flow.

CIG ONT releases

CIG WP 2.2 - Released with R11A EC and R12A EC

CIG WP 3 - Released with R11A EC and R12A EC

2.2 System Test Certification Milestone Deliverable

Calix shall successfully perform a complete system test of the latest release and correction package of all systems in the 1500 product line. This test shall be at the same level (full regression test suite) as that used for GA of the R12 and CIG products. The test setup shall include an ONT configuration that is equivalent to the resources currently

used at ETK, China, and San Jose for R11 and R12 testing, and the test shall be performed on the Calix test system after relocation of purchased assets to their final location. Completion of this milestone will denote final certification of the test system capability, under which Calix can properly create and test a new release of the 1500 with no diminished robustness from the Ericsson methodology.

2.3 In addition, emergency corrections might be required in between these scheduled releases and will be provided by Ericsson until Closing Date.

2.4 EMS knowledge transfer

2.4.1 The Parties seek to facilitate the move of EMS maintenance activities from Ericsson China to Calix in accordance with Ericsson's existing process

2.4.2 Milestones

- 2.4.2.1 Four employees trained and knowledgeable to assume the existing workload, including but not limited to systems engineering and architecture, design, coding, unit testing, and releasing.
- 2.4.2.2 Three employees trained on feature testing, system testing, automated testing.
- 2.4.2.3 Any employees willing to transfer from Ericsson Beijing to Calix Nanjing by the Closing Date may be counted against this total.

2.5 ONT knowledge transfer

2.5.1 The Parties seek to facilitate the move of ONT maintenance activities from Ericsson China to Calix in accordance with Ericsson's existing process

2.5.2 Milestones

- 2.5.2.1 Four employees trained and knowledgeable to assume the existing workload, including but not limited to systems engineering and architecture, design, coding, unit testing, and releasing.
- 2.5.2.2 Three employees trained on feature testing, system testing, automated testing.
- 2.5.2.3 ONT (CIG) knowledge transfer. One employee trained and knowledgeable to assume the technical interface responsibilities with CIG. One employee trained and knowledgeable about the VoIP and TR-69 interoperability functions of using CIG ONTs with the rest of the EDA1500 system in customer networks.
- 2.5.2.4 Any employees willing to transfer from Ericsson Beijing to Calix Nanjing by the Closing Date may be counted against this total.

2.6 OLT hardware knowledge transfer

2.6.1 The Parties seek to facilitate the move of OLT hardware maintenance activities from Ericsson China to Calix per Ericsson's existing process

2.6.2 Milestones

- 2.6.2.1 Four employees trained and knowledgeable to assume the existing workload, including but not limited to hardware engineering, component selection and verification, and regulatory and compliance testing.
- 2.6.2.2 Any employees willing to transfer from Ericsson Beijing to Calix Nanjing by the Closing Date

may be counted against this total.

2.7 Costs associated with Knowledge Transfer

- 2.7.1 Ericsson shall provide the required training at a facility that it specifies. Ericsson shall be responsible for the costs of the trainers.
- 2.7.2 Calix shall be responsible for all costs, including travel costs, associated with its employees.

2.8 Transfer of Suppliers interfaces (here ONT 3pp and ODMs)

- 2.8.1 The parties seek to facilitate the move of the Supplier interfaces from Ericsson per Ericsson's existing process.
- 2.8.2 Milestones
 - 2.8.2.1 One employee trained and knowledgeable about the day-to-day development and maintenance activities for each of Ericsson's ONT ODMs and OEMs.
 - 2.8.2.2 One employee trained and knowledgeable about the commercial contracts, supply agreements and forecasts, and warranty agreements for each of Ericsson's ONT ODM and OEMs.

2.9 Training of Missing Critical Resources

- 2.9.1 If at Closing one or more Employees listed in Schedule 6.4 elects not to become a Calix employee, Ericsson shall, in good faith and at Ericsson's expense, train a substitute employee, reasonably acceptable to Calix, to take the place of each such Employee.
- 2.9.2 If at Closing one or more of the key functional areas listed in Schedule 6.5 is not covered by employees transferring from Ericsson to Calix, Ericsson shall in good faith and at Ericsson's expense train an employee to provide that function, reasonably acceptable to Calix, to maintain that functional expertise.
- 2.9.3 The milestones described in Section 2.9 and Ericsson's obligation to train additional employees shall end on the earlier of (i) 90 days from Closing or (ii) when any and all additional employees have been trained to the reasonable satisfaction of Calix.

2.10 Customer Documentation Handling

- 2.10.1 Milestones
 - 2.10.1.1 Deliver source files for current CPI (Customer Product Information) documentation.
 - 2.10.1.2If Calix wishes to engage a third party CPI provider, Ericsson will provide up to two weeks training of a CPI Coordinator role. Training will be provided to one of the Ericsson employees to be hired by Calix, either before or after Closing Date. Training topics will include: (i) Current interworking with the 3PP vendor, including raw material to provide 3PP vendor; (ii) Internal CPI development process including raw material development and quality process etc., (iii) Project CPI planning as part of release project; (iv) CPI Maintenance process for maintenance releases.

2.11 Facility and Test Equipment

2.11.1 Ericsson will provide free of charge remote access to an Ericsson test facility, during the transition of equipment from Ericsson to Calix lab, and test services for the earlier to occur of (a) a period of three weeks from the lab relocation or (b) such time as Calix time to is able to complete its own lab facilities

or has access to Ericsson facilities that allow Calix to meet the milestone set forth in Section 2.2, above. After the three weeks, Calix may continue to utilize the remote test facility and test services for a fee. If full lab functionality cannot be achieved immediately after lab relocation, Ericsson will provide additional remote access free of charge until the milestone set forth in Section 2.2 can be met.

- 2.11.2 The provided test facility will provide all of the equipment necessary to allow for the development and maintenance of all EDA 1500 products, including regression testing, scalability testing and the troubleshooting of Customer problems (CSRs and TRs); source code management, build environment, debug tools, automated testing and system level testing.
- 2.11.3 Ericsson will document the test plant configuration and preserve necessary configuration files and provide to Calix within four weeks of signature of Asset Purchase Agreement. Information shall include an itemized equipment list, sufficient for all EDA 1500 development and test activities. Ericsson will update this information as necessary between the signing and the Closing Date.
- 2.11.4 Ericsson shall de-install, package, and make available at the Ericsson facilities on the Closing Date the capital assets purchased under the Asset Purchase Agreement.

3. Hardware Warranty Services

- 3.1. Ericsson shall provide Calix with all assistance reasonably necessary to accomplish the transition of PON Product Warranty Services to a Calix-designated facility. Such assistance shall include, without limitation:
 - 3.1.1. The transfer to the Calix-designated facility, at no additional charge to Calix, of all test fixtures required to repair Ericsson PON Products, all PON Products in WIP, and all raw materials required to perform repair process on PON Products;
 - 3.1.2. Provision to Calix of all PON Product repair rework instructions; and
 - 3.1.3. Two weeks of training performed by Ericsson on-site at the Calix-designated facility.
 - 3.1.4. Ericsson will provide hardware repair services on Ericsson PON Products until the transition to the Calix-designated repair facility has been completed.

4. Manufacturing Transfer

- 4.1 Ericsson will provide ongoing manufacturing support services to Calix for a maximum period of the Lights On Period ("Manufacturing Support Period), for the purpose of supporting the transition of manufacturing of Ericsson Products from Ericsson manufacturing facilities to Calix-designated facilities. Calix will work in good faith to minimize the duration of the Manufacturing Support Period. Among other things, Ericsson agrees to allow Calix to use Ericsson manufacturing operations (e.g. Ericsson's Flextronics relationship and CIG contract), facilities and systems to process new orders and builds of Ericsson Products on Calix's behalf. The primary areas in which Calix will require support from Ericsson during the Manufacturing Support Period are: operations, supply and production testing.
- 4.2 Information Requirements. On or before the Closing Date, Ericsson shall provide Calix with:
 - 4.2.1 Fully-costed bills of materials (BOMs) for all Ericsson Products to be transferred to Calix; and
 - 4.2.2 Designated points of contact at Ericsson and its affiliates for all key individuals necessary to the provision of manufacturing support services.
- 4.3 Orders Placed on Calix's Behalf. Any orders placed by Ericsson on Calix's behalf during the Manufacturing Support Period will be paid for by Calix. In addition, the Parties shall negotiate in good faith a transitional shipping process to eliminate any unnecessary movement of inventory between the Parties.

- 4.4 Ericsson Deliverables. During the Manufacturing Support Period, Ericsson shall provide Calix, on a weekly basis, with an updated:
 - 4.4.1 Demand plan for Ericsson Products;
 - 4.4.2 Detailed list of inventory subject to transfer to Calix; and
 - 4.4.3 Analysis of the gap likely to apply for Ericsson Product orders to be transitioned to Calix.
 - 4.4.4 In additional, Ericsson shall provide all other ancillary functions and services necessary to the success of the transition of Ericsson Product manufacturing from Ericsson to Calix.
- 4.5 Equipment Transition. During the Manufacturing Support Period, Ericsson shall work closely with Calix to arrange for the efficient transfer of necessary equipment (test fixtures, etc.) from Ericsson to Calix.
 - 4.5.1. Milestone: After move of OLT test fixtures, Ericsson and Calix will complete a successful production test.

Exhibit E Disclosure Schedule

Pursuant to Article II, Ericsson makes the following disclosures. Section references are to the respective section of the main body of the Agreement.

2.5(b) Government Consents; Approvals; Permits

Ericsson has the following export licenses and classifications related to the Business: License# Issued Ultimate Consignee(s) Additional Comments/Product lines [***]

See also Attachment 1 reflecting export classifications for the Products.

Employees. - Two employees are eligible to work under an L-1 Visa. Name will be provided separately for confidentiality reasons.

- 2.6 Material Contracts
- (i) Commercial Third Party Software

Please see the Exhibit reflecting all Ericsson Software, Third Party Commercial Software, and Open Source Software used in connection with the Products.

(ii) Continuing Contracts for purchase of materials

See Attachment 2 reflecting the inventory ongoing inventory obligations related to the Products:

(iv) Contracts for development of content

The following suppliers provide content related to the Products

[***]

(v) The following suppliers provide manufacturing services for the Products:

[***]

(xiii) The following settlement agreements may have implications for Calix

[***]

2.11 <u>Intellectual Property</u>

(a)(i) Ericsson Proprietary Software

Please see Attachment 3 reflecting all Ericsson Software, Third Party Commercial Software, and Open Source Software used in connection with the Products. This list includes both Software included in the Products and tools used to compile and test the Products.

(a)(ii) Third Party Software Components

Please see Attachment 3 reflecting all Ericsson Software, Third Party Commercial Software, and Open Source Software used in connection with the Products. This list includes both Software included in the Products and tools used to compile and test the Products.

(a)(ii) Third Party Hardware Components

Please see the Attachment 4 reflecting Bill of Materials for the Products.

For the ONTs, Ericsson purchases them as completed assemblies as follows:

[***]

(i) Open Source Software

Please see Attachment 3 reflecting all Ericsson Software, Third Party Commercial Software, and Open Source Software used in connection with the Products. This list includes both Software included in the Products and tools used to compile and test the Products.

2.12(a) Third Party Patent Contracts

The settlements of the following cases resulted in Third Party Patent Contracts.

[***]

2.13 Employee matters

Employees who elected to participate in the Ericsson Employee Stock Purchase plan will be eligible for accelerated matching of their shares per the terms of the ESPP.

2.15 Customer and Suppliers

Customer disputes (reference to "Ericsson" are to the local legal entity, unless noted)

[***]
Supplier disputes

[***]

2.16 Products; Warranties; Purchase Commitments

The following known issues relating to the Products exist [***]

Article IV: Conduct Prior to the Closing

Pursuant to section 4.1(c), Ericsson shall promptly notify Calix of any change, occurrence or event that would reasonably be expected to be materially adverse to the Business. As noted in the disclosure in 2.15, Ericsson has sent the Sales Directive to the regions outlining the parameters of future Product sales. This sales directive may have a negative impact on future sales of Products prior to the Closing.

Schedule 5.6 "Transferred Employees"

[***]

Attachment 1 Export Related Classifications

[***]

Attachment 2 Inventory Report



Attachment 3 Software Consolidated Report

[***]

Attachment 4

Bill of Materials

[***]

[***]

Exhibit F List of Ericsson Products

[***]

CALIX, INC.

Non-Employee Director Cash Compensation Policy

- 1. <u>General</u>. This Non-Employee Director Cash Compensation Policy ("<u>Policy</u>") was adopted by the Board of Directors ("<u>Board</u>") of Calix, Inc. ("<u>Company</u>") on October 18, 2011, and is effective January 1, 2012.
- 2. <u>Annual Cash Compensation</u> . Each member of the Board who is not employed by the Company or one of its affiliates shall be entitled to an annual retainer with the amount determined as follows (the net sum for each director his or her "<u>Annual Retainer</u>"):

	 Amount
Base Retainer	\$ 40,000
Committee Service Premiums	
Audit Committee	\$ 10,000
Compensation Committee	\$ 7,500
Nominating & Corporate Governance Committee	\$ 5,000
<u>Chair Premiums</u>	
Board Chair	\$ 35,000
Audit Committee	\$ 20,000
Compensation Committee	\$ 15,000
Nominating & Corporate Governance Committee	\$ 10,000

- 3. <u>Timing of Payment</u>. Annual Retainers shall be paid in quarterly installments in arrears on the date of each regularly scheduled quarterly board meeting. Installments will be pro-rated for any partial period of service.
- 4. <u>Policy Subject to Amendment, Modification and Termination</u>. This Policy may be amended, modified or terminated by the Board at any time in the future at its sole discretion.

CERTIFICATION OF CHIEF EXECUTIVE OFFICER PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Carl Russo, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Calix, Inc. for the quarter ended September 29, 2012;
- Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting;
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 1, 2012	/s/ Carl Russo
	Carl Russo
	Chief Executive Officer

CERTIFICATION OF CHIEF FINANCIAL OFFICER PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Michael Ashby, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Calix, Inc. for the quarter ended September 29, 2012;
- Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting;
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 1, 2012	/s/ Michael Ashby
	Michael Ashby
	Chief Financial Officer

CERTIFICATION OF CHIEF EXECUTIVE OFFICER AND CHIEF FINANCIAL OFFICER PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

I, Carl Russo, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that the Quarterly Report of Calix, Inc. (the "Company") on Form 10-Q for the fiscal quarter ended September 29, 2012 fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934 and that information contained in such Quarterly Report on Form 10-Q fairly presents in all material respects the financial condition and results of operations of the Company.

Date: November 1, 2012	/s/ Carl Russo
	Carl Russo
	Chief Executive Officer

I, Michael Ashby, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that the Quarterly Report of Calix, Inc. (the "Company") on Form 10-Q for the fiscal quarter ended September 29, 2012 fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934 and that information contained in such Quarterly Report on Form 10-Q fairly presents in all material respects the financial condition and results of operations of the Company.

Date: November 1, 2012 /s/ Michael Ashby

Michael Ashby

Chief Financial Officer

This certification accompanies the Form 10-Q to which it relates, is not deemed filed with the Securities and Exchange Commission and is not to be incorporated by reference into any filing of Calix, Inc. under the Securities Act of 1933, as amended, or the Securities Exchange Act of 1934, as amended (whether made before or after the date of the Form 10-Q), irrespective of any general incorporation language contained in such filing.